Registration Statement No. 333-

SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM S-8
REGISTRATION STATEMENT
UNDER THE
SECURITIES ACT OF 1933

AMERICAN INTERNATIONAL GROUP, INC. (Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization) 13-2592361 (I.R.S. Employer Identification No.)

70 Pine Street, New York, New York 10270 (Address, including zip code, of principal executive offices)

> Kathleen E. Shannon Vice President and Secretary 70 Pine Street New York, New York 10270 (212) 770-7000

(Name, address, including zip code, and telephone number, including area code, of agent for service)

Title of securities to be registered	Amount to be registered	Proposed maximum offering price per share	Proposed maximum aggregate offering price	Amount of registration fee
Common Stock, \$2.50 par value (1)	28,971,381 shares (2)(3)	(3)	\$1,826,408,625 (3)	\$228,778(4)

- In addition, pursuant to Rule 416(c) under the Securities Act of 1933, this registration statement also covers an indeterminate amount of interests to be offered or sold pursuant to the American General Employees' Thrift and Incentive Plan, the American General Agents' and Managers' Thrift Plan and the CommoLoCo Thrift Plan (each, as amended, the "Thrift Plans"). Pursuant to Rule 457(h)(2) no separate fee is payable with respect to the registration of these interests.
- 2. This registration statement also relates to an indeterminate number of additional shares of Common Stock that may be issued pursuant to anti-dilution and adjustment provisions of the above-referenced plans.

- 3. Estimated solely for purposes of calculating the registration fee. With respect to the above-referenced plans marked with an asterisk (the "Stock Option Plans"), such estimate has been computed in accordance with Rule 457(h)(1) and is calculated based on the exercise price of the options issued under such plans to which the Common Stock is subject (no further stock options or incentive awards are expected to be issued under the Stock Option Plans). With respect to the Thrift Plans and the American General Corporation Deferred Compensation Plan, such estimate has been computed in accordance with Rule 457(c) and (h)(1) and is calculated based upon the average of the high and low sales prices of the Common Stock of American International Group, Inc. on August 28, 2001, \$76.98 per share, as reported on the New York Stock Exchange Composite Tape.
- 4. This registration statement covers 12,328,144 shares of the Registrant's Common Stock, par value \$2.50 per share, which were originally registered on Form S-4 (Registration Statement No.333-62688). Pursuant to Rule 457(p) the filing fee for this registration Statemen on Form S-8 has been offset against, the filing fee of \$227,824 previously paid by the registrant with respect to such shares of Common Stock.

EXPLANATORY NOTE

This Registration Statement on Form S-8 registers shares of common stock, par value \$2.50 per share (the "Common Stock"), of American International Group, Inc. ("AIG" or the "Company"), which may be issued in connection with the plans set forth on the facing page of this Registration Statement (the "Plans"). In addition, this Registration Statement also covers an indeterminate amount of interests to be offered or sold pursuant to the American General Employees' Thrift and Incentive Plan, the American General Agents' and Managers' Thrift Plan and the CommoLoCo Thrift Plan (each, as amended, the "Thrift Plans").

The Agreement and Plan of Merger, dated as of May 11, 2001, among AIG, Washington Acquisition Corporation and American General Corporation ("AGC"), provides that each option and each award will be converted into an option or right to acquire, on the same terms and conditions as were applicable under the relevant Plan, shares of Common Stock. The number of shares of Common Stock subject to such converted option or award is determined by multiplying the number of shares of AGC common stock that were subject to the option or award immediately prior to the effective time of the merger on August 29, 2001 by 0.5790.

PART I

INFORMATION REQUIRED IN THE SECTION 10(a) PROSPECTUS

All information required by Part I to be contained in the prospectus is omitted from this registration statement in accordance with Rule 428 under the Securities Act of 1933, as amended (the "Securities Act").

INFORMATION REQUIRED IN THE REGISTRATION STATEMENT

ITEM 3. INCORPORATION OF DOCUMENTS BY REFERENCE

The following documents have been filed by AIG with the Securities and Exchange Commission (the "Commission") (File No. 1-8787) and are incorporated herein by reference:

- (1) AIG's Annual Report on Form 10-K for the year ended December 31, 2000;
- (2) AIG's Quarterly Reports on Form 10-Q for the quarters ended March 31 and June 30, 2001;
- (3) AIG's Current Reports on Form 8-K, dated May 11, 2001 and August 29, 2001; and
- (4) The description of Common Stock contained in the Registration Statement on Form 8-A, dated September 20, 1984, filed pursuant to Section 12(b) of the Securities Exchange Act of 1934, as amended (the "Exchange Act").

The Annual Reports on Form 11-K for the fiscal year ended December 31, 2000 of the American General Employees' Thrift and Incentive Plan and the American General Agents' and Managers' Thrift Plan have been filed by AGC with the Commission (File No. 1-7981) and are incorporated herein by reference.

All documents filed by AIG and the Plans after the date hereof pursuant to Sections 13(a), 13(c), 14 and 15(d) of the Exchange Act, prior to the filing of a post-effective amendment which indicates that all securities offered have been sold, or which deregisters all such securities then remaining unsold, shall be deemed to be incorporated by reference in this registration statement and to be a part hereof from the date of filing of such documents.

Any statement contained in a document incorporated by reference herein shall be deemed to be modified or superseded for purposes of this registration statement to the extent that a statement contained herein or in any other subsequently filed document which also is or is deemed to be incorporated by reference herein modifies or supersedes such statement. Any such statement so modified or superseded shall not be deemed, except as so modified or superseded, to constitute a part of this registration statement.

ITEM 4. DESCRIPTION OF SECURITIES

The Common Stock is registered under Section 12(b) of the Exchange Act.

ITEM 5. INTEREST OF NAMED EXPERTS AND COUNSEL

The consolidated financial statements of AIG and its subsidiaries and the related financial statement schedules of AIG included in its Annual Report on Form 10-K for the year ended December 31, 2000, incorporated herein by reference, are so incorporated in reliance upon the report of PricewaterhouseCoopers LLP, independent accountants, given on the authority of that firm as experts in accounting and auditing.

The financial statements and schedules of the American General Employees' Thrift and Incentive Plan, and the American General Agents' and Managers' Thrift Plan, each included in the Plan's Annual Report (Form 11-K) for the year ended December 31, 2000, have been audited by Ernst & Young LLP, independent auditors, as set forth in their reports thereon included therein and incorporated herein by reference. Such financial statements and schedules are incorporated herein by reference in reliance upon such reports given on the authority of such firm as experts in accounting and auditing.

The validity of the shares of Common Stock to be offered and sold pursuant to the Plans and the interests to be offered and sold pursuant to the Thrift Plans will be passed upon by Kathleen E. Shannon, Esq., Vice President and Associate General Counsel of AIG. Ms. Shannon is employed by AIG, participates in various AIG employee benefit plans under which she may receive shares of Common Stock and currently beneficially owns less than 1% of the outstanding shares of Common Stock.

ITEM 6. INDEMNIFICATION OF DIRECTORS AND OFFICERS

The Restated Certificate of Incorporation, as amended, of AIG (the "Certificate") provides that AIG shall indemnify to the fullest extent permitted by law any person made, or threatened to be made, a party to an action, suit or proceeding (whether civil, criminal, administrative or investigative) by reason of the fact that he, his testator or intestate is or was a director, officer or employee of AIG or serves or served any other enterprise at the request of AIG, including services by a director, officer or employee with respect to an employee benefit plan. Section 6.4 of AIG's By-laws contains a similar provision.

The Certificate also provides that a director will not be personally liable to AIG or its stockholders for monetary damages for breach of fiduciary duty as a director, except to the extent that such an exemption from liability or limitation thereof is not permitted by the Delaware General Corporation Law (the "GCL").

Section 145 of the GCL permits indemnification against expenses, fines, judgments and settlements incurred by any director, officer, employee or agent of AIG in the event of pending or threatened civil, criminal, administrative or investigative proceedings, if such person was, or was threatened to be made, a party by reason of the fact that the person is or was a director, officer, employee or agent of AIG. Section 145 also provides that the indemnification provided for therein shall not be deemed exclusive of any other rights to which those seeking indemnification may otherwise be entitled.

In addition, AIG and its subsidiaries maintain a directors' and officers' liability insurance policy.

The Thrift Plans, included as exhibits to this registration statement and each incorporated by reference hereto, in Section 13.7 of each, provide that the Company "shall indemnify all those to whom it has delegated fiduciary duties against any and all claims, loss, damages, expense, and liability arising from their responsibilities in connection with the [p]lan, unless the same is determined to be due to gross negligence or willful misconduct."

ITEM 7. EXEMPTION FROM REGISTRATION CLAIMED

Not applicable.

ITEM 8. EXHIBITS

The exhibits are listed in the exhibit index.

ITEM 9. UNDERTAKINGS

AIG hereby undertakes:

- (1) To file, during any period in which offers or sales are being made, a post-effective amendment to this Registration Statement:
 - (i) To include any prospectus required by Section 10(a)(3) of the Securities Act;
 - (ii) To reflect in the prospectus any facts or events arising after the effective date of this Registration Statement (or the most recent post-effective amendment thereof) which, individually or in the aggregate, represent a fundamental change in the information set forth in this Registration

Statement;

(iii) To include any material information with respect to the plan of distribution not previously disclosed in this Registration Statement or any material change to such information in this Registration Statement;

provided, however, that paragraphs (1)(i) and (1)(ii) do not apply if the information required to be included in a post-effective amendment by those paragraphs is contained in periodic reports filed with or furnished to the Commission by AIG pursuant to Section 13 or Section 15(d) of the Exchange Act that are incorporated by reference in this Registration Statement.

- (2) That, for the purpose of determining any liability under the Securities Act, each such post-effective amendment shall be deemed to be a new Registration Statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (3) To remove from registration by means of a post-effective amendment any of the securities being registered which remain unsold at the termination of the offering.

AIG hereby undertakes that, for purposes of determining any liability under the Securities Act, each filing of AIG's annual report pursuant to Section 13(a) or Section 15(d) of the Exchange Act that is incorporated by reference in this Registration Statement shall be deemed to be a new Registration Statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.

Each of the Thrift Plans hereby undertakes that, for purposes of determining any liability under the Securities Act, each filing of a Thrift Plan's annual report pursuant to Section 15(d) of the Exchange Act that is incorporated by reference in this Registration Statement shall be deemed to be a new Registration Statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.

Insofar as indemnification for liabilities arising under the Securities Act may be permitted to directors, officers and controlling persons of AIG or the Thrift Plans pursuant to the foregoing provisions, or otherwise, AIG and the Thrift Plans have been advised that in the opinion of the Commission such indemnification is against public policy as expressed in the Securities Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by AIG or any of the Thrift Plans of expenses incurred or paid by a director, officer or controlling person of AIG or any of the Thrift Plans in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, AIG or any of the Thrift Plans will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Securities Act and will be governed by the final adjudication of such issue.

SIGNATURES

The Registrant. Pursuant to the requirements of the Securities Act, the Registrant certifies that it has reasonable grounds to believe that it meets all of the requirements for filing on Form S-8 and has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of New York, State of New York, on the 29th day of August, 2001.

AMERICAN INTERNATIONAL GROUP, INC.

By: /s/ Howard I. Smith

Name: Howard I. Smith

Title: Executive Vice President and

Chief Financial Officer

KNOW ALL MEN BY THESE PRESENTS: that each person whose signature appears below constitutes and appoints M. R. Greenberg, Edward E. Matthews and Howard I. Smith, and each of them, as true and lawful attorneys-in-fact and agents with full power of substitution and resubstitution, for him or her and in his or her name, place and stead, in any and all capacities to sign any and all amendments (including post-effective amendments) to this Registration Statement on Form S-8, and to file the same, with all exhibits thereto, and other documents in connection herewith, with the Commission, granting unto said attorneys-in-law and agents, and each of them, full power and authority to do and perform each and every act and thing required and necessary to be done in and about the foregoing as fully for all intents and purposes as he or she might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents or any of them, or their or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

Pursuant to the requirements of the Securities Act, this Registration Statement on Form S-8 has been signed by the following persons in the capacities and on the dates indicated.

Signature	Title	Date
/s/ M.R. Greenberg	Chairman, Chief Executive Officer and Director (Principal	August 29, 2001
(M. R. Greenberg)	Executive Officer)	
/s/ Howard I. Smith	Executive Vice President, Chief Financial Officer and Director (Principal Financial	August 29, 2001
(Howard I. Smith)	Officer)	
/s/ Michael J. Castelli	Vice President and Comptroller (Principal	August 29, 2001
(Michael J. Castelli)	Accounting Officer)	
(M. Bernard Aidinoff)	Director	

Signature	Title	Date
(Eli Broad)	Director	
/s/ Pei-yuan Chia	Director	August 29, 2001
(Pei-yuan Chia)		
/s/ Marshall A. Cohen	Director	August 29, 2001
(Marshall A. Cohen)		
/s/ Barber B. Conable, Jr.	Director	August 29, 2001
(Barber B. Conable, Jr.)		
/s/ Martin S. Feldstein	Director	August 29, 2001
(Martin S. Feldstein)		
/s/ Ellen V. Futter	Director	August 29, 2001
(Ellen V. Futter)		
/s/ Carla A. Hills	Director	August 29, 2001
(Carla A. Hills)		
/s/ Frank J. Hoenemeyer	Director	August 29, 2001
(Frank J. Hoenemeyer)		
(Richard C. Holbrooke)	Director	
/s/ Edward E. Matthews	Director	August 29, 2001
(Edward E. Matthews)		- J
/s/ Thomas R. Tizzio	Director	August 29, 2001
(Thomas R. Tizzio)		
(Edmund S. W. Tse)	Director	
/s/ Jay S. Wintrob	Director	August 29, 2001
(Jay S. Wintrob)		

Signature	Title	Date
/s/ Frank G. Wisner	Director	August 29, 2001
(Frank G. Wisner)		
/s/ Frank G. Zarb	Director	August 29, 2001
(Frank G. Zarb)		

The Thrift Plans. Pursuant to the requirements of the Securities Act, the trustees (or other persons who administer each employee benefit plan indicated below) have duly caused this Registration Statement to be signed on their behalf by the undersigned, thereunto duly authorized in the City of New York, State of New York, on August 29, 2001.

AMERICAN GENERAL EMPLOYEES' THRIFT AND INCENTIVE PLAN

By: /s/ Gary D. Reddick

Name: Gary D. Reddick

Title: Executive Vice President-Administration

and Insurance Operations

AMERICAN GENERAL AGENTS' AND MANAGERS' THRIFT PLAN

By: /s/ Gary D. Reddick

Name: Gary D. Reddick

Title: Executive Vice President-Administration

and Insurance Operations

COMMOLOCO THRIFT PLAN

By: /s/ Gary D. Reddick

Name: Gary D. Reddick

 ${\bf Title:} \ {\bf Executive} \ {\bf Vice} \ {\bf President-Administration}$

and Insurance Operations

EXHIBIT INDEX

EXHIBIT NUMBER	DESCRIPTION	LOCATION
4	(a) American General Employees' Thrift and Incentive Plan, as restated July 1, 2001	Filed as exhibit hereto.
	(b) American General Agents' and Managers' Thrift Plan, as restated July 1, 2001	Filed as exhibit hereto.
	(c) CommoLoCo Thrift Plan, as restated July 1, 2001	Filed as exhibit hereto.
	(d) American General Corporation Deferred Compensation Plan, effective as of July 4, 1998, as restated on December 11, 2000	Incorporated herein by reference to exhibit 10.13 to American General Corporation's Form 10-K, as filed with the Commission on March 28, 2001 (File No. 001-07981).
5	Opinion of Kathleen E. Shannon re validity	Filed as exhibit hereto.
15	Letter re unaudited interim financial information	Not applicable.
23	Consents of experts and counsel (a) PricewaterhouseCoopers LLP	Filed as exhibit hereto. Filed as exhibit hereto.
24	Power of Attorney	Included in signature pages.

AMERICAN GENERAL

EMPLOYEES' THRIFT AND INCENTIVE PLAN

July 1, 2001 Restatement (incorporating the January 1, 1997 Restatement and Amendments 1-5 thereto)

ARTICLE I -	PURPOSE1
ARTICLE II	- DEFINITIONS
2.1	Administrative Board2
2.2	Base Pay2
2.3	Beneficiary2
2.4	Board of Directors2
2.5	Certified Public Accountant
2.6	Change Date3
2.7	Code3
2.8	Company3
2.9	Company Stock3
2.10	Controlled Entity3
2.11	Effective Date3
2.12	Employee3
2.13	Employee Contribution Account3
2.14	Employer3
2.15	Employer Contribution Account3
2.16	Employment Commencement Date4
2.17	Highly Compensated Participant4
2.18	Hour of Service5
2.19	Investment Funds5
2.20	Leased Employee5
2.21	Long-Term Disability Program5
2.22	Nonforfeitable Interest5
2.23	Non-Highly Compensated Participant5
2.24	Normal Retirement Date5
2.25	One-Year Break in Service6
2.26	Participant6
2.27	Period of Separation6
2.28	Period of Service6
2.29	Period of Severance7
2.30	Plan8
2.31	Plan Administrator8
2.32	Plan Year8
2.33	Prior Employee Contribution Account8
2.34	Reemployed Individual8
2.35	Rollover Account8
2.36	Rollover Contributions8

	.37	Secretary8
	.38	Stock Fund8
	.39	Total Disability8
	. 40 . 41	Transfer Date
	.41	Trust9
	.42	Trustee9
	.43	Trust Fund9
	. 45	Withdrawal Date9
۷.	. 40	withdrawal baccommission of the same and the
ARTICLE	III	- PARTICIPATION
3.	. 1	Eligibility10
	. 2	Reemployed Individuals10
3.	. 3	Participation10
4 D T T O L E	T) (CONTRIBUTIONS AND ALLOCATIONS
		CONTRIBUTIONS AND ALLOCATIONS
	.1	Employer Contributions
	.1a	Employer Safe Harbor Contributions
4. 4.	_	Employee Contributions
4.	-	Investment and Allocation of Contributions14
4.		Limitation on Allocation of Contributions and Forfeitures15
4.	-	Rollover Contributions
4.	-	Voting, Tendering, and Exchanging of Company Stock
4.		Restrictions on Employee Contributions
4.	-	Restrictions on Employer Contributions
	.10	Excess Contributions
ARTICLE	V -	VALUATION OF ACCOUNTS21
ARTTCI F	VT -	VESTING22
6.		Normal Retirement
6.	_	Plan Termination, Partial Plan Termination, or Complete Plan
		Discontinuance of Employer Contributions22
6.	. 3	Vesting of Employee Contributions22
6.	. 4	Vesting of Employer Contributions22
6.	. 5	Vesting After a Period of Severance23
6.	. 6	Vesting Before a Period of Severance of Five Consecutive
		One-Year Breaks in Service23
6.	. 7	Election to Make a Withdrawal23
	. 8	Transfer Between Plans23
6.	. 9	Independent Life Transferred Employees23

ARTICLE VII 7.1 7.2 7.3 7.4 7.5 7.6 7.7	- BENEFITS 24 Normal Retirement 24 Disability 24 Death 24 Termination of Service 24 Valuation Date to be Used 24 Withdrawals 25 Loans 27
ARTICLE VII 8.1 8.2 8.3 8.4	I - COMMENCEMENT OF BENEFITS
ARTICLE IX 9.1 9.2 9.3 9.4 9.5	- DISTRIBUTION OF BENEFITS
ARTICLE X - 10.1 10.2 10.3 10.4 10.5 10.6	PLAN AMENDMENT AND TERMINATION
ARTICLE XI 11.1 11.2 11.3 11.4 11.5 11.6	- ADMINISTRATION. 34 Named Fiduciaries. 34 Appointment of the Trustee. 34 Trustee's Powers and Duties. 34 Administrative Expenses. 34 Plan Administrator's Powers and Duties. 34 Named Fiduciary's Powers and Duties. 35 Allocation of Functions. 35
ARTICLE XII 12.1 12.2 12.3	- ADOPTION AND WITHDRAWAL

ARTICLE XIII - MISCELLANEOUS	. 38
13.1 Merger of this Plan with Another Plan	. 38
13.2 Assignment and Alienation of Benefits	. 38
13.3 Communication to Employees	. 38
13.4 Number and Gender	. 38
13.5 Construction	. 38
13.6 Not a Contract of Employment	. 38
13.7 Indemnity	. 39
13.8 Change of Beneficiary	. 39
13.9 Applications to the Administrative Board	. 39
13.10 Top-Heavy Rules	
13.11 Qualified Domestic Relations Orders	. 43
13.12 Uniformed Services Employment and Reemployment Rights Act	
Requirements	45

ARTICLE I

PURPOSE

Effective July 1, 2001 American General Corporation adopts the restated Plan, as set forth herein, to replace the American General Employees' Thrift and Incentive Plan previously in effect.

The Plan was originally established by American General Insurance Company for its Employees effective August 1, 1968. The Plan was restated in 1976, 1980, 1987, 1990, 1991 and 1997. American General Insurance Company was reorganized as a general business corporation named American General Corporation effective July 1, 1980.

The Plan was instituted by the Company to encourage systematic savings and the accumulation of financial reserves by Employees of the Company and its subsidiaries, and to give Employees an opportunity to acquire an ownership interest in American General Corporation as well as enabling Employees to reap greater direct benefits from the Company's success. The Plan is carried out through Employee contributions deducted from the payroll on a pre-tax basis and contributions by the Company or its subsidiaries. Company contributions are invested in American General common stock purchased by a Trustee responsible for administering the Trust Fund. Employee contributions are invested by the Trustee at each Participant's discretion in American General common stock or in other designated investment funds.

The Plan and Trust are intended to meet the requirements of sections 401(a) and 501(a) of the Internal Revenue Code of 1986, as amended, and the requirements of the Employee Retirement Income Security Act of 1974, as amended.

This Plan and the separate related Trust forming a part hereof are established and shall be maintained for the exclusive benefit of the eligible Employees of the Employers and their Beneficiaries. No part of the Trust Fund can ever revert to the Employers, except as provided in Article X, or be used for or diverted to purposes other than the exclusive benefit of the Employees of the Employers and their Beneficiaries.

ARTICLE II

DEFINITIONS

As used in this Plan, the following words and phrases shall have the meanings set forth below, unless the context clearly indicates otherwise.

- 2.1 Administrative Board. A board composed of at least three Company officers or Employees appointed by the Board of Directors to administer the Plan, located at Company headquarters at 2929 Allen Parkway, Houston, Texas 77019.
- 2.2 Base Pay. The compensation of the Employee as stated on Employer payroll records, such amount to exclude, however, any pay for overtime (which shall be deemed to refer as well to any shift differential payments), discretionary bonuses (which shall be deemed to refer as well to educational awards, educational reimbursements, instructor fees, referral awards, moving expenses, mortgage assistance, personal use of an Employer-owned vehicle, ACE awards, or workers' compensation payments), severance payments, and Employer contributions to this or any other deferred or noncash compensation program; provided, however, that any salary reduction amounts contributed on behalf of the Employee under a flexible benefits program pursuant to section 125 of the Code shall be included; and, provided, further, that salary reduction amounts contributed on behalf of the Employee under this Plan shall be included. Notwithstanding the foregoing, the Base Pay of any Employee taken into account under the Plan for any calendar year may not exceed \$150,000 (with such amount to be adjusted automatically to reflect any cost-of-living increases authorized by section 401(a)(17) of the Code).

2.3 Beneficiary.

- (A) The Beneficiary of a Participant shall be:
 - (1) the surviving spouse, if any, of the Participant; or
- (2) if there is no surviving spouse, or if the surviving spouse has executed a consent in accordance with Subsection 2.3(B), the person or persons designated in writing by the Participant; or
- (3) if there are no persons described in the preceding Subsections living at the date of the Participant's death, the Participant's estate.
- (B) The consent referred to in Subsection 2.3(A)(2) must be in writing, must acknowledge the effect of the Participant's designation of a person, other than the spouse, to receive the Participant's Nonforfeitable Interest upon the Participant's death, and must be witnessed by a Plan representative or a notary public.
 - 2.4 Board of Directors. The Board of Directors of the Company.

- 2.5 Certified Public Accountant.
- (A) A person who is a certified public accountant, certified by a regulatory authority of a state;
- (B) a person who is a licensed public accountant, licensed by a regulatory authority of a state; or
- (C) a person certified by the Secretary of Labor as a qualified public accountant.
- $2.6\,\,$ Change Date. The first day of any pay period as designated by a Participant.
- $2.7\,$ Code. The Internal Revenue Code of 1986, as amended from time to time.
 - 2.8 Company. American General Corporation and its successors.
 - 2.9 Company Stock. The common stock of the Company.
- 2.10 Controlled Entity. A corporation or other trade or business which is not an Employer hereunder, but which, together with an Employer, is "under common control" within the meaning of section 1.414(c)-2 of the Treasury Regulations.
- 2.11 Effective Date. January 1, 1997, as to this amendment and restatement of the Plan.
- 2.12 Employee. Any Employee of an Employer who is a resident of the United States or the U.S. Virgin Islands, excluding, however, Sales Employees (sometimes known as Home Service Representatives, Sales Representatives, Sales Managers, Field Representatives, Agents or Home Service Agents, or Sales or District Managers), Full-Time Life Insurance Agents, and any other field representatives with Flexi-Master Contracts, provided that the holder of a Flexi-Master Contract is deemed to be an employee under section 7701(a)(20) of the Code. Such term shall not include Leased Employees.
- 2.13 Employee Contribution Account. A separate account maintained for each Participant to which both basic and additional Employee contributions made on behalf of the Participant, and earnings and investment gains thereon, are credited, and which is invested in any of the separate investment funds.
- $2.14\ Employer.$ The Company and any other corporation which shall adopt this Plan pursuant to Article XII, and the successor, if any, to such corporation.
- 2.15 Employer Contribution Account. A separate account maintained for a Participant consisting of cash, dividends payable, and Company Stock purchased by Employer contributions and

reinvested dividends, and which may be invested in any of the separate investment funds after the Participant attains age sixty. Notwithstanding the foregoing, a separate subaccount of the Employer Contribution, Account known as the 'Employer Contribution Account Subaccount' shall be established and maintained for a Participant respecting Employer Contributions made to the Plan from and after January 1, 1999, and earnings thereon. Except as provided in Section 6.4 relating to vesting and Section 7.6 relating to withdrawals, a Participant's Employer Contribution Account Subaccount shall be considered as part of the Participant's Employer Contribution Account for all purposes of the Plan.

- $2.16\,$ Employment Commencement Date. The date on which an Employee first performs an Hour of Service.
- 2.17 Highly Compensated Participant. Each Participant or Former Participant who is an Employee who performs services during the Plan Year for which the determination of who is highly compensated is being made (the 'Determination Year') and who:
 - (A) is a five-percent owner of the Employer (within the meaning of section 416(i)(1)(A)(iii) of the Code) at any time during the Determination Year or the twelve-month period immediately preceding the Determination Year (the 'Look-Back Year'); or
 - (B) For the Look-Back Year:
 - (i) receives compensation (within the meaning of section 414(q)(4) of the Code; 'compensation' for purposes of this Paragraph) in excess of \$80,000 (with such amount to be adjusted automatically to reflect any cost-of-living adjustments authorized by section 414(q)(1) of the Code) during the Look-Back Year; and
 - (ii) if the Administrative Board elects the application of this clause for such Look-Back Year, is a member of the top 20% of Employees for the Look-Back Year (other than Employees described in section 414(q)(5) of the Code) ranked on the basis of compensation received during the year.

For purposes of the preceding sentence, (i) all employers aggregated with the Employer under section 414(b), (c), (m), or (o) of the Code shall be treated as a single employer, (ii) a former Employee who had a separation year (generally, the Determination Year such Employee separates from service) prior to the Determination Year and who was an active Highly Compensated Participant for either such separation year or any Determination Year ending on or after such Employee's fifty-fifth birthday shall be deemed to be a Highly Compensated Participant, and (ii) the Committee may elect, in accordance with the provisions of applicable Treasury regulations, rulings and notices, to make the Look-Back Year calculation for a Determination

Year on the basis of the calendar year ending with or within the applicable Determination Year (or, in the case of a Determination Year that is shorter than twelve months, the calendar year ending with or within the twelve-month period ending with the end of the applicable Determination Year). To the extent that the provisions of this Paragraph are inconsistent or conflict with the definition of a 'highly compensated employee' set forth in section 414(q) of the Code and the Treasury regulations thereunder, the relevant terms and provisions of section 414(q) of the Code and the Treasury regulations thereunder shall govern and control.

2.18 Hour of Service. An Hour of Service is each hour for which an Employee is directly or indirectly paid, or entitled to payment, by the Employer or a Controlled Entity for the performance of duties or for reasons other than the performance of duties. Such Hours of Service shall be credited to the Employee for the period in which such duties were performed or in which occurred the period during which no duties were performed. An Hour of Service also includes each hour, not credited above, for which back pay, irrespective of mitigation of damages, has been either awarded or agreed to by the Employer or a Controlled Entity. These Hours of Service shall be credited to the Employee for the period to which the award or agreement pertains rather than the period in which the award, agreement or payment is made.

The number of Hours of Service to be credited to an Employee for any period shall be governed by section 2530.200b-2(b) and (c) of the Labor Department Regulations relating to the Employee Retirement Income Security Act of 1974, as amended.

- 2.19 Investment Funds. Investment funds made available from time to time for the investment of plan assets as described in Article IV.
- 2.20 Leased Employee: Each person who is not an employee of the Employer or a Controlled Entity but who performs services for the Employer or a Controlled Entity pursuant to an agreement (oral or written) between the Employer or a Controlled Entity and any leasing organization, provided that such person has performed such services for the Employer or a Controlled Entity or for related persons (within the meaning of section 144(a)(3) of the Code) on a substantially full-time basis for a period of at least one year and such services are performed under primary direction or control by the Employer or a Controlled Entity.
- 2.21 Long-Term Disability Program. The American General Long-Term Disability Plan for Employees, the American General Long-Term Disability Plan for Executives, the American General Life and Accident Sales Employees' Non-Occupational Disability Income Plan, and any workers' compensation plan, program, or fund sponsored, maintained, or paid into by the Company or the other Employers for the benefit of its Employees and Career Agents, whichever may be applicable to the Participant at the relevant time.
- 2.22 Nonforfeitable Interest. The unconditional and legally enforceable right to which the Participant or Beneficiary (whichever is applicable) is entitled in the Participant's entire Employee Contribution Account balance and entire Prior Employee Contribution Account balance and in the

percentage of the Participant's Employer Contribution Account balance which has vested pursuant to Article ${\sf VI}$.

- 2.23 Non-Highly Compensated Participant. "Non-Highly Compensated Participant" shall mean any Participant or former Participant who is not a Highly Compensated Participant.
- 2.24 Normal Retirement Date. The first day of the month coincident with or next following a Participant's 65th birthday.
 - 2.25 One-Year Break in Service.
- (A) A 12-consecutive-month Period of Severance during which an Employee does not perform at least one hour of service.
- (B) Solely for the purpose of determining whether or not a Participant has a One-Year Break in Service, the Plan shall treat as an hour of service each hour of service which would otherwise have been credited to the Participant but for an absence beginning on or after January 1, 1985, for one of the following reasons:
 - (1) pregnancy of the Participant;
 - (2) birth of a child of the Participant;
 - $\hbox{(3)} \quad \hbox{placement of a child with the Participant in connection with an adoption proceeding; or} \\$
 - (4) caring for the child immediately following such birth or placement.

If those hours cannot be determined, the Plan shall treat eight hours per day of such absence as hours of service. The hours to be treated as hours of service because of such absence shall not exceed 501.

The hours required to be credited under (1)-(4) of this Subsection 2.24(B) shall be credited in the year in which such absence begins if it is necessary to prevent a break in service in that year; otherwise such hours shall be credited to the succeeding year.

No credit shall be given under this Subsection 2.24(B) unless the Employee furnishes to the Committee such information as it may reasonably require to establish that the absence is for one of the reasons listed in (1)-(4) of this Subsection 2.24(B), and to establish the number of days of such absence.

- 2.26 Participant. An Employee who satisfies the eligibility requirements of Article III and has enrolled in the Plan.
- $2.27\ \text{Period}$ of Separation. A period of time commencing with the date an Employee separates from service and ending with the date such Employee resumes employment with the

2.28 Period of Service. For purposes of determining an Employee's initial or continued eligibility to participate in the Plan or the vested interest in such individual's Employer Contribution Account, an Employee shall be credited for the time period commencing with his Employment Commencement Date, including years of service with the Employer in a category of Employees excluded from the Plan, and ending on the date his Period of Severance begins. A Period of Service for these purposes includes a Period of Separation of less than 12 consecutive months. In the case of an Employee who separates from service and later resumes employment with the Employer, the Period of Service prior to resumption of employment shall be aggregated only if such Employee is a Reemployed Individual.

In the case of a person who becomes an Employee, after being a "leased employee" under section 414(n) of the Code, the Employment Commencement Date will be deemed to be the date such Employee first performed services for the Employer as a "leased employee" or such other date as required by Treasury Regulations.

For purposes of determining a Period of Service, the following periods will be disregarded:

- (A) Periods of Service prior to June 29, 1991, which would not have been credited under the rules of the previous plan or its amendments or a predecessor plan. Such periods shall be determined according to provisions of the previous plan or predecessor plan, including rules which require minimum hours or other length of service or contributions by Participants;
- (B) Periods of Service prior to the date the Employer adopts this Plan or a predecessor plan; and $\,$
- (C) Periods of Service in which an Employee declined to contribute to the Plan or a predecessor plan prior to the date such Employee first commences participation in the Plan.

It is specifically provided that for purposes of determining Period of Service, service with a Controlled Entity shall be included.

- $2.29\,$ Period of Severance. A period of time commencing with the earlier of:
- (A) the date an Employee separates from service by reason of quitting, retirement, death, or discharge; or
- (B) the date 12 months after the date an Employee separates from service, including severance for reasons other than quitting, retirement, death, or discharge; and ending, in the case of an Employee who separates from service by reason other than death, with the date such Employee resumes employment with the Employer.
 - (C) An absence beginning on or after January 1, 1985, because of:

- (1) pregnancy of the Participant;
- (2) birth of a child of the Participant;
- (3) placement of a child with the Participant in connection with an adoption proceeding; or
 - (4) caring for the child immediately following such birth or placement;

shall not be considered a Period of Severance except as permitted by Treasury Regulations.

- (D) In order for the preceding Subsection 2.28(C) to apply, a Participant must furnish to the Administrative Board such information as it may reasonably require to establish that the absence is for one of the reasons listed in (1)-(4) of the preceding Subsection 2.28(C), and to establish the number of days of such absence.
- $2.30\,$ Plan. This document, the provisions of the Trust agreement, and any amendments to either.
- 2.31 Plan Administrator. The Administrative Board unless and until it designates such other person or persons. A member of the Administrative Board may resign or may be replaced by the Board of Directors at any time. No bond or other security shall be required of any member of the Administrative Board unless otherwise required by law.
- 2.32 Plan Year. The 12-month period beginning January 1 and ending December 31.
- 2.33 Prior Employee Contribution Account. A separate account maintained for each Participant who made Employee contributions to the Plan prior to June 29, 1991, to which such Employee contributions, and earnings and investment gains thereon, are credited, and which is invested in any of the separate investment funds.
- $2.34\,$ Reemployed Individual. A person who, after having separated from service, resumes employment:
 - (A) with a Nonforfeitable Interest in such individual's Employer Contribution Account; or
 - (B) with no such Nonforfeitable Interest, and who resumes employment either (1) before a One-Year Break in Service or (2) after a One-Year Break in Service but before the latest Period of Severance equals or exceeds the greater of (a) such individual's Period of Service or (b) five consecutive years of a Period of Service.
- 2.35 Rollover Account. An separate account maintained for each Participant who has made Rollover Contributions pursuant to Section 4.6 to which such Rollover Contributions and earnings and investment gains thereon, are credited, and which is invested any of the separate

- 2.36 Rollover Contributions. Contributions made by an Employee pursuant to Section 4.6.
 - 2.37 Secretary. The Secretary of the Treasury of the United States.
- 2.38 Stock Fund. The separate investment fund forming a part of the Trust which is invested in, or held for investment in, Company Stock. Dividends applicable to Company Stock will be applied to the purchase of additional Company Stock.
- 2.39 Total Disability. The definition of disability as used under the Company's Long Term Disability Program as in effect at the inception of the Participant's disability, respecting the applicable plan, program, or fund pursuant to which the Participant is entitled to receive disability benefits.
- 2.40 Transfer Date. The first date on which the New York Stock Exchange is open for business coinciding with or next following receipt by the Plan Administrator of a transfer request from the Participant in accordance with the procedures established from time to time by the Administrative Board.
- 2.41 Treasury Regulations. Regulations issued from time to time by the Secretary or the Secretary's delegate.
- 2.42 Trust. The agreement between the Trustee and the Employer entered into for the purpose of holding, managing, and administering all property held by the Trustee for the exclusive benefit of the Participants and their Beneficiaries.
- 2.43 Trustee. The person designated by the Company pursuant to Section 11.2 and in accordance with the Trust agreement, and any successor who is appointed pursuant to the terms of that Section.
- $2.44\,$ Trust Fund. All assets held by the Trustee pursuant to the terms of this Plan.
- 2.45 Withdrawal Date. The first date on which the New York Stock Exchange is open for business coinciding with or next following receipt by the Plan Administrator of a withdrawal request from the Participant in accordance with the procedures established from time to time by the Administrative Board.

ARTICLE III

PARTICIPATION

3.1 Age and Service.

- A. An Employee whose base pay is calculated on the basis of an annual, monthly, semi-monthly, bi-weekly or weekly salary or who is a regular Employee of American General Finance, Inc. or its subsidiaries shall be eligible to participate in this Plan upon the Employee's Employment Commencement Date.
- B. An Employee whose base pay is not calculated on the basis of an annual, monthly, semi-monthly, bi-weekly or weekly salary and who is not a regular Employee of American General Finance, Inc. or its subsidiaries shall be eligible to participate in this Plan upon the completion of a one-year Period of Service
- 3.2 Reemployed Individuals. A Reemployed Individual is eligible to participate in the Plan on the later of the date of reemployment by the Employer or the date described in Section 3.1.
- 3.3 Participation. Participation in the Plan shall commence as soon as administratively feasible following the date the eligibility requirement is satisfied, provided the Administrative Board has received a properly completed enrollment in accordance with the procedures established from time to time by the Administrative Board. A Reemployed Individual who was previously a Participant may recommence participation on the date of reemployment, provided such Participant again enrolls in the Plan in accordance with the procedures established from time to time by the Administrative Board on or prior to that date.

ARTICLE IV

CONTRIBUTIONS AND ALLOCATIONS

4.1 Employer Contributions. Simultaneously with basic Employee contributions, the Employer will make a contribution out of current or accumulated earnings, subject to the limits of Section 4.5, to the Employer Contribution Account of each Participant. The Employer's contribution to the Employer Contribution Account of each Participant will equal 100% of the Participant's basic Employee contribution as described in Section 4.2(A) not in excess of 3% of Base Pay plus 50% of the Participant's basic Employee contribution as described in Section 4.2(A) in excess of 3% but not in excess of 6% of Base Pay.

In addition to the Employer Contributions made pursuant to the paragraph above, for each Plan Year the Employer will make a contribution out of current or accumulated earnings, subject to the limits of section 415 of the Code, to the Employer Contribution Account of each Participant equal to the difference, if any, between (1) 100% of the Participant's basic Employee contribution as described in Section 4.2(A) not in excess of 3% of Base Pay for such Plan Year, plus 50% of the Participant's basic Employee contribution as described in Section 4.2(A) in excess of 3% but not in excess of 6% of Base Pay for such Plan Year and (2) the Employer contributions made pursuant to the paragraph above for the Participant for such Plan Year.

If any Employer is prevented from making a contribution which it would otherwise have made because of inadequate current or accumulated earnings, then the amount of the contribution which such Employer was prevented from making shall be made for the benefit of the Employees of such Employer by the remaining Employers as follows:

- (A) if the Employer files a consolidated income tax return with the Company for that year for Federal income tax purposes, then such contribution shall be made in such proportions as the Company shall specify; or
- (B) if no consolidated income tax return is filed by the Employer with the Company for that year for Federal income tax purposes, each remaining Employer not so prevented from making a contribution shall make a supplemental contribution. Such contribution shall be in an amount equal to that portion of its total current income and accumulated earnings which the total contributions that one or more Employers were so prevented from making bears to the total current and accumulated earnings of all Employers having current or accumulated earnings. As used in the preceding sentence, total current income and accumulated earnings are calculated after deducting contributions which would have been without considering the additional supplemental contributions permitted by this Subsection (B).

If any Employer is prevented from making a contribution which it would otherwise have made, and part or all of such contribution is made by one or more other Employers, contributions so made shall be deductible for Federal income tax purposes by the Employer or Employers making

such contributions. For the purpose of determining amounts which may be carried forward and deducted in succeeding taxable years, contributions shall be deemed to have been made by the Employer(s) on behalf of whom such contributions were made.

Notwithstanding anything to the contrary herein, the Company's contributions are contingent upon the deductibility of such contributions under section 404 of the Code. To the extent that a deduction for contributions is disallowed, such contributions may be returned within one year after the date of disallowance. Finally, if Company contributions are made under a mistake of fact, such contributions may be returned to the Company within one year after the payment thereof.

4.1a Employer Safe Harbor Contributions. In addition to the Employer contributions made pursuant to Section 4.1, for each Plan Year, the Employer, in its discretion, may contribute to the Plan as a "safe harbor contribution" for such Plan Year out of current or accumulated earning the amounts necessary to cause the Plan to satisfy the restrictions set forth in Section 4.8 (with respect to certain restrictions on Employee contributions) and Section 4.9 (with respect to certain restrictions on Employer contributions). Amounts contributed in order to satisfy the restrictions set forth in Section 4.8 shall be considered "qualified matching contributions" (within the meaning of Treasury regulation Section 1.401(k)-1(g)(13) for purposes of such Section, and amounts contributed in order to satisfy the restrictions set forth in Section 4.9 shall be considered Employer contributions in accordance with the provisions of Section 4.1.

4.2 Employee Contributions.

- (A) Basic Contributions. As a condition of participation in the Plan, a Participant must elect to have a basic Employee contribution made on his behalf for a Plan Year by entering into a salary reduction agreement wherein the Participant elects to defer an integral percentage of from 1% to 6% of Base Pay for future pay periods and have the Employer contribute the amount so deferred to the Plan.
- (B) Additional Contributions. Each Participant who is having a basic Employee contribution of 6% of Base Pay made to the Plan on his behalf may also elect to have an additional contribution made on his behalf for a Plan Year by entering into a salary reduction agreement wherein the Participant elects to defer an integral percentage of from 1% to 10% of Base Pay for future pay periods and have the Employer contribute the amount so deferred to the Plan. Notwithstanding the foregoing, Highly Compensated Participants shall not be allowed to make additional Employee contributions if such contributions will adversely affect the restrictions under Section 4.8 of the Plan. The above restriction shall apply to Highly Compensated Participants, not heretofore considered to be highly compensated, as of the first day of the first pay period commencing after the determination that the Participants are considered to be highly compensated.
- (C) Procedures. Base Pay for a Plan Year not so deferred as provided in Subsections (A) and (B) above shall be received by each such Participant in Cash. The reduction in a Participant's Base Pay for a Plan Year pursuant to such election under a salary reduction agreement shall be effected by payroll deductions each period within such Plan Year following the effective date of such agreement and will change automatically to reflect changes in Base Pay.

A Participant's salary reduction agreement shall remain in force and effect for all periods following the date of its execution until changed or suspended or until such Participant terminates his employment.

A Participant who has elected to have Employee contributions made to the Plan on his behalf may change his Employee contribution election (within the limitations described in Subsections (A) and (B) above), effective as of any Change Date, by communicating his new deferral election to his Employer in the manner and within the time period prescribed by the Administrative Board.

A Participant may cancel his Employee contribution election, effective as any Change Date, by communicating such cancellation to his Employer in the manner and within the time period prescribed by the Administrative Board. A Participant who so cancels his deferral election may resume deferrals, effective as of any Change Date, by communicating his new Employee contribution election to his Employer in the manner and within the time period prescribed by the Administrative Board.

Employee contributions shall be automatically suspended during periods of unpaid military leave or during a period of unpaid authorized leave of absence granted pursuant to a nondiscriminatory leave of absence program established by the Employer, but such Employee contributions shall automatically resume upon the Employee's return to active work.

- (D) Basic and additional Employee contributions shall be contributed by the Employer to the Plan out of current or accumulated earnings as soon as administratively practicable each payroll period.
- (E) Prior Employee Contributions. Basic and additional Employee contributions made by Participants on an after-tax basis prior to June 29, 1991, shall no longer be permitted. Such Employee contributions shall be held, maintained and administered in each such Participant's Prior Employee Contribution Account in accordance with the provisions of the Plan.
- 4.3 Forfeitures. Notwithstanding any provisions to the contrary, if a Participant whose Nonforfeitable Interest is less than 100% of such individual's Employer Contribution Account terminates employment or withdraws from participation and receives a distribution from such account, a separate account shall be established for such Participant's interest in the Plan as of the time of the distribution. At any relevant time prior to incurring a Period of Severance equal to at least five consecutive One-Year Breaks in Service, such Participant's nonforfeitable portion of such separate account shall be determined in accordance with the following formula:

$$X ' P (AB + (R \times D)) - (R \times D).$$

For purposes of applying the formula: "X" is the nonforfeitable portion of such separate account at the relevant time; "P" is the Participant's Nonforfeitable Interest at the relevant time; "AB" is the balance of such separate account at the relevant time; "R" is the ratio of the balance

of such separate account at the relevant time to the balance immediately after the distribution; and "D" is the amount of the distribution. The foregoing shall be applied to a Participant's Employer Contribution Account without regard to the Participant's Employer Contribution Account Subaccount.

Upon incurring a Period of Severance equal to at least five consecutive One-Year Breaks in Service, the forfeitable portion of a terminated Participant's account shall be forfeited and such forfeiture shall be available to reduce future Employer contributions.

Any Participant who terminates employment with an Employer on or after February 1, 1989, shall forfeit the forfeitable portion of such Participant's account on the earlier of (A) the distribution of the entire nonforfeitable portion of such Participant's account or (B) upon incurring a Period of Severance equal to at least five consecutive One-Year Breaks in Service.

Notwithstanding the preceding paragraph, if any former Participant shall be reemployed by an Employer before a Period of Severance equal to five consecutive One-Year Breaks in Service, and such former Participant had received a distribution of the entire vested interest prior to reemployment, such Participant's forfeited account shall be reinstated only if the full amount distributed to such individual is repaid before the earlier of (A) five years after the first date on which the Participant is subsequently reemployed by the Employer, or (B) the close of the first period of five consecutive One-Year Breaks in Service commencing after the distribution. In the event the former Participant does repay the full amount distributed, the undistributed portion of the Participant's account must be restored in full, unadjusted by any gains or losses occurring subsequent to the distribution.

- 4.4 Investment and Allocation of Contributions. Contributions will be deposited promptly with the Trustee for investment as follows:
 - (A) Employer Contribution Account. Employer contributions shall be invested in, or held for investment in, the Stock Fund by the Trustee. The Administrative Board will make a monthly allocation of shares of Company Stock purchased with Employer contributions to each Participant's Employer Contribution Account. Notwithstanding the foregoing, upon attainment of age 60, a Participant may elect to have his Employer Contribution Account invested in, or held for investment in, any one or more of the Stock Fund or other Investment Funds, in such increments as the Administrative Board may prescribe.
 - (B) Employee Contribution Account. Basic and additional Employee contributions shall be allocated to a Participant's Employee Contribution Account and shall be invested in, or held for investment in, any one or more of the Stock Fund or other Investment Funds, in such increments as the Administrative Board may prescribe, pursuant to a Participant's designation.
 - (C) Prior Employee Contribution Account. Employee contributions made prior to June 29, 1991, and allocated to a Participant's Prior Employee Contribution

Account shall be invested in, or held for investment in, any one or more of the Stock Fund or other Investment Funds, in such increments as the Administrative Board may prescribe, pursuant to a Participant's designation.

- (D) Rollover Account. Rollover Contributions shall be allocated to a Participant's Rollover Account and shall be invested in, or held for investment in, any one or more of the Stock Fund or other Investment Funds, in such increments as the Administrative Board may prescribe, pursuant to a Participant's designation.
- (E) Investment Designations and Transfers. A Participant may change the investment for subsequent Employee contributions. Any such change shall be made as of the first date on which the New York Stock Exchange is open for business coinciding with or next following receipt by the Plan Administrator of a properly completed change request from the Participant in accordance with procedures established by the Administrative Board, and the frequency of such changes may be limited by the Administrative Board.

As of any Transfer Date, a Participant may transfer any part, in such increments as the Administrative Board may prescribe, or all of the amounts in such Participant's (1) Employee Contribution Account, (2) Prior Employee Contribution Account, (3) Rollover Account and (4) Employer Contribution Account following a Participant's 60th birthday among the separate investment funds.

If a separate investment fund is eliminated as an available investment option under the Plan, the American General Corporation Personnel Policy Committee upon recommendation by the Administrative Board shall designate an appropriate default separate investment fund for such eliminated separate investment fund in the event one or more substitute separate investment funds are not timely designated by a Participant.

- 4.5 Limitation on Allocation of Contributions and Forfeitures.
- (A) Contrary Plan provisions notwithstanding, the Annual Additions allocated to a Participant's accounts for any Limitation Year shall be limited to the Maximum Annual Additions for such Participant for such year.

If as a result of allocation of forfeitures, a reasonable error in estimating a Participant's Base Pay, or because of other limited facts and circumstances, the Annual Additions which would be credited to a Participant's accounts for a Limitation Year would nonetheless exceed the Maximum Annual Additions for such Participant for such year, the excess Annual Additions which, but for this Section 4.5, would have been allocated to such Participant's Employee Contribution Account shall, to the extent possible, first be reduced by returning to such Participant the additional Employee contributions which are considered in determining such Participant's Annual Additions. Next, excess Annual Additions in the form of Employer contributions and forfeitures which, but for this Section, would have been allocated to such Participant's Employer Contribution Account shall be allocated instead to a suspense account,

and shall be used to reduce Employer contributions in the same manner as a forfeiture. Any remaining excess Annual Additions in the form of basic Employee contributions which, but for this Section, would have been allocated to such Participant's Employee Contribution Account shall be returned to such Participant. If a suspense account is in existence at any time during a Limitation Year pursuant to this Subsection (A), it will not participate in allocations of the net income (or net loss) of the Trust Fund.

Employee contribution elections of affected Participants pursuant to Section 4.2 may be revised prospectively by the Administrative Board on a temporary basis to the extent necessary to meet such limitations in the manner described in Section 4.8(C).

For purposes of determining whether such Participant's Maximum Annual Additions exceed the limitations herein provided, all defined contribution plans of the Employer are to be treated as one defined contribution plan. In addition, all defined contribution plans of Controlled Entities shall be aggregated for this purpose. For purposes of this paragraph only, a "Controlled Entity" shall be determined by application of a 50% control standard in lieu of an 80% control standard.

In the case of a Participant who also participated in a defined benefit plan of the Employer or a Controlled Entity (as defined in the immediately preceding paragraph), the Employer shall reduce the Annual Additions credited to the Accounts of such Participant under this Plan pursuant to the provisions of this Section 4.5(A) to the extent necessary to prevent the limitation set forth in section 415(e) of the Code from being exceeded. Notwithstanding the foregoing, the provisions of this Paragraph shall apply only if such defined benefit plan does not provide for a reduction of benefits thereunder to ensure that the limitation set forth in section 415(e) of the Code is not exceeded. Further, this Paragraph shall not apply for Limitation Years beginning after December 31, 1999.

- (B) For purposes of this Section 4.5, the following terms and phrases shall have these respective meanings.
 - (1) "Limitation Year" shall mean the calendar year.
 - (2) "Remuneration" with respect to a Participant's Limitation Year shall mean:
 - (a) the Participant's wages, salaries, fees for professional services, and other amounts received for personal services actually rendered in the course of employment with an Employer maintaining the Plan (including, but not limited to, commissions paid to sales personnel, compensation for services on the basis of a percentage of profits, commissions on insurance premiums, tips, and bonuses);
 - (b) in the case of a Participant who is an employee within the meaning of section 401(c)(1) of the Code and the regulations thereunder, the

Participant's earned income (as described in section 401(c)(2) of the Code and the regulations thereunder);

- (c) for purposes of Subsection 4.5(B)(2)(a) and (b), earned income from sources outside the United States (as defined in section 911(b) of the Code), whether or not excludable from gross income under section 911 of the Code or deductible under section 913 of the Code;
- (d) amounts described in sections 104(a)(3), 105(a), and 105(h) of the Code, but only to the extent that these amounts are includable in the gross income of the Employee;
- (e) amounts described in section 105(d) of the Code, whether or not these amounts are excludable from the gross income of the Employee under that section; and
- (f) amounts paid or reimbursed by the Employer for moving expenses incurred by an Employee, but only to the extent that these amounts are not deductible by the Employee under section 217 of the Code.
- (3) Participant's Remuneration for a Limitation Year shall not include:
- (a) contributions made by the Employer to a plan of deferred compensation to the extent that, before application of the limitations imposed under section 415 of the Code with respect to such plan, the contributions are not includable in the gross income of the Participant for the year in which contributed (except as to elective deferrals to a qualified plan described in section 401(k) of the Code or to a tax-sheltered annuity plan described in section 403(b) of the Code, and except as to elective contributions to a deferred compensation plan described in section 457 of the Code or to a cafeteria plan described in section 125 of the Code), Employer contributions made on behalf of a Participant to a simplified employee pension plan described in section 408(k) of the Code to the extent deductible by the Employee under section 219(b)(7) of the Code, and distributions from a plan of deferred compensation other than an unfunded non-qualified plan;
- (b) amounts realized from the exercise of a non-qualified stock option, or when restricted stock (or property) either becomes freely transferable or is no longer subject to a substantial risk of forfeiture;
- (d) other amounts which receive special tax benefits, such as premiums for group term life insurance to the extent not includable in the gross income of the Participant.

- (4) "Annual Additions" of a Participant, for any Limitation Year, shall mean the total (a) of the Employer contributions and forfeitures allocated to such Employee's accounts under the Plan for such year and (b) the amount of such Participant's Employee contributions to the Plan (excluding any rollover contributions).
- (5) "Maximum Annual Additions" of a Participant for any Limitation Year shall mean the lesser of \$30,000 or 25%, of such Participant's Remuneration during such year; provided, however, that the \$30,000 limitation shall be adjusted automatically to reflect any cost-of-living adjustments authorized by the Code or Treasury Regulations.

4.6 Rollover Contributions.

- (a) Qualified Rollover Contributions may be made to the Plan by any Employee of amounts received by such Employee from certain individual retirement accounts or annuities or from an employees' trust described in section 401(a) of the Code, which is exempt from tax under section 501(a) of the Code, but only if any such Rollover Contribution is made pursuant to and in accordance with applicable provisions of the Code and Treasury regulations promulgated thereunder. A Rollover Contribution of amounts that are "eligible rollover distributions" within the meaning of section 402(f)(2)(A) of the Code may be made to the Plan irrespective of whether such eligible rollover distribution was paid to the Employee or paid to the Plan as a "direct" Rollover Contribution. A direct Rollover Contribution to the Plan may be effectuated only by wire transfer directed to the Trustee or by issuance of a check made payable to the Trustee, which is negotiable only by the Trustee and which identifies the Employee for whose benefit the Rollover Contribution is being made. Any Employee desiring to effect a Rollover Contribution to the Plan must complete a rollover request in accordance with the procedures established from time to time by the Administrative Board. The Administrative Board may require as a condition to accepting any Rollover Contribution that such Employee furnish any evidence that the Administrative Board in its discretion deems satisfactory to establish that the proposed Rollover Contribution is in fact eligible for rollover to the Plan and is made pursuant to and in accordance with applicable provisions of the Code and Treasury regulations. All Rollover Contributions to the Plan must be made in cash. A Rollover Contribution shall be credited to the Rollover Account of the Employee for whose benefit such Rollover Contribution is being made as soon as administratively feasible.
- (b) An Employee who has made a Rollover Contribution in accordance with this Section, but who has not otherwise become a Participant in the Plan in accordance with Article III, shall become a Participant coincident with such Rollover Contribution; provided, however, that such Participant shall not have a right to defer Base Pay or have Employer Contributions made on his behalf until he has otherwise satisfied the requirements imposed by Article III.
- 4.7 Voting, Tendering, and Exchanging of Company Stock. Notwithstanding any provisions herein to the contrary, the following provisions shall govern with respect to the voting, tendering, exchanging and other rights concerning Company Stock:

- (A) Notice. At the time of mailing to shareholders of the notice of any shareholders' meeting of the Company, or any notice of a tender or exchange offer for Company Stock or notice by the Company of any other action with respect to Company Stock, the Company shall use its reasonable best efforts to cause to be delivered to each Participant and former Participant (whose account has allocated to it any shares of Company Stock) such notices and informational statements as are furnished to the Company's shareholders in respect of the exercise of voting, tendering, exchanging, or other rights, together with forms by which the Participant or former Participant may instruct the Trustee, or revoke such instructions, with respect to the exercise of voting, tendering, exchanging, or other rights applicable to shares of Company Stock credited to such account.
- (B) Voting. Every Participant or former Participant (whose account has allocated to it any shares of Company Stock) shall have the right to direct the Trustee with respect to voting Company Stock allocated (or allocable) to such accounts, and the Trustee shall vote such allocated shares as directed. All of the shares of Company Stock for which no voting instructions are received shall be voted by the Trustee in a uniform manner as a single block in accordance with the instructions received with respect to a majority of such shares for which instructions are received.
- (C) Tendering and Exchanging. In the event of a tender or exchange offer, every Participant or former Participant (whose account has allocated to it any shares of Company Stock) shall have the right to direct the Trustee whether to accept or decline the offer with respect to Company Stock allocated (or allocable) to such accounts, and the Trustee shall take such actions as directed. All of the shares of Company Stock for which no instructions are received with respect to tender offer or exchange rights shall not be tendered or exchanged by the Trustee.
- 4.8 Restrictions on Employee Contributions.
- (A) In restriction of the Participants' Employee contribution elections provided in Section 4.2, the Employee contributions on behalf of any Participant for any calendar year shall not exceed \$7,000 (with such amount to be adjusted automatically to reflect any cost-of-living adjustments authorized by section 402(g)(5) of the Code), reduced by any "excess deferrals" from other plans allocated to the Plan by March 1 of the next following calendar year within the meaning of, and pursuant to the provisions of, section 402(g)(2) of the Code.
- (B) It is specifically provided that in each Plan Year the Plan shall comply with the alternative method of satisfying the 'actual deferral percentage' tests as set forth in section 401(k)(12) of the Code and applicable regulatory authority.
- 4.9 Restrictions on Employer Contributions. It is specifically provided that in each Plan Year the Plan shall comply with the alternative method of satisfying the 'actual contribution percentage' tests set forth in section 401(m) of the Code and applicable regulatory authority.

4.10 Excess Contributions.

- (A) Anything to the contrary herein notwithstanding, any Employee contributions to the Plan for a calendar year on behalf of a Participant in excess of the restrictions set forth in Section 4.8(A) shall be distributed to such Participant not later than April 15 of the next following calendar year. At the same time, any related Employer contributions shall be forfeited.
- (B) In coordinating excess contributions pursuant to this Section, such excess contributions shall be treated in the following order:
 - (1) first, excess Employee contributions described in Subsection (A) above not considered in determining related Employer contributions shall be distributed; and
 - (2) next, excess Employee contributions described in Subsection (A) above considered in determining related Employer contributions shall be distributed, and related Employer contributions shall be forfeited.
- (C) Any distribution or forfeiture of excess contributions pursuant to Subsections (A) or (B) of this Section shall be adjusted for income or loss allocated thereto in a manner consistent with applicable Treasury Regulations, rulings and notices, and such distribution (or forfeiture, if applicable) will include such income or be reduced by such loss. Any distribution or forfeiture of excess contributions pursuant to this Section shall be made in cash.

ARTICLE V

VALUATION OF ACCOUNTS

All amounts contributed to the Trust shall be invested as soon as administratively feasible following their receipt by the Trustee, and the balance of each Account shall reflect the result of daily pricing of the assets in which such Account is invested from the time of receipt by the Trustee until the time of distribution.

ARTICLE VI

VESTING

- 6.1 Normal Retirement: Notwithstanding the provisions of Section 6.4, a Participant shall have a Nonforfeitable Interest in such Participant's entire Employer Contribution Account if the Participant is actively employed (A) on or after age 65, (B) on the day immediately preceding such Participant's death, or (C) on separation from service of the Employer upon incurring a Total Disability. No forfeiture shall thereafter arise under Section 4.3.
- 6.2 Plan Termination, Partial Plan Termination, or Complete Plan Discontinuance of Employer Contributions. Notwithstanding any other provision of this Plan, in the event of a termination or partial termination of the Plan, or a complete discontinuance of Employer contributions under the Plan, all affected Participants shall have a Nonforfeitable Interest in their Employer Contribution Accounts determined as of the date of such event. The value of these accounts and their Employee Contribution Accounts and Prior Employee Contribution Accounts as of such date shall be determined in accordance with the provisions of the Plan.
- 6.3 Vesting of Employee Contributions. A Participant shall accrue a Nonforfeitable Interest in such Participant's Employee Contribution Account and Prior Employee Contribution Account at all times.
- 6.4 Vesting of Employer Contributions. A Participant shall accrue a Nonforfeitable Interest in such Participant's Employer Contribution Account at the rate of 2% per month up to a maximum of 100%, for each month of such Participant's Period of Service, commencing after the Participant's first year of service. If a Participant's Nonforfeitable Interest under any prior vesting schedule of the Plan is greater than the Nonforfeitable Interest calculated under the above schedule, such Participant shall retain that Nonforfeitable Interest, and if the Nonforfeitable Interest is less than 100%, it will increase at the rate of 2% per month. Notwithstanding the above, a Participant will have a 100% Nonforfeitable Interest in such Participant's Employer Contribution Account upon retirement under the American General Retirement Plan or any successor plan.

Each Participant with a Period of Service of five years or more with the Employer on the effective date of any amendment to the preceding paragraph may elect, within a reasonable period after the adoption of the amendment, to have such Participant's Nonforfeitable Interest computed under the Plan without regard to such amendment. The period during which the election may be made shall commence with the date the amendment is adopted and shall end on the later of:

- (A) 60 days after the restatement is adopted:
- (B) 60 days after the amendment becomes effective; or
- (C) 60 days after the Participant is issued written notice of the amendment by $% \left(1\right) =\left(1\right) \left(1\right) +\left(1\right) \left(1\right) \left(1\right) +\left(1\right) \left(1\right)$

the Employer or the Administrative Board.

Further notwithstanding the above, a Participant will have a 100% Nonforfeitable Interest in such Participant's Employer Contribution Account Subaccount at all times.

- 6.5 Vesting After a Period of Severance. No Period of Service after a Period of Severance equal to at least five consecutive One-Year Breaks in Service shall be taken into account in determining the nonforfeitable percentage in a Participant's Employer Contribution Account accrued up to any such Period of Severance.
- 6.6 Vesting Before a Period of Severance of Five Consecutive One-Year Breaks in Service. A Participant who separates from service of the Employer and received a distribution of the Nonforfeitable Interest (of less than 100%), in such Participant's Employer Contribution Account in accordance with Section 7.4, shall forfeit amounts that are forfeitable in accordance with Section 4.3.
- 6.7 Election to Make a Withdrawal. If a Participant elects to make a withdrawal pursuant to Section 7.6(B), (C), (D) or (E) while remaining employed by the Company, such distribution will be subject to the vesting provisions of Section 6.4. Upon incurring a Period of Severance equal to at least five consecutive One-Year Breaks in Service, that part of such individual's Employer Contribution Account which is not vested shall be forfeited, and such forfeiture shall be available to reduce future Employer Contributions.
- 6.8 Transfer Between Plans. If a Participant ceases to be eligible as an Employee under this Plan but immediately becomes an Employee for purposes of another thrift plan maintained by the Company or its subsidiaries ("Other Plan"), such change in employment status will not be deemed a termination of employment for purposes of this Plan.
- 6.9 Independent Life Transferred Employees. A Transferred Employee is an Employee who was an employee of Independent Life and Accident Insurance Company on the day preceding his employment by the Employer. For purposes of computing a Participant's Nonforfeitable Interest in this Plan, a Transferred Employee's Period of Service will include full and partial years of credited service under the Independent Life INVEST Plan plus, if applicable, service with the Company from date of transfer from Independent Life and Accident Insurance Company to April 1, 1997.
- A Participant's Employee Contribution Account, Prior Employee Contribution Account and Employer Contribution Account will be frozen, and no future Employee or Employer contributions will be made. However, the Participant will continue to accrue vesting in this Plan based on years of service credited in the Other Plan at the rate which would have been applicable had such individual continued participation in this Plan.

If an Employee has ceased to be a Participant under an Other Plan and immediately becomes a Participant under this Plan, such Employee will not be required to complete the one-year Period of Service or one year of "participation service" in Section 3.1 before participating in this Plan; and the Employee's Period of Service and Hours of Service under the Other Plan will be added to the Period of Service and Hours of Service under this Plan.

ARTICLE VII

BENEFITS

- 7.1 Normal Retirement. Upon separation from service on or after satisfying eligibility requirements for early, normal, or late retirement under the Company's retirement plan, other than by reason of death, a Participant shall be entitled to a benefit based on the combined balance of the Participant's Employee Contribution Account, Prior Employee Contribution Account and Employer Contribution Account distributed in a manner provided in Article IX.
- 7.2 Disability. In the event that a Participant separates from service of the Employer upon incurring a Total Disability before the Normal Retirement Date, such Participant shall be entitled to a benefit based on the combined balance of such individual's Employee Contribution Account, Prior Employee Contribution Account, and Employer Contribution Account distributed in the manner provided in Article IX.
- 7.3 Death. In the event of the death of a Participant prior to the commencement of a benefit described in Sections 7.1, 7.2, or 7.4, the Beneficiary shall be paid the combined balance in the Participant's Employee Contribution Account, Prior Employee Contribution Account and Employer Contribution Account distributed in the manner provided in Article IX.
- 7.4 Termination of Service. In the event a Participant separates from service of the Employer prior to the Normal Retirement Date for any reason other than death or Total Disability and Section 7.1 does not apply, the Nonforfeitable Interest in such Participant's Employer Contribution Account determined pursuant to Article VI and the balance of such Participant's Employee Contribution Account and Prior Employee Contribution Account shall be distributable to the Participant in accordance with Article IX.

7.5 Valuation Date to be Used.

- (A) If a Participant or Beneficiary becomes entitled to a benefit pursuant to Section 7.1, 7.2, 7.3, or 7.4, and elects to receive a distribution of such benefit as a result thereof, the value of the account balances to be distributed shall be determined as of the date of the event giving rise to the distribution. For purposes of Section 7.1, the event occasioning the benefit shall be the Participant's separation from service on or after satisfying the eligibility requirements for early, normal, or late retirement under the Company's retirement plan, other than by reason of death. For purposes of Section 7.2, the event occasioning the benefit shall be the Participant's separation from service of the Employer on account of Total Disability. For purposes of Section 7.3, the event occasioning the benefit shall be the death of the Participant prior to the commencement of a benefit described in Section 7.1, 7.2, or 7.4. For purposes of Section 7.4, the event occasioning the benefit shall be the date the Participant separated from service.
- (B) If pursuant to Section 8.4, a Participant elects to receive a benefit after an event set forth in Section 7.1, 7.2, 7.3, or 7.4, the value of the account balances to be distributed

30

shall be determined as of the Withdrawal Date on which the Plan Administrator receives the Participant's withdrawal request in accordance with the procedures established from time to time by the Administrative Board.

7.6 Withdrawals.

- (A) A Participant may withdraw from his Prior Employee Contribution Account any or all amounts held in such Account which have been so held for six months or more.
- (B) A Participant who has withdrawn all amounts in his Prior Employee Contribution Account but who has not made or had made on his behalf to the Plan basic Employee contributions for at least 60 cumulative months may withdraw from his Employer Contribution Account other than his Employer Contribution Account Subaccount any or all amounts held in such account which have been so held for 24 months or more, but not in excess of his Nonforfeitable Interest in the then value of such account. Any amounts in an Employer Contribution Account which is a Rollover Account must be withdrawn prior to amounts in any other Employer Contribution Account.
- (C) A Participant who has withdrawn all amounts in his Prior Employee Contribution Account and who has made or had made on his behalf to the Plan basic Employee contributions for at least 60 cumulative months may withdraw from his Employer Contribution Account other than his Employer Contribution Account Subaccount an amount not exceeding his Nonforfeitable Interest in the then value of such Account. Any amounts in an Employer Contribution Account which is a Rollover Account must be withdrawn prior to amounts in any other Employer Contribution Account.
- (D) A Participant who has attained age 59 1/2 and who has made all available withdrawals pursuant to Subsections (A), (B), and (C) above may withdraw from his Employer Contribution Account Subaccount an amount not exceeding the then value of such Account.
- (E) A Participant who has attained age 59-1/2 and who has made all available withdrawals pursuant to Subsections (A), (B), (C) and (D) above may withdraw from his Employee Contribution Account an amount not exceeding the then value of such Account.
- (F) A Participant who has a financial hardship, as determined by the withdrawal committee ("Withdrawal Committee") appointed by the Administrative Board, and who has made all available withdrawals pursuant to the Subsections above and pursuant to the provisions of any other plans of the Employer or any Controlled Entities of which he is a member and who has obtained all available loans pursuant to the provisions of any plans of the Employer and any Controlled Entities of which he is a member may withdraw from his Employee Contribution Account amounts not to exceed the lesser of (1) his Nonforfeitable Interest in the then value of such account or (2) the amount determined by the Withdrawal Committee as being available for withdrawal pursuant to this Subsection. No withdrawals on account of financial hardship will be permitted for amounts less than \$500. For purposes of this Subsection, financial hardship means the immediate and heavy financial

needs of the Participant. A withdrawal based upon financial hardship pursuant to this Subsection shall not exceed the amount required to meet the immediate financial need created by the hardship and not reasonably available from other resources of the Participant. The determination of the existence of a Participant's financial hardship and the amount required to be distributed to meet the need created by the hardship shall be made by the Withdrawal Committee based upon information requested from the Participant by the Withdrawal Committee and considering all relevant facts and circumstances. The amount of an immediate and heavy financial need may include any amounts necessary to pay any federal, state, or local income taxes or penalties reasonably anticipated to result from the distribution. In addition to any immediate and heavy financial need as may be determined by the Withdrawal Committee, a withdrawal shall be deemed to be made on account of an immediate and heavy financial need of a Participant if the withdrawal is on account of:

- (1) medical expenses described in section 213(d) of the Code incurred by the Participant, the Participant's spouse or any dependents of the Participant (as defined in section 152 of the Code) or necessary for those persons to obtain medical care described in section 213(d) of the Code, and not reimbursed by insurance;
- (2) costs directly related to the purchase (excluding mortgage payments) of a principal residence of the Participant;
- (3) payment of tuition and related educational fees for the next 12 months of post-secondary education of the Participant, or the Participant's spouse, children or dependents (as defined in section 152 of the Code); or
- (4) the need to prevent the eviction of the Participant from his principal residence or foreclosure on the mortgage of the Participant's principal residence.

The decision of the Withdrawal Committee shall be final and binding, provided that all Participants similarly situated shall be treated in a uniform and nondiscriminatory manner. The above notwithstanding, withdrawals under this Subsection from a Participant's Employee Contribution Account shall be limited to the sum of the Participant's Employee contributions to the Plan, less any previous withdrawals of such amounts. A Participant who makes a withdrawal under this Subsection may not again make elective contributions to the Plan or any other qualified or nonqualified plan of the Employer or any Controlled Entity for a period of 12 months following such withdrawal. Further, such Participant may not make elective contributions under the Plan or any other plan maintained by the Employer or any Controlled Entity for such Participant's taxable year immediately following the taxable year of the withdrawal in excess of the applicable limit set forth in Section 4.8(A) for such next taxable year less the amount of such Participant's elective contributions for the taxable year of the withdrawal.

(G) All withdrawals pursuant to this Section shall be made as of the Withdrawal Date on which the Plan Administrator receives a withdrawal request in accordance with the procedures established from time to time by the Administrative Board. All withdrawals shall be made pro rata from each fund in which such account is invested.

7.7 Loans.

- (A) Upon application by (1) any Participant who is an Employee or (2) any Participant no longer employed by the Employer, a beneficiary of a deceased Participant or an Alternate Payee under a Qualified Domestic Relations Order, as defined in Section 13.11, who retains a balance in his Accounts under the Plan and who is a party-in-interest, as that term is defined in section 3(14) of ERISA, as to the Plan (an individual who is eligible to apply for a loan under this Section being hereinafter referred to as a "Participant" for purposes of this Section), the loan committee ("Loan Committee") appointed by the Administrative Board may in its discretion direct the Trustee to make a loan or loans to such Participant. Such loans shall be made pursuant to the provisions of the Loan Committee's written loan procedure, which procedure is hereby incorporated by reference as a part of the Plan.
- (B) A loan to a Participant may not exceed 50% of the then value of such Participant's Nonforfeitable Interest in his Accounts.
- (C) Paragraph (B) above to the contrary notwithstanding, the amount of a loan made to a Participant under this Section shall not exceed an amount equal to the difference between:
 - (1) The lesser of \$50,000 (reduced by the excess, if any, of (A) the highest outstanding balance of loans from the Plan during the one-year period ending on the day before the date on which the loan is made over (B) the outstanding balance of loans from the Plan on the date on which the loan is made) or one-half of the present value of the Participant's total nonforfeitable accrued benefit under all qualified plans of the Employer or Controlled Entity; minus
 - (2) The total outstanding loan balance of the Participant under all other loans from all qualified plans of the Employer or a Controlled Entity.

ARTICLE VIII

COMMENCEMENT OF BENEFITS

8.1 Benefits After Normal Retirement Date. Payment shall be made within 60 days after the Participant separates from service of the Employer (including separation by reason of death or Total Disability) on or after the Participant's Normal Retirement Date.

Although full vesting is granted at age 65, distribution of the account of a Participant who continues active work past the Normal Retirement Date shall be postponed until such Participant's actual termination of employment or prior Plan withdrawal. Any such Participant may continue active participation in the Plan until the Participant's actual retirement date.

Notwithstanding the foregoing, subject to the provisions of Section 8.4, a Participant who terminates employment (for a reason other than death) may, if his total accounts (both vested and nonvested) exceed \$5,000, elect to defer payment of his benefit by making an appropriate election to defer with the Plan Administrator.

- 8.2 Certain Benefits Before Normal Retirement Date. Except as provided in Section 8.1, upon the death of the Participant, payment shall be made not later than 60 days after receipt by the Plan Administrator of proof of death. Except as provided in Section 8.1, upon separation from service of the Employer upon incurring a Total Disability, payment shall be made no later than 60 days after the determination that Total Disability exists.
- 8.3 Termination of Service Before Normal Retirement Date. Upon separation from service before Normal Retirement Date other than by reason of death or Total Disability, the benefit to which a Participant is entitled under Section 7.4 shall be paid no later than 60 days after the Participant's separation from service, and is to be paid in the form prescribed in Article IX.
- 8.4 Commencement of Benefits. Notwithstanding anything in Section 8.1, 8.2, or 8.3, other than an election to defer made by a Participant pursuant to Section 8.1, payments of benefits shall be made or shall commence no later than 60 days after the close of the Plan Year in which the Participant attains or would have attained age 65, or, if later, separates from service.

Notwithstanding anything in this Section or in Sections 8.1, 8.2, or 8.3 to the contrary, the entire Nonforfeitable Interest of each Participant shall be distributed to such Participant no later than April 1 of the calendar year following the later of (i) the calendar year in which the Participant attains age 70 1/2 or (ii) the calendar year in which the Participant terminates his employment with the Employer (provided, however, that clause (ii) of this sentence shall not apply prior to January 1, 1999, and shall not apply in the case of a Participant who is a 'five-percent owner' (as defined in section 416 of the Code) with respect to the Plan Year ending in the calendar year in which such Participant attains age 70 1/2. Notwithstanding the foregoing, in accordance with procedures adopted by the Administrative Board, (1) a Participant (who is not a 'five-percent owner' with respect to the calendar year in which the Participant attains age 70 1/2

who attains age 70 1/2 in calendar year 1996, 1997, or 1998, may elect to defer his payment as if such age were attained after calendar year 1998, and (2) a Participant who attains age 70 1/2 prior to calendar year 1997 and who has not terminated employment with the Employer may elect to stop distributions which have commenced as of a prior payment date.

If a Participant dies before the entire Nonforfeitable Interest has been distributed to such Participant, such interest shall be distributed, within five years of the Participant's death, to the designated Beneficiary.

No benefit shall be paid to a Participant who is under the age of 65, without such person's consent, if the total value (both vested and unvested) of such Participant's accounts exceeds \$5,000.

A Participant's commencement of benefits shall be in compliance with the provisions of section 401(a)(9) of the Code and applicable Treasury Regulations thereunder.

ARTICLE IX

DISTRIBUTION OF BENEFITS

- 9.1 Mode of Benefit Payments. Upon death, retirement, Total Disability, termination of employment for any other reason or withdrawal, amounts in the Participant's Employer Contribution Account (subject to the vesting provisions of Article VI), and amounts in the Participant's Employee Contribution Account, Prior Employee Contribution Account, and Rollover Account shall be distributed as follows:
 - (A) from the Stock Fund, either in Company Stock or in cash at the Participant's election;
 - (B) from the Investment Funds, in cash.

Notwithstanding the above and notwithstanding the following sentence, if a distribution would include fractional shares of Company Stock, the value of such fractional shares shall be paid in cash instead. Except in situations subject to Section 9.3, if an election is not received by the Plan Administrator prior to commencement of payment of benefits pursuant to Section 8.4, the distribution of the vested interest in the Participant's Employer Contribution Account and of any amounts in the Participant's Employee Contribution Account, Prior Employee Contribution Account, and Rollover Account which are held in the Stock Fund shall be made in Company Stock, and the distribution of all other amounts shall be in cash.

All assets held in the Participant's Accounts shall be valued as of the date prescribed in Section 7.5. No interest or other earnings shall be paid after the valuation date prescribed in Section 7.5. Distributions of benefits shall be made as soon as administratively feasible and in accordance with Article VIII.

- 9.2 Stock Certificates. Where certificates for shares of stock are distributed, they will be issued in the name of the Participant only. However, upon the death of a Participant, certificates will be issued in the name of the Beneficiary.
- 9.3 Automatic Distributions. A Participant whose total accounts (both vested and nonvested) are valued at \$5,000 or less must make an election authorized by Section 9.1 within 30 days following termination of employment, retirement, or Total Disability. A Beneficiary must make an election within 30 days following the Participant's death. If an election is not received by the Plan Administrator within the applicable time periods stated above, the distribution shall be made in cash.

Nothing in this Article shall be construed to permit or require distribution of assets in which the Participant does not have a Nonforfeitable Interest, as determined under Article VI. Likewise, this Article shall not be construed to permit or require a distribution without a Participant's consent if such distribution is prohibited by Section 8.4.

9.4 Unclaimed Benefits. In the case of a benefit payable on behalf of a Participant, if the Plan Administrator is unable to locate the Participant or Beneficiary to whom such benefit is payable, upon the Plan Administrator's determination thereof, such benefit shall be forfeited and shall be available to reduce future Employer contributions. Notwithstanding the foregoing, if subsequent to any such forfeiture the Participant or Beneficiary to whom such benefit is payable makes a valid claim for such benefit, such forfeited benefit shall be restored to the Plan in the manner provided in Section 4.3. No interest or other earnings shall be paid or accrued in regard to the restored benefit.

9.5 Direct Rollovers.

- (A) This Section 9.5 applies to distributions made on or after January 1, 1993. Notwithstanding any provision of the Plan to the contrary that would otherwise limit a distributee's election under this Section, a distributee may elect, at the time and in the manner prescribed by the Plan Administrator, to have any portion of an "eligible rollover distribution" paid directly to an "eligible retirement plan" specified by the "distributee" in a "direct rollover."
- (B) For purposes of this Section 9.5, the following terms and phrases shall have these respective meanings:
 - (1) "Eligible rollover distribution" is any distribution of all or any portion of the balance to the credit of the distributee, except that an eligible rollover distribution does not include: any distribution that is one of a series of substantially equal periodic payments (not less frequently than annually) made for the life (or life expectancy) of the distributee or the joint lives (or joint life expectancies) of the distributee and the distributee's designated beneficiary, or for a specified period of ten years or more; any distribution to the extent such distribution is required under section 401(a)(9) of the Code; and the portion of any distribution that is not includible in gross income (determined without regard to the exclusion for net unrealized appreciation with respect to employer securities). Further, a distribution pursuant to newly relettered Section 7.6(F) shall not constitute an Eligible Rollover Distribution to the extent provided in section 402(c)(4) of the Code and the interpretive authority thereunder.
 - (2) "Eligible retirement plan" is an individual retirement account described in section 408(a) of the Code, an individual retirement annuity described in section 408(b) of the Code, an annuity plan described in section 403(a) of the Code, or a qualified trust described in section 401(a) of the Code, that accepts the distributee's eligible rollover distribution. However, in the case of an eligible rollover distribution to the surviving spouse, an eligible retirement plan is an individual retirement account or individual retirement annuity.
 - (3) "Distributee" includes an employee or former employee. In addition, the employee's or former employee's surviving spouse and the employee's or former employee's spouse or former spouse who is the alternate payee under a qualified

domestic relations order, as defined in section 414(p) of the Code, are distributees with regard to the interest of the spouse or former spouse.

(4) "Direct rollover" is a payment by the plan to the eligible retirement plan specified by the distributee.

ARTICLE X

PLAN AMENDMENT AND TERMINATION

10.1 Amendment of the Plan. The Company shall have the right to amend or modify this Plan and (with the consent of the Trustee) the Trust agreement at any time, and from time to time, to any extent that it may deem advisable. Any such amendment or modification shall be set out in an instrument in writing, duly authorized by the Board of Directors and executed by the Company. No such amendment or modification shall, however, increase the duties or responsibilities of the Trustee without its consent thereto in writing, or have the effect of transferring to or vesting in any Employer any interest or ownership in any properties of the Trust Fund, or of permitting the same to be used for or diverted to purposes other than for the exclusive benefit of the Participants and their Beneficiaries. No such amendment shall decrease the accounts of any Participant or shall decrease any Participant's Nonforfeitable Interest in such account. Notwithstanding anything herein to the contrary, the Plan or the Trust agreement may be amended in such manner as may be required at any time to make it conform to the requirements of the Code or any United States statutes with respect to employee trusts, or any amendment thereto, or any regulations or rulings issued pursuant thereto, and no such Plan or Trust amendment shall be considered prejudicial to any then existing rights of any Participant or Beneficiary under the Plan.

For purposes of this Section and subject to regulations to be issued by the Secretary, a Plan amendment made after July 30, 1984, which has the effect of reducing an early retirement benefit or an optional form of benefit, with respect to benefits attributable to service before the amendment (even though unvested), shall be treated as reducing a Participant's account.

- 10.2 Communication of Amendments. All amendments, including one to terminate the Plan, shall be adopted in writing by the Company's Board of Directors. Any material modification of the Plan by amendment or termination shall be communicated to all interested parties and the Secretaries of Labor and the Treasury in the time and manner prescribed by law.
- 10.3 Termination or Discontinuance of Employer Contributions. Upon plan termination or discontinuance of Employer contributions under the Plan all account balances shall be valued in accordance with Section 6.2. The Trustee shall then, as soon as administratively feasible, pay each Participant and Beneficiary the entire interest in the Trust attributable to such Participant or Beneficiary in a lump sum, and shall pay any remaining amount to the Employer. In case of any Participant whose residence is unknown, the Plan Administrator shall notify such Participant at the last known address by certified mail with return receipt requested advising such Participant of a pending distribution.
- 10.4 Acceptance or Rejection of Amendment Employers. The Company shall promptly deliver to each other Employer any amendment to this Plan or the Trust agreement. Upon delivery to an Employer of an executed copy of an amendment properly authorized and adopted by the Company, the Plan as to such Employer shall be thereupon amended in accordance therewith.

10.5 Termination of the Plan by an Employer. An Employer may at any time, by adoption of a resolution, terminate the Plan with respect to the Employees of said Employer, and may direct and require the Trustee to liquidate the share of the Trust Fund allocable to its Employees or their Beneficiaries. If the Plan is terminated by fewer than all Employers, the Plan shall continue in effect for the Employees of the remaining Employers. In the event that an Employer shall cease to exist, the Plan shall be terminated with respect to the Employees of such Employer, unless a successor organization adopts and continues the Plan.

10.6 Notice of Terminating Employer. A terminating or withdrawing Employer shall give 90 days notice in writing of its intention to the Administrative Board, the Company, and the Trustee, unless a shorter notice is agreed to by the Company. If an Employer withdraws from this Plan and provides for a successor plan for its Employees, the Trust Fund assets of this Trust held on behalf of such Employer may be determined and transferred to a successor trust upon approval by the appropriate District Director of Internal Revenue.

ARTICLE XI

ADMINISTRATION

- 11.1 Named Fiduciaries. The named fiduciaries shall be the Plan Administrator and the Trustee.
 - 11.2 Appointment of the Trustee.
- (A) The Company shall designate the Trustee in a written statement filed with the Company's Board of Directors. The appointment of the Trustee shall become effective at such time as the Trustee and the Company execute a valid written trust which definitely and affirmatively precludes prohibited diversion.
- (B) The resignation of a Trustee shall be made in writing, submitted to the Company, and recorded in the minutes of the Board of Directors. The discharge of any person described in the preceding sentence shall be effectuated in writing by the Company and delivered to such person with the details thereof recorded in the minutes of the Company's Board of Directors. Appointment of a successor Trustee shall be carried out in the manner prescribed in Subsection (A).
- 11.3 Trustee's Powers and Duties. The powers and duties of the Trustee shall be to manage and control the funds of the Trust in accordance with the terms of the Trust agreement forming a part hereof.
- 11.4 Administrative Expenses. Except for fees relating to Participant loans and commissions on and other costs relating to acquisitions or disposition of securities, including any reasonable fees in connection with the Investment Funds, the Employer may pay the administrative expenses of the Plan and Trust, including the reasonable compensation of the Trustee and reimbursement of its reasonable expenses. To the extent not paid by the Employer, the administrative expenses of the Plan and Trust will be paid by the Trustee of the Trust
- 11.5 Plan Administrator's Powers and Duties. The Plan Administrator shall have the following powers and duties:
 - (A) to construe and interpret the provisions of the Plan;
 - (B) to decide all questions of eligibility for Plan participation and for the payment of benefits;
 - (C) to provide appropriate parties, including government agencies, with such returns, reports, schedules, descriptions, and individual statements as required by law within the times prescribed by law, to furnish to the Employer, upon request, copies of any or all such materials, and further, to make copies of such instruments, reports, and descriptions as are required by law available for examination by Participants (and such of

41

their Beneficiaries who are or may be entitled to benefits under the Plan) in such places and in such manner as required by law;

- (D) to obtain from the Employer, the Employees, and the Trustee such information as shall be necessary for the proper administration of the Plan;
- (E) to determine the amount, manner, and time of payment of benefits hereunder; $% \left(1\right) =\left(1\right) \left(1\right) \left$
- (F) subject to the approval of the Company only as to any additional expense, to appoint and retain such agents, counsel, and accountants for the purpose of properly administering the Plan and, when required to do so by law, to engage an independent Certified Public Accountant to annually prepare the audited financial statement of the Plan's operations;
- (G) to take all actions and to communicate to the Trustee in writing all necessary information to carry out the terms of the Plan and the Trust agreement;
- (H) to notify the Trustee in writing of the termination of the Plan or the complete discontinuance of Employer contributions,
- (I) to direct the Trustee to distribute assets of the Trust to each Participant and Beneficiary in accordance with Article IX of the Plan;
- (J) to furnish each recipient of a "qualifying rollover distribution," as defined by section 402 of the Code, with a written explanation of the Code provisions under which such distribution will not be subject to tax if transferred to an eligible retirement plan within 60 days after which the recipient received the distribution and, if applicable, the provisions concerning "Capital Gains Treatment for Portion of Lump Sum Distribution" and the provisions concerning "Tax on Lump Sum Distributions" provided by section 402 of the Code;
- (K) to establish, at a meeting duly called for such purpose, a funding policy and method consistent with the objectives of the Plan and the requirements of Title I of ERISA. The Administrative Board shall meet at least annually to review such funding policy and method; and
- (L) to do such other acts reasonably required to administer the Plan in accordance with its provisions or as may be provided for or required by law.

All actions of the Plan Administrator taken pursuant to this Section shall be presented in a meeting of the Administrative Board and recorded in the minutes thereof. An annual report shall be presented to the Board of Directors.

11.6 Named Fiduciary's Powers and Duties. It shall be the responsibility of the Plan Administrator to provide a notice in writing to any Participant or Beneficiary whose claim for

benefits under this Plan has been denied by the Plan Administrator, setting forth the specific reasons for such denial, and to afford such Participant or Beneficiary a reasonable opportunity for a full and fair review of that decision.

11.7 Allocation of Functions. Where more than one person serves as Plan Administrator, such persons may agree in writing to allocate among themselves the various powers and duties prescribed in Section 11.5, provided all such persons sign such agreement. A copy of any such agreement shall be promptly relayed to the Company.

ARTICLE XII

ADOPTION AND WITHDRAWAL

- 12.1 Procedure for Adoption. Any Controlled Entity may, with the approval of the Board of Directors, adopt and become an Employer under this Plan pursuant to appropriate written resolutions of the board of directors of such adopting Controlled Entity, by executing and delivering to the Company and the Trustee an adoptive instrument specifying the classification of its Employees who are to be eligible to participate in the Plan, and by agreeing to be bound as an Employer by all the terms of the Plan with respect to its eligible Employees. The adoptive instrument may contain such changes and variations in the terms of the Plan as may be acceptable to the Company. However, the sole, exclusive right of any other amendment of whatever kind or extent to the Plan or Trust are reserved by the Company. The adoption agreement shall become, as to such adopting organization and its Employees, a part of this Plan, as then or thereafter amended, and the related Trust. It shall not be necessary for the adopting organization to sign or execute the original or the amended Plan and Trust documents. The effective date of the Plan for any such adopting organization shall be that stated in the adoption agreement, and from and after such effective date such adopting organization shall assume all the rights, obligations, and liabilities of an Employer hereunder and under the Trust. The Company's administrative powers and control of the Plan, as provided in this instrument and the Trust, including the sole right of amendment, and of appointment and removal of the Administrative Board and the Trustee and their successors, shall not be diminished by reason of the participation of any such adopting organization in the Plan and Trust.
- 12.2 Effect of Adoption. The following special provisions shall apply to all Employees.
 - (A) An Employee shall be considered in service while regularly employed simultaneously or successively by one or more Employers.
 - (B) The transfer of an Employee from one Employer to another Employer shall not be deemed a termination of service.
- 12.3 Withdrawal. Any participating Employer by action of its board of directors or other governing authority and notice to the Company and the Trustee, may withdraw from the Plan and Trust at any time, without affecting other Employers not withdrawing, by complying with the provisions of the Plan and Trust. A withdrawing Employer may arrange for the continuation, by itself or its successor, of this Plan and Trust in separate forms for its own Employees, with such amendments, if any, as it may deem proper, and may arrange for continuation of the Plan and Trust by merger with an existing plan and trust and transfer of Trust assets. The Company may, in its absolute discretion, terminate an adopting Employer's participation at any time when in its judgment such adopting Employer fails or refuses to discharge its obligations under the Plan. If an Employer is no longer a Controlled Entity, the Company may at any time terminate such Employer's participation under this Plan.

ARTICLE XIII

MISCELLANEOUS

- 13.1 Merger of this Plan with Another Plan. In the event of any merger or consolidation of the Plan with, or transfer, in whole or in part, of the assets and liabilities of the Trust Fund to another trust fund held under any other plan of deferred compensation maintained or to be established for the benefit of all or some of the Participants of this Plan, the assets of the Trust Fund applicable to such Participants shall be transferred to the other trust fund only if:
 - (A) each Participant would (if either this Plan or the other plan then terminated) receive a benefit immediately after the merger, consolidation, or transfer which is equal to or greater than the benefit such Participant would have been entitled to receive immediately before the merger, consolidation, or transfer (if this Plan had then terminated);
 - (B) resolutions of the board of directors of the Employer under this Plan, or of any new or successor employer' of the affected Participants, shall authorize such transfer of assets; and, in the case of the new or successor employer of the affected Participants, its resolutions shall include an assumption of liabilities with respect to such Participants' inclusion in the new employer's plan; and
 - (C) such other plan and trust are qualified under sections 401(a) and 501(a) of the Code.
- 13.2 Assignment and Alienation of Benefits. Except as otherwise provided in Section 13.11, except as to certain judgments and settlements described in section 401(a)(13) of the Code, and except as otherwise provided under other applicable law, no right or interest of any kind in any benefit under this Plan shall be transferable or assignable by any Participant or by any beneficiary or be subject to anticipation, adjustment, alienation, encumbrance, garnishment, attachment, execution, or levy of any kind.
- 13.3 Communication to Employees. The Plan Administrator shall furnish to each Participant and each Beneficiary receiving benefits under the Plan a copy of a summary plan description and a summary of any material modifications thereof at the time and in the manner prescribed by law.
- 13.4 Number and Gender. The masculine pronoun shall include the feminine pronoun, and the singular number shall include the plural number, unless the context of the Plan requires otherwise.
- 13.5 Construction. The terms of the Plan shall be construed under the laws of the State of Texas except to the extent such laws are preempted by federal law.
- 13.6 Not a Contract of Employment. The adoption of this Plan by an Employer shall

not constitute a contract of employment between the Employer and any Employee.

- 13.7 Indemnity. The Company shall indemnify all those to whom it has delegated fiduciary duties against any and all claims, loss, damages, expense, and liability arising from their responsibilities in connection with the Plan, unless the same is determined to be due to gross negligence or willful misconduct. Plan benefits shall be provided only from the Trust Fund, and neither the Employer nor the Trustee guarantees or assumes any liability that the Trust Fund will at any time be sufficient therefor.
- 13.8 Change of Beneficiary. A Participant may change the Beneficiary designation under the Plan at any time by submitting a Change of Beneficiary request in accordance with the procedures established from time to time by the Administrative Board.
- 13.9 Applications to the Administrative Board. Correspondence shall be sent to such address as the Administrative Board shall designate from time to time or to the following address:

Administrative Board for the American General Employees'
Thrift and Incentive Plan
c/o Plan Administrator
2929 Allen Parkway
Houston, Texas 77019

- 13.10 Top-Heavy Rules.
- - (1) In general:
 - (a) Plans Not Required to be Aggregated. Except as provided in Subsection 13.10(A)(1)(b), the term "Top-Heavy Plan" means, with respect to any plan year, any defined contribution plan if, as of the Determination Date; the aggregate of the accounts of Key Employees under the plan exceeds 60% of the aggregate of the accounts of all Employees under such plan.
 - (b) Aggregated Plans. Each plan of an Employer required to be included in an Aggregation Group shall be treated as a Top-Heavy Plan if such group is a Top-Heavy Group.
 - (2) Aggregation. For purposes of this Section 13.10:
 - (a) Aggregation Group:
 - (i) Required Aggregation. The term "Aggregation

Group" means:

- each plan of the Employer in which a Key Employee is a Participant; and
- 2) each other plan of the Employer which enables any plan described in the preceding Subsection (I) to meet the requirements of section 401(a)(4) or 410 of the Code.
- (ii) Permissive Aggregation. The Employer may treat any plan not required to be included in an Aggregation Group under Subsection 13.10(A)(2)(a)(i) as being part of such group, if such group would continue to meet the requirements of sections 401(a)(4) and 410 of the Code with such plan being taken into account.
- (b) Top-Heavy Group. The term "Top-Heavy Group"
 means any Aggregation Group if:
 - (i) the sum (as of the Determination Date) of:
 - 1) the present value of the cumulative accrued benefits for Key Employees under all defined benefit plans included in such group; and
 - 2) the aggregate of the accounts of Key Employees under all defined contribution plans included in such group;
 - (ii) exceeds 60% of a similar sum determined for all Employees.

For purposes of making the foregoing determination, an Employee's account balance as of a Determination Date shall be valued as of the most recent date within the 12-month period prior to such Determination Date as of which the Trust Fund was valued, and the net income (or loss) thereof allocated to the Participants' accounts.

(3) Distributions During Last Five Years Taken into Account. For purposes of determining (a) the present value of the cumulative accrued benefit for any Employee, or (b) the amount of the account of any Employee, such present value or amount shall be increased by the aggregate distributions made with respect to such Employee under the plan during the five-year period ending on the Determination Date. The preceding sentence shall also apply to distributions under a terminated plan which, if it had not been terminated, would

- (4) Other Special Rules. For purposes of this Section 13.10:
 - (a) Rollover Contributions to a Plan Not Taken into Account. Except to the extent provided in Treasury Regulations, any rollover contribution (or similar transfer) initiated by the Employee and made after December 31, 1983 to a plan shall not be taken into account with respect to the transferee plan for purposes of determining whether such plan is a Top-Heavy Plan (or whether any Aggregation Group which includes such plan is a Top-Heavy Group).
 - (b) Benefits Not Taken into Account if Employee Ceases to be Key Employee. If any individual is a non-Key Employee with respect to any plan for any plan year, but such individual was a Key Employee with respect to such plan for any prior plan year, any accrued benefit for such Employee (and the account of such Employee) shall not be taken into account.
 - - (i) the last day of the preceding plan year;
 - (ii) in the case of the first plan year of any plan, the last day of such plan year.
 - (d) Years. To the extent provided in Treasury Regulations, this Section shall be applied on the basis of any year specified in such regulations in lieu of plan years.
 - (e) Benefits Not Taken into Account if Employee Not Employed for Last Five Years. If any individual has not received any compensation from any Employer maintaining the Plan (other than benefits under the Plan) at any time during the five-year period ending on the Determination Date, any accrued benefit for such individual (and the account of such individual) shall not be taken into account.
 - (f) For purposes of determining whether the Plan is Top Heavy, an individual is a "Key Employee" if such person is an individual described in section 416(i)(1) of the Code and the Treasury Regulations promulgated thereunder and, for purposes of applying such provisions, an individual's "compensation" (as such term is used in such section and such regulations) shall be deemed to be equal to such individual's

Remuneration, as defined in Subsection 4.5(B)(2).

Should the Plan become Top Heavy, the following provisions will apply:

(B) A Participant shall have a Nonforfeitable Interest in such Participant's Employer Contribution Account in accordance with Section 6.4 or the following table, whichever is greater:

Years of Service	Nonforfeitable Percentage
2	20
3	40
4	60
5	80
6	100

- (C) The Employer shall contribute to the Plan for such Plan Year on behalf of each Participant who is not a Key Employee and who has not terminated employment as of the last day of such Plan Year an amount equal to:
 - (1) the lesser of (a) 3%, of such Participant's Remuneration as described in Subsection 4.5(B)(2) for such Plan Year, or (b) a percent of such Participant's Remuneration as described in Subsection 4.5(B)(2) for such Plan Year equal to the greatest percent determined by dividing for each Key Employee the amount allocated to such Key Employee's accounts for such Plan Year, by such Key Employee's Remuneration not in excess of \$160,000 (adjusted automatically to reflect any amendments to section 401(a)(17) of the Code and any cost-of-living increases authorized by section 401(a)(17) of the Code) for such Plan Year; reduced by
 - (2) the amount allocated to such Participant's accounts for such Plan Year.

The minimum contribution required to be made for a Plan Year pursuant to this Subsection (C), for a Participant employed on the last day of such Plan Year, shall be made regardless of whether such Participant is otherwise ineligible to receive an allocation of the Employer's contributions for such Plan Year. Notwithstanding the foregoing, no contribution shall be made pursuant to this Subsection (C) for a Plan Year, with respect to a Participant who is a participant in a defined benefit plan sponsored by the Employer or a Controlled Entity, if such Participant accrued under such defined benefit plan sponsored by the Employer or a Controlled Entity (for the Plan Year of such plan ending with or within the Plan Year of this Plan) a benefit which is at least equal to the benefit described in section 416(c)(1) of the Code. Further

notwithstanding the foregoing, and to the extent permitted by applicable law and Treasury Regulations, no contribution shall be made pursuant to this Subsection (C) for a Plan Year, with respect to a Participant who is a participant in another defined contribution plan sponsored by the Employer or a Controlled Entity, if such Participant receives under such other defined contribution plan (for the Plan Year of such plan ending with or within the Plan Year of this Plan) a contribution which is equal to or greater than the minimum contribution required by section 416(c)(2) of the Code. Notwithstanding the foregoing, if the Plan is deemed to be Top Heavy for a Plan Year, the Employers' contribution for such Plan Year pursuant to this Subsection (C) shall be increased by substituting "4%" in lieu of "3%" in Subsection 13.10(C)(1) hereof to the extent that the Board of Directors determine to so increase such contribution to comply with the provisions of section 416(h)(2) of the Code. Plan provisions to the contrary notwithstanding, the portion of a Participant's Employer Contribution Account which is attributable to the minimum contributions required to be made for a Plan Year pursuant to this Subsection (C) shall not be subject to forfeiture by reason of such Participants' withdrawal of basic Employee contributions; provided, however, that such Participant is not a Key Employee.

- (D) The Base Pay of any Participant taken into account under the Plan shall not exceed \$200,000 (or such other amount as shall be prescribed by Treasury Regulations).
- (E) All other requirements of section 416 of the Code and Treasury Regulations thereunder shall be complied with.
- (F) If the Plan has been deemed to be Top Heavy for one or more Plan Years and thereafter ceases to be Top Heavy, the provisions of this Section 13.10 shall cease to apply to the Plan effective as of the Determination Date on which it is determined to no longer be Top Heavy. Notwithstanding the foregoing, the nonforfeitable percentage of each Participant who is a Participant on such Determination Date, or who has terminated employment but has not incurred a Period of Severance equal to at least five consecutive One-Year Breaks in Service as of such Determination Date, shall not be reduced and, with respect to each Participant who has completed five or more years of service with the Employer on such Determination Date, the nonforfeitable percentage of each such Participant shall continue to be determined in accordance with the schedule set forth in Subsection 13.10(C).
 - 13.11 Qualified Domestic Relations Orders.
- (A) The prohibitions of Section 13.2 shall not apply to an assignment or alienation of benefits, a distribution of benefits, or the creation or recognition of other rights on behalf of a spouse, former spouse, child, or other dependent (referred to herein as "Alternate Payee") of a Participant pursuant to any Qualified Domestic Relations Order.
- (B) A "Qualified Domestic Relations Order" shall mean any judgment, decree, or order (including approval of a property settlement agreement) which is made pursuant to a state domestic relations law (including a community property law) and which relates to the provision of child support, alimony payments, or marital property rights of an Alternate Payee.

- (C) To be qualified, such order must specify the following facts:
- (1) the name and the last known mailing address (if any) of the Participant, and the name and mailing address of each Alternate Payee covered by the order;
- (2) the amount or percentage of the Participant's benefits to be paid by the Plan to each such Alternate Payee, or the manner in which such amount or percentage is to be determined;
- $\hbox{(3)} \qquad \hbox{the number of payments or period to which such order} \\ \hbox{applies; and} \\$
 - (4) each Plan to which such order applies.
 - (D) Such order must not require a Plan to do the following:
- (1) provide increased benefits (determined on the basis of actuarial value);
- (2) pay benefits to an Alternate Payee which are required to be paid to another Alternate Payee under another order previously determined to be a Qualified Domestic Relations Order; or
- (3) provide any type or form of benefit, or any option, not otherwise provided under the Plan .
- (E) In the case of any payment before a Participant has separated from service, a domestic relations order shall not be treated as failing to meet the requirements of the preceding Subsection (3) solely because such order requires that payment of benefits be made to an Alternate Payee:
 - (1) prior to the Participant's "earliest retirement age" as such term is defined in section 414(p)(4)(B) of the Code;
 - (2) as if the Participant had retired on the date on which such payment is to begin under such order; and
 - (3) in any form in which such benefits may be paid under the Plan to the Participant.
- (F) Procedures. The Plan Administrator shall establish reasonable procedures to determine whether domestic relations orders constitute Qualified Domestic Relations Orders, meeting the requirements set out in Subsection 13.10(B) above, and procedures to administer distributions under such orders.

- (1) The Plan Administrator shall promptly notify the Participant and any Alternate Payee of the receipt of any domestic relations order and advise them of the Plan's procedures for determining the order's qualified status.
- (2) Within a reasonable time after receiving an order, the Plan Administrator must determine whether it is qualified and then promptly notify the Participant and Alternate Payee(s) of the decision.
- (3) If there is an issue as to whether an order is qualified, payments which would otherwise be paid under the order shall be deferred while this determination is being made (by the Plan Administrator, by a court of competent jurisdiction, or otherwise). The Plan Administrator shall segregate in a separate account in the Plan or in an escrow account the amounts which would have been payable to the Alternate Payee during such period if the order had been determined to be a Qualified Domestic Relations Order.
- (4) If within 18 months of deferral of payment the order (or modification thereof) is determined to be a Qualified Domestic Relations Order, the Plan Administrator shall pay the segregated amounts (plus any interest thereon) to the person or persons entitled thereto.
- (5) If within 18 months of deferral of payments it is determined that the order is not a Qualified Domestic Relations Order, or the issue as to whether such order is a Qualified Domestic Relations Order is not resolved, then the Plan Administrator shall pay the segregated amounts (plus any interest) to the person or persons who should have been entitled to such amounts if there had been no order. Any determination that an order is a Qualified Domestic Relations Order which is made after the close of the 18-month period shall be applied prospectively only. The Plan shall not be liable to the Alternate Payee for payments before the order is determined to be qualified.
- (6) The Plan Administrator may treat any domestic relations order entered into before January 1, 1985 as a Qualified Domestic Relations Order, even if it does not meet the requirements of Subsections 13.11(B)-(E). The Plan Administrator shall treat any domestic relations order entered into before January 1, 1985 as a Qualified Domestic Relations Order if payments are being made from the Plan pursuant to that order.
- (G) Forfeitures. Forfeitures are not permitted of amounts which are payable to an Alternate Payee under a Qualified Domestic Relations Order during any period in which the Alternate Payee cannot be located, unless full reinstatement is made when the Alternate Payee is located.
- (H) Death Benefits. The former spouse of a Participant shall be treated as a surviving spouse for purposes of Section 2.3 to the extent provided in any Qualified Domestic Relations Order.

13.12 Uniformed Services Employment and Reemployment Rights Act Requirements. Notwithstanding any provision of the Plan to the contrary, contributions, benefits and service credit with respect to qualified military service will be provided in accordance with section 414(u) of the Code.

IN WITNESS WHEREOF, the parties hereto have executed this American General Employees' Thrift and Incentive Plan, on this 28th day of June, 2001.

AMERICAN GENERAL CORPORATION

/S/ Gary D. Reddick

Gary D. Reddick Executive Vice President -Administration and Insurance Operations

Exhibit 4(b)

AMERICAN GENERAL

AGENTS' AND MANAGERS' THRIFT PLAN

July 1, 2001 Restatement (incorporating the January 1, 1997 Restatement and Amendments 1-6 thereto)

TABLE OF CONTENTS

PAGE

ARTICLE I - PUF	RPOSE1
ARTICLE II - DE	EFINITIONS2
2.1	Administrative Board2
2.2	Base Pay2
2.3	Beneficiary2
2.4	Board of Directors2
2.5	Certified Public Accountant3
2.6	Change Date3
2.7	Code3
2.8	Company
2.9	Company Stock3
2.10	Controlled Entity3
2.11	Effective Date
2.12	Employee3
2.13	Employee Contribution Account
2.14	Employer4
2.15	Employer Contribution Account4
2.16	Employment Commencement Date4
2.17	Highly Compensated Participant4
2.18	Hour of Service5
2.19	Investment Funds5
2.20	Leased Employee5
2.21	Long-Term Disability Program5
2.22	Nonforfeitable Interest5
2.23	Non-Highly Compensated Participant5
2.24	Normal Retirement Date6
2.25	One-Year Break in Service6
2.26	Participant6
2.27	Period of Separation6
2.28	Period of Service6
2.29	Period of Severance7
2.30	Plan8
2.31	Plan Administrator8
2.32	Plan Year8
2.33	Prior Employee Contribution Account8
2.34	Reemployed Individual8
2.35	Rollover Account9
2.36	Rollover Contributions9
2.37	Secretary9

	2.38	Stock Fund	. 9
	2.39	Total Disability	
	2.40	Transfer Date	
	2.41	Treasury Regulations	
	2.42	Trust	
	2.43	Trustee	
	2.44	Trust Fund	
	2.45	Withdrawal Date	٠.
ARTICLE		TICIPATION	
	3.1	Service	
	3.2	Reemployed Individuals	
	3.3	Participation	
	3.4	Independent Life Transferred Employee	16
		DEDUTTOUS AND ALLOCATIONS	
ARTICLE		RIBUTIONS AND ALLOCATIONS	
	4.1	Employer Contributions	
	4.1a	Employer Safe Harbor Contributions	
	4.2	Employee Contributions	
	4.3	Forfeitures	
	4.4	Investment and Allocation of Contributions	
	4.5	Limitation on Allocation of Contributions and Forfeitures	
	4.6	Rollover Contributions	
	4.7	Voting, Tendering, and Exchanging of Company Stock	
	4.8	Restrictions on Employee Contributions	
	4.9	Restrictions on Employer Contributions	
	4.10	Excess Contributions	20
	., .,	TTOU 05 4000UUT0	
ARTICLE	V - VALUA	TION OF ACCOUNTS	23
ARTICI E	VT - VEST	ING	24
ANTICLL	6.1	Normal Retirement	
	6.2	Plan Termination, Partial Plan Termination, or Complete Plan	
	0.2	Discontinuance of Employer Contributions	2/
	6.3	Vesting of Employee Contributions	2/
	6.4	Vesting of Employee Contributions	
	6.5		
	6.6	Vesting After a Period of Severance	۲۵
	0.0	One-Year Breaks in Service	2.5
	6.7	Election to Make a Withdrawal	
	6.8	Transfer Between Plans	
	6.9	Independent Life Transferred Employee	
	0.9	THEOREM I THE HAUSTELLED FUNDIOVER	/:

ARTICLE	VII - BEN 7.1 7.2 7.3 7.4 7.5 7.6 7.7	EFITS
ARTICLE	VIII - CO 8.1 8.2 8.3 8.4	MMENCEMENT OF BENEFITS
ARTICLE	IX - DIST 9.1 9.2 9.3 9.4 9.5	RIBUTION OF BENEFITS
ARTICLE	X - PLAN 10.1 10.2 10.3 10.4 10.5	AMENDMENT AND TERMINATION
ARTICLE	XI - ADMI 11.1 11.2 11.3 11.4 11.5 11.6 11.7	NISTRATION
ARTICLE	XII - ADO 12.1 12.2 12.3	PTION AND WITHDRAWAL41 Procedure for Adoption41 Effect of Adoption41 Withdrawal41

II - MISCELLANEOUS	ARTICLE XIII -
3.1 Merger of this Plan with Another Plan42	13.1
	13.2
	13.3
3.4 Number and Gender	13.4
3.5 Construction	13.5
3.6 Not a Contract of Employment43	13.6
3.7 Indemnity	13.7
3.8 Change of Beneficiary43	13.8
3.9 Applications to the Administrative Board43	13.9
	13.10
3.11 Qualified Domestic Relations Orders48	13.11
3.12 Uniformed Services Employment and Reemployment Rights Act Requirements56	13.12

ARTICLE I

PURPOSE

Effective July 1, 2001, American General Corporation adopts the restated Plan, as set forth herein, to replace the American General Agents' and Managers' Thrift Plan previously in effect.

The Plan was originally established by American General Insurance Company for its agents and managers effective August 1, 1968. The Plan was restated in 1976, 1987, 1990, 1991 and 1997. American General Insurance Company was reorganized as a general business corporation named American General Corporation effective July 1, 1980.

The Plan was instituted by the Company to encourage systematic savings and the accumulation of financial reserves by agents and managers of the Company and its subsidiaries, and to give agents and managers an opportunity to acquire an ownership interest in American General Corporation as well as enabling agents and managers to reap greater direct benefits from the Company's success. The Plan is carried out through agent and manager contributions deducted from the payroll on a pre-tax basis and contributions by the Company or its subsidiaries. Company contributions are invested in American General common stock purchased by a Trustee responsible for administering the Trust Fund. Agent and manager contributions are invested by the Trustee at each Participant's discretion in American General common stock or in other designated investment funds.

The Plan and Trust are intended to meet the requirements of sections 401(a) and 501(a) of the Internal Revenue Code of 1986, as amended, and the requirements of the Employee Retirement Income Security Act of 1974, as amended.

This Plan and the separate related Trust forming a part hereof are established and shall be maintained for the exclusive benefit of the eligible agents and managers of the Employers and their Beneficiaries. No part of the Trust Fund can ever revert to the Employers, except as provided in Article X, or be used for or diverted to purposes other than the exclusive benefit of the agents and managers of the Employers and their Beneficiaries.

ARTICLE II

DEFINITIONS

As used in this Plan, the following words and phrases shall have the meanings set forth below, unless the context clearly indicates otherwise.

- 2.1 Administrative Board. A board composed of at least three Company officers or Employees appointed by the Board of Directors to administer the Plan, located at Company headquarters at 2929 Allen Parkway, Houston, Texas 77019.
- 2.2 Base Pay. The compensation of the Employee as stated on Employer payroll records, such amount to exclude, however, any pay for overtime (which shall be deemed to refer as well to any shift differential payments), discretionary bonuses (which shall be deemed to refer as well to educational awards, educational reimbursements, instructor fees, referral awards, moving expenses, mortgage assistance, personal use of an Employer-owned vehicle, or workers' compensation payments), severance payments, and Employer contributions to this or any other deferred or noncash compensation program; provided, however, that any salary reduction amounts contributed on behalf of the Employee under a flexible benefits program pursuant to section 125 of the Code shall be included; and, provided, further, that salary reduction amounts contributed on behalf of the Employee under this Plan shall be included. Notwithstanding the foregoing, the Base Pay of any Employee taken into account under the Plan for any calendar year may not exceed \$150,000 (with such amount to be adjusted automatically to reflect any cost-of-living increases authorized by section 401(a)(17) of the Code).
 - 2.3 Beneficiary.
 - (A) The Beneficiary of a Participant shall be:
 - (1) the surviving spouse, if any, of the Participant; or
 - (2) if there is no surviving spouse, or if the surviving spouse has executed a consent in accordance with Subsection 2.3(B), the person or persons designated in writing by the Participant; or
 - (3) if there are no persons described in the preceding Subsections living at the date of the Participant's death, the Participant's estate.
- (B) The consent referred to in Subsection 2.3(A)(2) must be in writing, must acknowledge the effect of the Participant's designation of a person, other than the spouse, to receive the Participant's Nonforfeitable Interest upon the Participant's death, and must be witnessed by a Plan representative or a notary public.
 - 2.4 Board of Directors. The Board of Directors of the Company.

- (A) A person who is a certified public accountant, certified by a regulatory authority of a state;
- (B) a person who is a licensed public accountant, licensed by a regulatory authority of a state; or
- (C) a person certified by the Secretary of Labor as a qualified public accountant.
- ${\tt 2.6}$ Change Date. The first day of any pay period as designated by a Participant.
- $\,$ 2.7 Code. The Internal Revenue Code of 1986, as amended from time to time.
 - 2.8 Company. American General Corporation and its successors.
 - 2.9 Company Stock. The common stock of the Company.
- 2.10 Controlled Entity. A corporation or other trade or business which is not an Employer hereunder, but which, together with an Employer, is "under common control" within the meaning of section 1.414(c)-2 of the Treasury Regulations.
- 2.11 Effective Date. July 1,2001, as to this amendment and restatement of the Plan.
- 2.12 Employee. An individual performing services for an Employer as a Career Agent (sometimes known as a Sales Employee, Home Service Representative, Sales Representative, Field Representative, Agent or Home Service Agent, or Sales or District Manager), as a Full-Time Life Insurance Agent, or as any other field representative with a Flexi-Master Contract, provided that the holder of a Flexi-Master Contract is deemed to be an employee under section 7701(a)(20) of the Code, excluding, however, any such individual whose terms and conditions of employment are governed by a collective bargaining agreement, unless such agreement provides for his coverage under the Plan. Such term shall not include leased employees.
- 2.13 Employee Contribution Account. A separate account maintained for each Participant to which both basic and additional Employee contributions made on behalf of the Participant, and earnings and investment gains thereon, are credited, and which is invested in any of the separate investment funds.
- 2.14 Employer. The Company and any other corporation which shall adopt this Plan pursuant to Article XII, and the successor, if any, to such corporation.
- 2.15 Employer Contribution Account. A separate account maintained for a Participant consisting of cash, dividends payable, and Company Stock purchased by Employer contributions

- and reinvested dividends, and which may be invested in any of the Investment Funds after the Participant attains age sixty.
- $2.16\ {
 m Employment}\ {
 m Commencement}\ {
 m Date}.$ The date on which an Employee first performs an Hour of Service.
- 2.17 Highly Compensated Participant. Each Participant or Former Participant who is an Employee who performs services during the Plan Year for which the determination of who is highly compensated is being made (the 'Determination Year') and who:
 - (A) is a five-percent owner of the Employer (within the meaning of section 416(i)(1)(A)(iii) of the Code) at any time during the Determination Year or the twelve-month period immediately preceding the Determination Year (the 'Look-Back Year'); or
 - (B) For the Look-Back Year:
 - (i) receives compensation (within the meaning of section 414(q)(4) of the Code; 'compensation' for purposes of this Paragraph) in excess of \$80,000 (with such amount to be adjusted automatically to reflect any cost-of-living adjust adjustments authorized by section 414(q)(1) of the Code) during the Look-Back Year; and
 - (ii) if the Administrative Board elects the application of this clause for such Look-Back Year, is a member of the top 20% of Employees for the Look-Back Year (other than Employees described in section 414(q)(5) of the Code) ranked on the basis of compensation received during the year.

For purposes of the preceding sentence, (i) all employers aggregated with the Employer under section 414(b), (c), (m), or (o) of the Code shall be treated as a single employer, (ii) a former Employee who had a separation year (generally, the Determination Year such Employee separates from service) prior to the Determination Year and who was an active Highly Compensated Participant for either such separation year or any Determination Year ending on or after such Employee's fifty-fifth birthday shall be deemed to be a Highly Compensated Participant, and (ii) the Committee may elect, in accordance with the provisions of applicable Treasury regulations, rulings and notices, to make the Look-Back Year calculation for a Determination Year on the basis of the calendar year ending with or within the applicable Determination Year (or, in the case of a Determination Year that is shorter than twelve months, the calendar year ending with or within the twelve- month period ending with the end of the applicable Determination Year). To the extent that the provisions of this Paragraph are inconsistent or conflict

with the definition of a 'highly compensated employee' set forth in section 414(q) of the Code and the Treasury regulations thereunder, the relevant terms and provisions of section 414(q) of the Code and the Treasury regulations thereunder shall govern and control.

- 2.18 Hour of Service. An Hour of Service is each hour for which an Employee is directly or indirectly paid, or entitled to payment, by the Employer or a Controlled Entity for the performance of duties or for reasons other than the performance of duties. Such Hours of Service shall be credited to the Employee for the period in which such duties were performed or in which occurred the period during which no duties were performed. An Hour of Service also includes each hour, not credited above, for which back pay, irrespective of mitigation of damages, has been either awarded or agreed to by the Employer or a Controlled Entity. These Hours of Service shall be credited to the Employee for the period to which the award or agreement pertains rather than the period in which the award, agreement or payment is made. The number of Hours of Service to be credited to an Employee for any period shall be governed by section 2530.200b-2(b) and (c) of the Labor Department Regulations relating to the Employee Retirement Income Security Act of 1974, as amended.
- 2.19 Investment Funds. Investment funds made available from time to time for the investment of plan assets as described in Article IV.
- 2.20 Leased Employee. Each person who is not an employee of the Employer or a Controlled Entity but who performs services for the Employer or a Controlled Entity pursuant to an agreement (oral or written) between the Employer or a Controlled Entity and any leasing organization, provided that such person has performed such services for the Employer or a Controlled Entity or for related persons (within the meaning of section 144(a)(3) of the Code) on a substantially full-time basis for a period of at least one year and such services are performed under primary direction or control by the Employer or a Controlled Entity.
- 2.21 Long-Term Disability Program. The American General Long-Term Disability Plan for Employees, the American General Long-Term Disability Plan for Executives, the American General Life and Accident Sales Employees' Non-Occupational Disability Income Plan, and any workers' compensation plan, program, or fund sponsored, maintained, or paid into by the Company or the other Employers for the benefit of its Employees and Career Agents, whichever may be applicable to the Participant at the relevant time.
- 2.22 Nonforfeitable Interest. The unconditional and legally enforceable right to which the Participant or Beneficiary (whichever is applicable) is entitled in the Participant's entire Employee Contribution Account balance and entire Prior Employee Contribution Account balance and in the percentage of the Participant's Employer Contribution Account balance which has vested pursuant to Article VI.
- 2.23 Non-Highly Compensated Participant. "Non-Highly Compensated Participant" shall mean any Participant or former Participant who is not a Highly Compensated Participant.

- 2.24 Normal Retirement Date. The first day of the month coincident with or next following a Participant's 65th birthday.
 - 2.25 One-Year Break in Service.
- (A) A 12-consecutive-month Period of Severance during which an Employee does not perform at least one hour of service.
- (B) Solely for the purpose of determining whether or not a Participant has a One-Year Break in Service, the Plan shall treat as an hour of service each hour of service which would otherwise have been credited to the Participant but for an absence beginning on or after January 1, 1985, for one of the following reasons:
 - (1) pregnancy of the Participant;
 - (2) birth of a child of the Participant;
 - (3) placement of a child with the Participant in connection with an adoption proceeding; or
 - $\ensuremath{\text{(4)}}$ caring for the child immediately following such birth or placement.

If those hours cannot be determined, the Plan shall treat eight hours per day of such absence as hours of service. The hours to be treated as hours of service because of such absence shall not exceed 501.

The hours required to be credited under (1)-(4) of this Subsection 2.24(B) shall be credited in the year in which such absence begins if it is necessary to prevent a break in service in that year; otherwise such hours shall be credited to the succeeding year.

No credit shall be given under this Subsection 2.24(B) unless the Employee furnishes to the Committee such information as it may reasonably require to establish that the absence is for one of the reasons listed in (1)-(4) of this Subsection 2.24(B), and to establish the number of days of such absence.

- 2.26 Participant. An Employee who satisfies the eligibility requirements of Article III and has enrolled in the Plan.
- 2.27 Period of Separation. A period of time commencing with the date an Employee separates from service and ending with the date such Employee resumes employment with the Employer.
- 2.28 Period of Service. For purposes of determining an Employee's initial or continued eligibility to participate in the Plan or the vested interest in such individual's Employer Contribution

of:

Account, an Employee shall be credited for the time period commencing with his Employment Commencement Date, including years of service with the Employer in a category of Employees excluded from the Plan, and ending on the date his Period of Severance begins. A Period of Service for these purposes includes a Period of Separation of less than 12 consecutive months. In the case of an Employee who separates from service and later resumes employment with the Employer, the Period of Service prior to resumption of employment shall be aggregated only if such Employee is a Reemployed Individual.

In the case of a person who becomes an Employee, after being a "leased employee" under section 414(n) of the Code, the Employment Commencement Date will be deemed to be the date such Employee first performed services for the Employer as a "leased employee" or such other date as required by Treasury Regulations.

- (A) Periods of Service prior to September 30, 1991, which would not have been credited under the rules of the previous plan or its amendments or a predecessor plan. Such periods shall be determined according to provisions of the previous plan or predecessor plan, including rules which require minimum hours or other length of service or contributions by Participants;
- (B) Periods of Service prior to the date the Employer adopts this Plan or a predecessor plan; and
- (C) Periods of Service in which an Employee declined to contribute to the Plan or a predecessor plan prior to the date such Employee first commences participation in the Plan.
- It is specifically provided that for purposes of determining Period of Service, service with a Controlled Entity shall be included.
 - 2.29 Period of Severance. A period of time commencing with the earlier
 - (A) the date an Employee separates from service by reason of quitting, retirement, death, or discharge; or
 - (B) the date 12 months after the date an Employee separates from service, including severance for reasons other than quitting, retirement, death, or discharge; and ending, in the case of an Employee who separates from service by reason other than death, with the date such Employee resumes employment with the Employer.

(C) An absence beginning on or after January 1, 1985, because

of:

- (1) pregnancy of the Participant;
- (2) birth of a child of the Participant;
- (3) placement of a child with the Participant in connection with an adoption proceeding; or
- (4) caring for the child immediately following such birth or placement;

shall not be considered a Period of Severance except as permitted by Treasury Regulations.

- (D) In order for the preceding Subsection 2.28(C) to apply, a Participant must furnish to the Administrative Board such information as it may reasonably require to establish that the absence is for one of the reasons listed in (1)-(4) of the preceding Subsection 2.28(C), and to establish the number of days of such absence.
- $2.30\ \text{Plan}.$ This document, the provisions of the Trust agreement, and any amendments to either.
- 2.31 Plan Administrator. The Administrative Board unless and until it designates such other person or persons. A member of the Administrative Board may resign or may be replaced by the Board of Directors at any time. No bond or other security shall be required of any member of the Administrative Board unless otherwise required by law.
- 2.32 Plan Year. The 12-month period beginning January 1 and ending December 31.
- 2.33 Prior Employee Contribution Account. A separate account maintained for each Participant who made Employee contributions to the Plan prior to September 30, 1991, to which such Employee contributions, and earnings and investment gains thereon, are credited, and which is invested in any of the separate investment funds.
- 2.34 Reemployed Individual. A person who, after having separated from service, resumes employment:
 - (A) with a Nonforfeitable Interest in such individual's Employer Contribution Account; or
 - (B) with no such Nonforfeitable Interest, and who resumes employment either (1) before a One-Year Break in Service or (2) after a One-Year Break in Service but before the latest Period of Severance equals or exceeds the greater of (a) such individual's Period of Service or (b) five consecutive years of a Period of Service.

- 2.35 Rollover Account. An separate account maintained for each Participant who has made Rollover Contributions pursuant to Section 4.6 to which such Rollover Contributions and earnings and investment gains thereon, are credited, and which is invested any of the separate Investment Funds.
 - 2.36 Contributions made by an Employee pursuant to Section 4.6.
 - 2.37 Secretary. The Secretary of the Treasury of the United States.
- 2.38 Stock Fund. The separate investment fund forming a part of the Trust which is invested in, or held for investment in, Company Stock. Dividends applicable to Company Stock will be applied to the purchase of additional Company Stock.
- 2.39 Total Disability. The definition of disability as used under the Company's Long Term Disability Program as in effect at the inception of the Participant's disability, respecting the applicable plan, program, or fund pursuant to which the Participant is entitled to receive disability benefits.
- 2.40 Transfer Date. The first date on which the New York Stock Exchange is open for business coinciding with or next following receipt by the Plan Administrator of a transfer request from the Participant in accordance with the procedures established from time to time by the Administrative Board.
- 2.41 Treasury Regulations. Regulations issued from time to time by the Secretary or the Secretary's delegate.
- 2.42 Trust. The agreement between the Trustee and the Employer entered into for the purpose of holding, managing, and administering all property held by the Trustee for the exclusive benefit of the Participants and their Beneficiaries.
- 2.43 Trustee. The person designated by the Company pursuant to Section 11.2 and in accordance with the Trust agreement, and any successor who is appointed pursuant to the terms of that Section.
- $2.44\ \mbox{Trust}$ Fund. All assets held by the Trustee pursuant to the terms of this Plan.
- 2.45 Withdrawal Date. The first date on which the New York Stock Exchange is open for business coinciding with or next following receipt by the Plan Administrator of a withdrawal request from the Participant in accordance with the procedures established from time to time by the Administrative Board.

ARTICLE III

PARTICIPATION

- 3.1 Eligibility. An Employee shall be eligible to participate in this Plan upon the lapse of thirty days from the Employee's Employment Commencement Date.
- 3.2 Reemployed Individuals. A Reemployed Individual is eligible to participate in the Plan on the later of the date of reemployment by the Employer or the date described in Section 3.1.
- 3.3 Participation. Participation in the Plan shall commence as soon as administratively feasible following the date the eligibility requirement is satisfied, provided the Administrative Board has received a properly completed enrollment in accordance with the procedures established from time to time by the Administrative Board. A Reemployed Individual who was previously a Participant may recommence participation on the date of reemployment, provided such Participant again enrolls in the Plan in accordance with the procedures established from time to time by the Administrative Board on or prior to that date.
- 3.4 Independent Life Transferred Employee. A Transferred Employee is an Employee who was an employee of Independent Life and Accident Insurance Company on the day preceding his employment by the Employer. For purposes of meeting the one-year Period of Service requirement for participation in this Plan, a Transferred Employee will receive credit for full and partial years of credited service under the American General Invest Plan (formerly the Independent Life INVEST Plan).

ARTICLE IV

CONTRIBUTIONS AND ALLOCATIONS

4.1 Employer Contributions. Simultaneously with basic Employee contributions, the Employer will make a contribution out of current or accumulated earnings, subject to the limits of Section 4.5, to the Employer Contribution Account of each Participant. The Employer's contribution to the Employer Contribution Account of each Participant will equal 331/3% of the Participant's basic Employee contribution as described in Section 4.2(A).

If any Employer is prevented from making a contribution which it would otherwise have made because of inadequate current or accumulated earnings, then the amount of the contribution which such Employer was prevented from making shall be made for the benefit of the Employees of such Employer by the remaining Employers as follows:

- (A) if the Employer files a consolidated income tax return with the Company for that year for Federal income tax purposes, then such contribution shall be made in such proportions as the Company shall specify; or
- (B) if no consolidated income tax return is filed by the Employer with the Company for that year for Federal income tax purposes, each remaining Employer not so prevented from making a contribution shall make a supplemental contribution. Such contribution shall be in an amount equal to that portion of its total current income and accumulated earnings which the total contributions that one or more Employers were so prevented from making bears to the total current and accumulated earnings of all Employers having current or accumulated earnings. As used in the preceding sentence, total current income and accumulated earnings are calculated after deducting contributions which would have been without considering the additional supplemental contributions permitted by this Subsection (B).

If any Employer is prevented from making a contribution which it would otherwise have made, and part or all of such contribution is made by one or more other Employers, contributions so made shall be deductible for Federal income tax purposes by the Employer or Employers making such contributions. For the purpose of determining amounts which may be carried forward and deducted in succeeding taxable years, contributions shall be deemed to have been made by the Employer(s) on behalf of whom such contributions were made.

Notwithstanding anything to the contrary herein, the Company's contributions are contingent upon the deductibility of such contributions under section 404 of the Code. To the extent that a deduction for contributions is disallowed, such contributions may be returned within one year after the date of disallowance. Finally, if Company contributions are made under a mistake of fact, such contributions may be returned to the Company within one year after the payment thereof.

4.1a Employer Safe Harbor Contributions. In addition to the Employer contributions made pursuant to Section 4.1, for each Plan Year, the Employer, in its discretion, may contribute to the Plan as a "safe harbor contribution' for such Plan Year out of current or accumulated earning the amounts necessary to cause the Plan to satisfy the restrictions set forth in Section 4.8 (with respect to certain restrictions on Employee contributions) and Section 4.9 (with respect to certain restrictions on Employer contributions). Amounts contributed in order to satisfy the restrictions set forth in Section 4.8 shall be considered "qualified matching contributions" (within the meaning of Treasury regulation Section 1.401(k)-1(g)(13) for purposes of such Section, and amounts contributed in order to satisfy the restrictions set forth in Section 4.9 shall be considered Employer contributions in accordance with the provisions of Section 4.1.

4.2 Employee Contributions.

- (A) Basic Contributions. As a condition of participation in the Plan, a Participant must elect to have a basic Employee contribution made on his behalf for a Plan Year by entering into a salary reduction agreement wherein the Participant elects to defer 3% of Base Pay for future pay periods and have the Employer contribute the amount so deferred to the Plan.
- (B) Additional Contributions. Each Participant who is having a basic Employee contribution made to the Plan on his behalf may also elect to have an additional contribution made on his behalf for a Plan Year by entering into a salary reduction agreement wherein the Participant elects to defer an integral percentage of from 1% to 13% of Base Pay for future pay periods and have the Employer contribute the amount so deferred to the Plan. Notwithstanding the foregoing, Highly Compensated Participants shall not be allowed to make additional Employee contributions if such contributions will adversely affect the restrictions under Section 4.8 of the Plan. The above restriction shall apply to Highly Compensated Participants, not heretofore considered to be highly compensated, as of the first day of the first pay period commencing after the determination that the Participants are considered to be highly compensated.
- (C) Procedures. Base Pay for a Plan Year not so deferred as provided in Subsections (A) and (B) above shall be received by each such Participant in Cash. The reduction in a Participant's Base Pay for a Plan Year pursuant to such election under a salary reduction agreement shall be effected by payroll deductions each period within such Plan Year following the effective date of such agreement and will change automatically to reflect changes in Base Pay.

A Participant's salary reduction agreement shall remain in force and effect for all periods following the date of its execution until changed or suspended or until such Participant terminates his employment.

A Participant who has elected to have Employee contributions made to the Plan on his behalf may change his Employee contribution election (within the limitations described in Subsections (A) and (B) above), effective as of any Change Date, by communicating his new deferral election to his Employer in the manner and within the time period prescribed by the Administrative Board.

A Participant may cancel his Employee contribution election, effective as any Change Date, by communicating such cancellation to his Employer in the manner and within the time period prescribed by the Administrative Board. A Participant who so cancels his deferral election may resume deferrals, effective as of any Change Date, by communicating his new Employee contribution election to his Employer in the manner and within the time period prescribed by the Administrative Board.

Employee contributions shall be automatically suspended during periods of unpaid military leave or during a period of unpaid authorized leave of absence granted pursuant to a nondiscriminatory leave of absence program established by the Employer, but such Employee contributions shall automatically resume upon the Employee's return to active work.

- (D) Basic and additional Employee contributions shall be contributed by the Employer to the Plan out of current or accumulated earnings as soon as administratively practicable each payroll period.
- (E) Prior Employee Contributions. Basic and additional Employee contributions made by Participants on an after-tax basis prior to September 30, 1991, shall no longer be permitted. Such Employee contributions shall be held, maintained and administered in each such Participant's Prior Employee Contribution Account in accordance with the provisions of the Plan.
- 4.3 Forfeitures. Notwithstanding any provisions to the contrary, if a Participant whose Nonforfeitable Interest is less than 100% of such individual's Employer Contribution Account terminates employment or withdraws from participation and receives a distribution from such account, a separate account shall be established for such Participant's interest in the Plan as of the time of the distribution. At any relevant time prior to incurring a Period of Severance equal to at least five consecutive One-Year Breaks in Service, such Participant's nonforfeitable portion of such separate account shall be determined in accordance with the following formula:

$$X = P (AB + (R \times D)) - (R \times D).$$

For purposes of applying the formula: "X" is the nonforfeitable portion of such separate account at the relevant time; "P" is the Participant's Nonforfeitable Interest at the relevant time; "AB" is the balance of such separate account at the relevant time; "R" is the ratio of the balance of such separate account at the relevant time to the balance immediately after the distribution; and "D" is the amount of the distribution.

Upon incurring a Period of Severance equal to at least five consecutive One-Year Breaks in Service, the forfeitable portion of a terminated Participant's account shall be forfeited and such forfeiture shall be available to reduce future Employer contributions.

Any Participant who terminates employment with an Employer on or after December 1, 1988, shall forfeit the forfeitable portion of such Participant's account on the earlier of (A) the

distribution of the entire nonforfeitable portion of such Participant's account or (B) upon incurring a Period of Severance equal to at least five consecutive One-Year Breaks in Service.

Notwithstanding the preceding paragraph, if any former Participant shall be reemployed by an Employer before a Period of Severance equal to five consecutive One-Year Breaks in Service, and such former Participant had received a distribution of the entire vested interest prior to reemployment, such Participant's forfeited account shall be reinstated only if the full amount distributed to such individual is repaid before the earlier of (A) five years after the first date on which the Participant is subsequently reemployed by the Employer, or (B) the close of the first period of five consecutive One-Year Breaks in Service commencing after the distribution. In the event the former Participant does repay the full amount distributed, the undistributed portion of the Participant's account must be restored in full, unadjusted by any gains or losses occurring subsequent to the distribution.

- 4.4 Investment and Allocation of Contributions. Contributions will be deposited promptly with the Trustee for investment as follows:
 - (A) Employer Contribution Account. Employer contributions shall be invested in, or held for investment in, the Stock Fund by the Trustee. The Administrative Board will make a monthly allocation of shares of Company Stock purchased with Employer contributions to each Participant's Employer Contribution Account. Notwithstanding the foregoing, upon attainment of age 60, a Participant may elect to have his Employer Contribution Account invested in, or held for investment in, any one or more of the Stock Fund or other Investment Funds, in such increments as the Administrative Board may prescribe.
 - (B) Employee Contribution Account. Basic and additional Employee contributions shall be allocated to a Participant's Employee Contribution Account and shall be invested in, or held for investment in, any one or more of the Stock Fund or other Investment Funds, in such increments as the Administrative Board may prescribe, pursuant to a Participant's designation.
 - (C) Prior Employee Contribution Account. Employee contributions made prior to June 29, 1991, and allocated to a Participant's Prior Employee Contribution Account shall be invested in, or held for investment in, any one or more of the Stock Fund or other Investment Funds, in such increments as the Administrative Board may prescribe, pursuant to a Participant's designation.
 - (D) Rollover Account. Rollover Contributions shall be allocated to a Participant's Rollover Contribution Account and shall be invested in, or held for investment in, any one or more of the Stock Fund or other Investment Funds, in such increments as the Administrative Board may prescribe, pursuant to a Participant's designation.

(E) Investment Designations and Transfers. A Participant may change the investment for subsequent Employee contributions. Any such change shall be made as of the first date on which the New York Stock Exchange is open for business coinciding with or next following receipt by the Plan Administrator of a properly completed change request from the Participant in accordance with procedures established by the Administrative Board, and the frequency of such changes may be limited by the Administrative Board.

As of any Transfer Date, a Participant may transfer any part, in such increments as the Administrative Board may prescribe, or all of the amounts in such Participant's (1) Employee Contribution Account, (2) Prior Employee Contribution Account, (3) Rollover Contribution Account and (4) Employer Contribution Account following a Participant's 60th birthday among the separate investment funds.

If a separate investment fund is eliminated as an available investment option under the Plan, the American General Corporation Personnel Policy Committee upon recommendation by the Administrative Board shall designate an appropriate default separate investment fund for such eliminated separate investment fund in the event one or more substitute separate investment funds are not timely designated by a Participant.

4.5 Limitation on Allocation of Contributions and Forfeitures.

(A) Contrary Plan provisions notwithstanding, the Annual Additions allocated to a Participant's accounts for any Limitation Year shall be limited to the Maximum Annual Additions for such Participant for such year.

If as a result of allocation of forfeitures, a reasonable error in estimating a Participant's Base Pay, or because of other limited facts and circumstances, the Annual Additions which would be credited to a Participant's accounts for a Limitation Year would nonetheless exceed the Maximum Annual Additions for such Participant for such year, the excess Annual Additions which, but for this Section 4.5, would have been allocated to such Participant's Employee Contribution Account shall, to the extent possible, first be reduced by returning to such Participant the additional Employee contributions which are considered in determining such Participant's Annual Additions. Next, excess Annual Additions in the form of Employer contributions and forfeitures which, but for this Section, would have been allocated to such Participant's Employer Contribution Account shall be allocated instead to a suspense account, and shall be used to reduce Employer contributions in the same manner as a forfeiture. Any remaining excess Annual Additions in the form of basic Employee contributions which, but for this Section, would have been allocated to such Participant's Employee Contribution Account shall be returned to such Participant. If a suspense account is in existence at any time during a Limitation Year pursuant to this Subsection (A), it will not participate in allocations of the net income (or net loss) of the Trust Fund.

Employee contribution elections of affected Participants pursuant to Section 4.2 may be revised prospectively by the Administrative Board on a temporary basis to the extent necessary to meet such limitations in the manner described in Section 4.8(C).

For purposes of determining whether such Participant's Maximum Annual Additions exceed the limitations herein provided, all defined contribution plans of the Employer are to be treated as one defined contribution plan. In addition, all defined contribution plans of Controlled Entities shall be aggregated for this purpose. For purposes of this paragraph only, a "Controlled Entity" shall be determined by application of a 50% control standard in lieu of an 80% control standard.

In the case of a Participant who also participated in a defined benefit plan of the Employer or a Controlled Entity (as defined in the immediately preceding paragraph), the Employer shall reduce the Annual Additions credited to the Accounts of such Participant under this Plan pursuant to the provisions of this Section 4.5(A) to the extent necessary to prevent the limitation set forth in section 415(e) of the Code from being exceeded. Notwithstanding the foregoing, the provisions of this Paragraph shall apply only if such defined benefit plan does not provide for a reduction of benefits thereunder to ensure that the limitation set forth in section 415(e) of the Code is not exceeded. Further, this Paragraph shall not apply for Limitation Years beginning after December 31, 1999.

- (B) For purposes of this Section 4.5, the following terms and phrases shall have these respective meanings.
 - (1) "Limitation Year" shall mean the calendar year.
 - - (a) the Participant's wages, salaries, fees for professional services, and other amounts received for personal services actually rendered in the course of employment with an Employer maintaining the Plan (including, but not limited to, commissions paid to sales personnel, compensation for services on the basis of a percentage of profits, commissions on insurance premiums, tips, and bonuses);
 - (b) in the case of a Participant who is an employee within the meaning of section 401(c)(1) of the Code and the regulations thereunder, the Participant's earned income (as described in section 401(c)(2) of the Code and the regulations thereunder);
 - (c) for purposes of Subsection 4.5(B)(2)(a) and (b), earned income from sources outside the United States (as defined in section 911(b) of the Code), whether or not excludable from gross income under section 911 of the Code or deductible under section 913 of the Code;

- (d) amounts described in sections 104(a)(3), 105(a), and 105(h) of the Code, but only to the extent that these amounts are includable in the gross income of the Employee;
- (e) amounts described in section 105(d) of the Code, whether or not these amounts are excludable from the gross income of the Employee under that section; and
- (f) amounts paid or reimbursed by the Employer for moving expenses incurred by an Employee, but only to the extent that these amounts are not deductible by the Employee under section 217 of the Code.
- (3) Participant's Remuneration for a Limitation Year shall not include:
 - (a) contributions made by the Employer to a plan of deferred compensation to the extent that, before application of the limitations imposed under section 415 of the Code with respect to such plan, the contributions are not includable in the gross income of the Participant for the year in which contributed (except as to elective deferrals to a qualified plan described in section 401(k) of the Code or to a tax-sheltered annuity plan described in section 403(b) of the Code, and except as to elective contributions to a deferred compensation plan described in section 457 of the Code or to a cafeteria plan described in section 125 of the Code), Employer contributions made on behalf of a Participant to a simplified employee pension plan described in section 408(k) of the Code to the extent deductible by the Employee under section 219(b)(7) of the Code, and distributions from a plan of deferred compensation other than an unfunded non-qualified plan:
 - (b) amounts realized from the exercise of a non-qualified stock option, or when restricted stock (or property) either becomes freely transferable or is no longer subject to a substantial risk of forfeiture;
 - (c) amounts realized from the sale, exchange, or other disposition of stock acquired under a qualified or incentive stock option; or
 - (d) other amounts which receive special tax benefits, such as premiums for group term life insurance to the extent not includable in the gross income of the Participant.
- (4) "Annual Additions" of a Participant, for any Limitation Year, shall mean the total (a) of the Employer contributions and forfeitures allocated to such Employee's accounts under the Plan for such year and (b) the amount of such Participant's Employee contributions to the Plan (excluding any Rollover Contributions).

- (5) "Maximum Annual Additions" of a Participant for any Limitation Year shall mean the lesser of \$30,000 or 25%, of such Participant's Remuneration during such year; provided, however, that the \$30,000 limitation shall be adjusted automatically to reflect any cost-of-living adjustments authorized by the Code or Treasury Regulations.
- 4.6 Rollover Contributions. (A) Qualified Rollover Contributions may be made to the Plan by any Employee of amounts received by such Employee from certain individual retirement accounts or annuities or from an employees' trust described in section 401(a) of the Code, which is exempt from tax under section 501(a) of the Code, but only if any such Rollover Contribution is made pursuant to and in accordance with applicable provisions of the Code and Treasury regulations promulgated thereunder. A Rollover Contribution of amounts that are "eliquible rollover distributions" within the meaning of section 402(f)(2)(A) of the Code may be made to the Plan irrespective of whether such eligible rollover distribution was paid to the Employee or paid to the Plan as a "direct" Rollover Contribution. A direct Rollover Contribution to the Plan may be effectuated only by wire transfer directed to the Trustee or by issuance of a check made payable to the Trustee, which is negotiable only by the Trustee and which identifies the Employee for whose benefit the Rollover Contribution is being made. Any Employee desiring to effect a Rollover Contribution to the Plan must complete a rollover request in accordance with the procedures established from time to time by the Administrative Board. The Administrative Board may require as a condition to accepting any Rollover Contribution that such Employee furnish any evidence that the Administrative Board in its discretion deems satisfactory to establish that the proposed Rollover Contribution is in fact eligible for rollover to the Plan and is made pursuant to and in accordance with applicable provisions of the Code and Treasury regulations. All Rollover Contributions to the Plan must be made in cash. A Rollover Contribution shall be credited to the Rollover Account of the Employee for whose benefit such Rollover Contribution is being made as soon as administratively feasible. (B) An Employee who has made a Rollover Contribution in accordance with this Section, but who has not otherwise become a Participant in the Plan in accordance with Article III, shall become a Participant coincident with such Rollover Contribution; provided, however, that such Participant shall not have a right to defer Base Pay or have Employer Contributions made on his behalf until he has otherwise satisfied the requirements imposed by Article III.
- 4.7 Voting, Tendering, and Exchanging of Company Stock. Notwithstanding any provisions herein to the contrary, the following provisions shall govern with respect to the voting, tendering, exchanging and other rights concerning Company Stock:
 - (A) Notice. At the time of mailing to shareholders of the notice of any shareholders' meeting of the Company, or any notice of a tender or exchange offer for Company Stock or notice by the Company of any other action with respect to Company Stock, the Company shall use its reasonable best efforts to cause to be delivered to each Participant and former Participant (whose account has allocated to it any shares of Company Stock) such notices and informational statements as are furnished to the Company's shareholders in respect of the exercise of voting, tendering, exchanging, or other rights, together with forms by which the Participant or former Participant may

instruct the Trustee, or revoke such instructions, with respect to the exercise of voting, tendering, exchanging, or other rights applicable to shares of Company Stock credited to such account.

- (B) Voting. Every Participant or former Participant (whose account has allocated to it any shares of Company Stock) shall have the right to direct the Trustee with respect to voting Company Stock allocated (or allocable) to such accounts, and the Trustee shall vote such allocated shares as directed. All of the shares of Company Stock for which no voting instructions are received shall be voted by the Trustee in a uniform manner as a single block in accordance with the instructions received with respect to a majority of such shares for which instructions are received.
- (C) Tendering and Exchanging. In the event of a tender or exchange offer, every Participant or former Participant (whose account has allocated to it any shares of Company Stock) shall have the right to direct the Trustee whether to accept or decline the offer with respect to Company Stock allocated (or allocable) to such accounts, and the Trustee shall take such actions as directed. All of the shares of Company Stock for which no instructions are received with respect to tender offer or exchange rights shall not be tendered or exchanged by the Trustee.

4.8 Restrictions on Employee Contributions.

- (A) In restriction of the Participants' Employee contribution elections provided in Section 4.2, the Employee contributions on behalf of any Participant for any calendar year shall not exceed \$7,000 (with such amount to be adjusted automatically to reflect any cost- of-living adjustments authorized by section 402(g)(5) of the Code), reduced by any "excess deferrals" from other plans allocated to the Plan by March 1 of the next following calendar year within the meaning of, and pursuant to the provisions of, section 402(g)(2) of the Code.
- (B) In further restriction of the Participants' Employee contribution elections provided in Section 4.2, it is specifically provided that one of the "actual deferral percentage" tests set forth in section 401(k)(3) of the Code and the Treasury Regulations thereunder must be met in each Plan Year. Such testing shall utilize the current year testing method as such term is defined in Internal Revenue Service Notice 98-1. For purposes of administering the tests described in this Section and to the extent permitted under section 414(s) of the Code and the Treasury Regulations thereunder, the Administrative Board shall include in a Participant's compensation all Employee contributions made by the Employer on behalf of such Participant that are not includible in the gross income of such Participant by reason of sections 125, 402(a)(8) or 402(h) of the Code.
- (C) If the restrictions set forth in Subsection (B) above would not otherwise be met for any Plan Year, the Employee contribution elections made pursuant to Section 4.2 of all Participants who are Highly-Compensated Participants shall be revised by the Administrative Board on a temporary basis to the extent necessary to meet such restrictions. Any reduction of

amounts to be deferred by Participants who are Highly-Compensated Participants shall be applied by first reducing on an equal basis Employee contribution elections of 16%, then reducing on an equal basis Employee contribution elections of 15% and continuing in such manner until the restrictions set forth in Subsection (B) above are met. A Participant whose Employee contribution election percentage has been reduced pursuant to this Subsection shall be notified of such reduction in writing by the Administrative Board. The intent of the foregoing provision is to effect a prospective reduction in a Participant's Employee contribution election percentage. If the Administrative Board temporarily reduces Participants' Employee contribution elections pursuant to this Subsection and subsequently determines at any time that, on a projected basis for such Plan Year, such Participants' Employee contribution elections, as originally made, may be wholly or partially restored, the Administrative Board shall increase the Employee contribution elections of such Participants in the same manner as such Employee contribution elections were reduced to the extent consistent with meeting the restrictions referred to in the first sentence of this Subsection as of the last day of such Plan Year.

4.9 Restrictions on Employer Contributions. In restriction of the Employer contributions provided in Section 4.1, it is specifically provided that one of the "actual contribution percentage" tests set forth in section 401(m) of the Code and the Treasury Regulations thereunder must be met in each Plan Year. Such testing shall utilize the current year testing method as such term is defined in Internal Revenue Service Notice 98-1. The Administrative Board may elect, in accordance with applicable Treasury Regulations, to treat Employee contributions to the Plan as Employer contributions for purposes of meeting this requirement.

In further restriction of the Employer contributions provided in Section 4.1, it is specifically provided that the "multiple use of alternative limitation" tests set forth in Treasury Regulation section 1.401(m)-2(b) must be met in each Plan Year. Such tests will be met by reduction of the actual contribution percentage with respect to all Highly Compensated Participants under the Plan, in the manner described in Treasury Regulation sections 1.401(m)-1(e)(2) and 1.401(m)-2(c)(3), to the extent necessary to meet such tests.

4.10 Excess Contributions.

(A) Anything to the contrary herein notwithstanding, any Employee contributions to the Plan for a calendar year on behalf of a Participant in excess of the restrictions set forth in Section 4.8(A) shall be distributed to such Participant not later than April 15 of the next following calendar year. At the same time, any related Employer contributions shall be forfeited, except to the extent such related Employer contributions must be distributed pursuant to Subsection (C) below.

(B) Anything to the contrary herein notwithstanding, if, for any Plan Year, the aggregate Employee contributions made by the Employer on behalf of Highly Compensated Participants exceeds the maximum amount of Employee contributions permitted on behalf of such Highly Compensated Participants pursuant to Section 4.8(B) (determined by reducing Employee contributions on behalf of Highly Compensated Participants in order of the highest

dollar amounts contributed on behalf of such Highly Compensated Participants in accordance with section 401(k)(8)(C) of the Code and the Treasury regulations thereunder), then such excess shall be distributed to the Highly Compensated Participants on whose behalf such excess was contributed before the end of the next following Plan Year at the same time, any related Employer contributions shall be forfeited, except to the extent such related Employer contributions must be distributed pursuant to Subsection (C) below.

- (C) Anything to the contrary herein notwithstanding, if, for any Plan Year, the aggregate Employer contributions allocated to the Accounts of Highly Compensated Participants exceeds the maximum amount of such Employer contributions permitted on behalf of such Highly Compensated Participants pursuant to Section 4.9 (determined by reducing Employer contributions made on behalf of Highly Compensated Participants in order of the highest dollar amounts contributed on behalf of such Highly Compensated Participants in accordance with section 401(m)(6)(C) of the Code and the Treasury regulations thereunder), then such excess shall be distributed to the Highly Compensated Participants on whose behalf such excess contributions were made (or, if such excess contributions are forfeitable, they shall be forfeited) before the end of the next following Plan Year. Employer contributions shall be forfeited pursuant to this Paragraph only if distribution of all vested Employer contributions is insufficient to meet the requirements of this Paragraph. If vested Employer contributions are distributed to a Participant and nonvested Employer contributions remain credited to such Participant's Accounts, such nonvested Employer contributions shall vest at the same rate as if such distribution had not been made. Any excess contribution which is forfeited shall be considered forfeited on March 15 of the next following Plan Year.
- (D) In coordinating excess contributions pursuant to this Section, such excess contributions shall be treated in the following order:
 - (1) first, excess Employee contributions described in Subsection (A) above shall be distributed, and related Employer contributions shall be forfeited, except to the extent such related Employer contributions must be distributed pursuant to Subsection (C) above;
 - (2) second, excess Employee contributions described in Subsection (B) above shall be distributed, and related Employer contributions shall be forfeited, except to the extent such related Employer contributions must be distributed pursuant to Subsection (C) above; and
 - (3) third, excess Employer contributions described in Subsection (C) above shall be distributed or, if forfeitable, forfeited.
- (E) Any distribution or forfeiture of excess contributions pursuant to Subsections (A), (B), or (C) of this Section shall be adjusted for income or loss allocated thereto in a manner consistent with applicable Treasury Regulations, rulings and notices, and such distribution (or forfeiture, if applicable) will include such income or be reduced by such loss.

Any distribution or forfeiture of excess contributions pursuant to this Section shall be made in cash.

ARTICLE V

VALUATION OF ACCOUNTS

All amounts contributed to the Trust shall be invested as soon as administratively feasible following their receipt by the Trustee, and the balance of each Account shall reflect the result of daily pricing of the assets in which such Account is invested from the time of receipt by the Trustee until the time of distribution.

ARTICLE VI

VESTING

- 6.1 Normal Retirement. Notwithstanding the provisions of Section 6.4, a Participant shall have a Nonforfeitable Interest in such Participant's entire Employer Contribution Account if the Participant is actively employed (A) on or after age 65, (B) on the day immediately preceding such Participant's death, or (C) on separation from service of the Employer upon incurring a Total Disability. No forfeiture shall thereafter arise under Section 4.3.
- 6.2 Plan Termination, Partial Plan Termination, or Complete Plan Discontinuance of Employer Contributions. Notwithstanding any other provision of this Plan, in the event of a termination or partial termination of the Plan, or a complete discontinuance of Employer contributions under the Plan, all affected Participants shall have a Nonforfeitable Interest in their Employer Contribution Accounts determined as of the date of such event. The value of these accounts and their Employee Contribution Accounts as of such date shall be determined in accordance with the provisions of the Plan.
- 6.3 Vesting of Employee Contributions. A Participant shall accrue a Nonforfeitable Interest in such Participant's Employee Contribution Account and Prior Employee Contribution Account at all times.
- 6.4 Vesting of Employer Contributions. A Participant shall accrue a Nonforfeitable Interest in such Participant's Employer Contribution Account at the rate specified in the following table:

PERIOD OF SERVICE
(AS DEFINED IN SECTION 2.27)

NONFORFEITABLE INTEREST

Less than 5 years 5 years or more

0% 100%

Notwithstanding the above, a Participant will have a 100% Nonforfeitable Interest in such Participant's Employer Contribution Account upon retirement under the American General Retirement Plan or any successor plan.

Each Participant with a Period of Service of five years or more with the Employer on the effective date of any amendment to the preceding paragraph may elect, within a reasonable period after the adoption of the amendment, to have such Participant's Nonforfeitable Interest computed under the Plan without regard to such amendment. The period during which the election may be made shall commence with the date the amendment is adopted and shall end on the later of:

(A) 60 days after the restatement is adopted;

- (B) 60 days after the amendment becomes effective; or
- (C) 60 days after the Participant is issued written notice of the amendment by the Employer or the Administrative Board.
- 6.5 Vesting After a Period of Severance. No Period of Service after a Period of Severance equal to at least five consecutive One-Year Breaks in Service shall be taken into account in determining the nonforfeitable percentage in a Participant's Employer Contribution Account accrued up to any such Period of Severance.
- 6.6 Vesting Before a Period of Severance of Five Consecutive One-Year Breaks in Service. A Participant who separates from service of the Employer and received a distribution of the Nonforfeitable Interest (of less than 100%), in such Participant's Employer Contribution Account in accordance with Section 7.4, shall forfeit amounts that are forfeitable in accordance with Section 4.3.
- 6.7 Election to Make a Withdrawal. If a Participant elects to make a withdrawal pursuant to Section 7.6(B), (C), (D) or (E) while remaining employed by the Company, such distribution will be subject to the vesting provisions of Section 6.4. Upon incurring a Period of Severance equal to at least five consecutive One-Year Breaks in Service, that part of such individual's Employer Contribution Account which is not vested shall be forfeited, and such forfeiture shall be available to reduce future Employer Contributions.
- 6.8 Transfer Between Plans. If a Participant ceases to be eligible as an Employee under this Plan but immediately becomes an Employee for purposes of another thrift plan maintained by the Company or its subsidiaries ("Other Plan"), such change in employment status will not be deemed a termination of employment for purposes of this Plan.

A Participant's Employee Contribution Account, Prior Employee Contribution Account and Employer Contribution Account will be frozen, and no future Employee or Employer contributions will be made. However, the Participant will continue to accrue vesting in this Plan based on years of service credited in the Other Plan at the rate which would have been applicable had such individual continued participation in this Plan.

If an Employee has ceased to be a Participant under an Other Plan and immediately becomes a Participant under this Plan, such Employee will not be required to complete the one-year Period of Service or one year of "participation service" in Section 3.1 before participating in this Plan; and the Employee's Period of Service and Hours of Service under the Other Plan will be added to the Period of Service and Hours of Service under this Plan.

6.9 Independent Life Transferred Employee. A Transferred Employee is an Employee who was an employee of Independent Life and Accident Insurance Company on the day preceding his employment by the Employer. For purposes of computing & Participant's Nonforfeitable Interest in this Plan, a Transferred Employee's Period of Service will include full and partial years of

credited service under the American General Invest Plan (formerly the Independent Life INVEST Plan) plus, if applicable, service with the Company from date of transfer from Independent Life and Accident Insurance Company to April

ARTICLE VII

BENEFITS

- 7.1 Normal Retirement. Upon separation from service on or after satisfying eligibility requirements for early, normal, or late retirement under the Company's retirement plan, other than by reason of death, a Participant shall be entitled to a benefit based on the combined balance of the Participant's Employee Contribution Account, Prior Employee Contribution Account and Employer Contribution Account distributed in a manner provided in Article
- 7.2 Disability. In the event that a Participant separates from service of the Employer upon incurring a Total Disability before the Normal Retirement Date, such Participant shall be entitled to a benefit based on the combined balance of such individual's Employee Contribution Account, Prior Employee Contribution Account, and Employer Contribution Account distributed in the manner provided in Article IX.
- 7.3 Death. In the event of the death of a Participant prior to the commencement of a benefit described in Sections 7.1, 7.2, or 7.4, the Beneficiary shall be paid the combined balance in the Participant's Employee Contribution Account, Prior Employee Contribution Account and Employer Contribution Account distributed in the manner provided in Article IX.
- 7.4 Termination of Service. In the event a Participant separates from service of the Employer prior to the Normal Retirement Date for any reason other than death or Total Disability and Section 7.1 does not apply, the Nonforfeitable Interest in such Participant's Employer Contribution Account determined pursuant to Article VI and the balance of such Participant's Employee Contribution Account and Prior Employee Contribution Account shall be distributable to the Participant in accordance with Article IX.

7.5 Valuation Date to be Used.

- (A) If a Participant or Beneficiary becomes entitled to a benefit pursuant to Section 7.1, 7.2, 7.3, or 7.4, and elects to receive a distribution of such benefit as a result thereof, the value of the account balances to be distributed shall be determined as of the date of the event giving rise to the distribution. For purposes of Section 7.1, the event occasioning the benefit shall be the Participant's separation from service on or after satisfying the eligibility requirements for early, normal, or late retirement under the Company's retirement plan, other than by reason of death. For purposes of Section 7.2, the event occasioning the benefit shall be the Participant's separation from service of the Employer on account of Total Disability. For purposes of Section 7.3, the event occasioning the benefit shall be the death of the Participant prior to the commencement of a benefit described in Section 7.1, 7.2, or 7.4. For purposes of Section 7.4, the event occasioning the benefit shall be the date the Participant separated from service.
- (B) If pursuant to Section 8.4, a Participant elects to receive a benefit after an event set forth in Section 7.1, 7.2, 7.3, or 7.4, the value of the account balances to be distributed

33

shall be determined as of the Withdrawal Date on which the Plan Administrator receives the Participant's withdrawal request in accordance with the procedures established from time to time by the Administrative Board.

7.6 Withdrawals.

- (A) A Participant may withdraw from his Prior Employee Contribution Account any or all amounts held in such Account which have been so held for six months or more.
- (B) A Participant who has withdrawn all amounts in his Prior Employee Contribution Account but who has not made or had made on his behalf to the Plan basic Employee contributions for at least 60 cumulative months may withdraw from his Employer Contribution Account any or all amounts held in such account which have been so held for 24 months or more, but not in excess of his Nonforfeitable Interest in the then value of such account. Any amounts in an Employer Contribution Account which is a Rollover Account must be withdrawn prior to amounts in any other Employer Contribution Account.
- (C) A Participant who has withdrawn all amounts in his Prior Employee Contribution Account and who has made or had made on his behalf to the Plan basic Employee contributions for at least 60 cumulative months may withdraw from his Employer Contribution Account an amount not exceeding his Nonforfeitable Interest in the then value of such Account. Any amounts in an Employer Contribution Account which is a Rollover Account must be withdrawn prior to amounts in any other Employer Contribution Account.
- (D) A Participant who has attained age 59-1/2 and who has made all available withdrawals pursuant to Subsections (A), (B), and (C) above may withdraw from his Employer Contribution Account Subaccount an amount not exceeding the then value of such Account.
- (E) A Participant who has attained age 59-1/2 and who has made all available withdrawals pursuant to Subsections (A), (B), (C) and (D) above may withdraw from his Employee Contribution Account an amount not exceeding the then value of such Account.
- (F) A Participant who has a financial hardship, as determined by the withdrawal committee ("Withdrawal Committee") appointed by the Administrative Board, and who has made all available withdrawals pursuant to the Subsections above and pursuant to the provisions of any other plans of the Employer or any Controlled Entities of which he is a member and who has obtained all available loans pursuant to the provisions of any plans of the Employer and any Controlled Entities of which he is a member may withdraw from his Employee Contribution Account amounts not to exceed the lesser of (1) his Nonforfeitable Interest in the then value of such account or (2) the amount determined by the Withdrawal Committee as being available for withdrawal pursuant to this Subsection. No withdrawals on account of financial hardship will be permitted for amounts less than \$500. For purposes of this Subsection, financial hardship means the immediate and heavy financial needs of the Participant. A withdrawal based

upon financial hardship pursuant to this Subsection shall not exceed the amount required to meet the immediate financial need created by the hardship and not reasonably available from other resources of the Participant. The determination of the existence of a Participant's financial hardship and the amount required to be distributed to meet the need created by the hardship shall be made by the Withdrawal Committee based upon information requested from the Participant by the Withdrawal Committee and considering all relevant facts and circumstances. The amount of an immediate and heavy financial need may include any amounts necessary to pay any federal, state, or local income taxes or penalties reasonably anticipated to result from the distribution. In addition to any immediate and heavy financial need as may be determined by the Withdrawal Committee, a withdrawal shall be deemed to be made on account of an immediate and heavy financial need of a Participant if the withdrawal is on account of:

- (1) medical expenses described in section 213(d) of the Code incurred by the Participant, the Participant's spouse or any dependents of the Participant (as defined in section 152 of the Code) or necessary for those persons to obtain medical care described in section 213(d) of the Code, and not reimbursed by insurance;
- (2) costs directly related to the purchase (excluding mortgage payments) of a principal residence of the Participant;
- (3) payment of tuition and related educational fees for the next 12 months of post-secondary education of the Participant, or the Participant's spouse, children or dependents (as defined in section 152 of the Code); or
- (4) the need to prevent the eviction of the Participant from his principal residence or foreclosure on the mortgage of the Participant's principal residence.

The decision of the Withdrawal Committee shall be final and binding, provided that all Participants similarly situated shall be treated in a uniform and nondiscriminatory manner. The above notwithstanding, withdrawals under this Subsection from a Participant's Employee Contribution Account shall be limited to the sum of the Participant's Employee contributions to the Plan, less any previous withdrawals of such amounts. A Participant who makes a withdrawal under this Subsection may not again make elective contributions to the Plan or any other qualified or nonqualified plan of the Employer or any Controlled Entity for a period of 12 months following such withdrawal. Further, such Participant may not make elective contributions under the Plan or any other plan maintained by the Employer or any Controlled Entity for such Participant's taxable year immediately following the taxable year of the withdrawal in excess of the amount of such Participant's elective contributions for the taxable year less the withdrawal.

(G) All withdrawals pursuant to this Section shall be made as of the Withdrawal Date on which the Plan Administrator receives a withdrawal request in accordance with the procedures established from time to time by the Administrative Board. All withdrawals shall be made pro rata from each fund in which such account is invested.

7.7 Loans.

- (A) Upon application by (1) any Participant who is an Employee or (2) any Participant no longer employed by the Employer, a beneficiary of a deceased Participant or an Alternate Payee under a Qualified Domestic Relations Order, as defined in Section 13.11, who retains a balance in his Accounts under the Plan and who is a party-in-interest, as that term is defined in section 3(14) of ERISA, as to the Plan (an individual who is eligible to apply for a loan under this Section being hereinafter referred to as a "Participant" for purposes of this Section), the loan committee ("Loan Committee") appointed by the Administrative Board may in its discretion direct the Trustee to make a loan or loans to such Participant. Such loans shall be made pursuant to the provisions of the Loan Committee's written loan procedure, which procedure is hereby incorporated by reference as a part of the Plan.
- (B) A loan to a Participant may not exceed 50% of the then value of such Participant's Nonforfeitable Interest in his Accounts.
- (C) Paragraph (B) above to the contrary notwithstanding, the amount of a loan made to a Participant under this Section shall not exceed an amount equal to the difference between:
 - (1) The lesser of \$50,000 (reduced by the excess, if any, of (A) the highest outstanding balance of loans from the Plan during the one-year period ending on the day before the date on which the loan is made over (B) the outstanding balance of loans from the Plan on the date on which the loan is made) or one-half of the present value of the Participant's total nonforfeitable accrued benefit under all qualified plans of the Employer or Controlled Entity; minus
 - (2) The total outstanding loan balance of the Participant under all other loans from all qualified plans of the Employer or a Controlled Entity.

ARTICLE VIII

COMMENCEMENT OF BENEFITS

8.1 Benefits After Normal Retirement Date. Payment shall be made within 60 days after the Participant separates from service of the Employer (including separation by reason of death or Total Disability) on or after the Participant's Normal Retirement Date.

Although full vesting is granted at age 65, distribution of the account of a Participant who continues active work past the Normal Retirement Date shall be postponed until such Participant's actual termination of employment or prior Plan withdrawal. Any such Participant may continue active participation in the Plan until the Participant's actual retirement date.

Notwithstanding the foregoing, subject to the provisions of Section 8.4, a Participant who terminates employment (for a reason other than death) may, if his total accounts (both vested and nonvested) exceed \$5,000, elect to defer payment of his benefit by making an appropriate election to defer with the Plan Administrator.

- 8.2 Certain Benefits Before Normal Retirement Date. Except as provided in Section 8.1, upon the death of the Participant, payment shall be made not later than 60 days after receipt by the Plan Administrator of proof of death. Except as provided in Section 8.1, upon separation from service of the Employer upon incurring a Total Disability, payment shall be made no later than 60 days after the determination that Total Disability exists.
- 8.3 Termination of Service Before Normal Retirement Date. Upon separation from service before Normal Retirement Date other than by reason of death or Total Disability, the benefit to which a Participant is entitled under Section 7.4 shall be paid no later than 60 days after the Participant's separation from service, and is to be paid in the form prescribed in Article IX.
- 8.4 Commencement of Benefits. Notwithstanding anything in Section 8.1, 8.2, or 8.3, other than an election to defer made by a Participant pursuant to Section 8.1, payments of benefits shall be made or shall commence no later than 60 days after the close of the Plan Year in which the Participant attains or would have attained age 65, or, if later, separates from service.

Notwithstanding anything in this Section or in Sections 8.1, 8.2, or 8.3 to the contrary, the entire Nonforfeitable Interest of each Participant shall be distributed to such Participant no later than April 1 of the calendar year following the later of (i) the calendar year in which the Participant attains age 70 1/2 or (ii) the calendar year in which the Participant terminates his employment with the Employer (provided, however, that clause (ii) of this sentence shall not apply prior to January 1, 1999, and shall not apply in the case of a Participant who is a 'five- percent owner' (as defined in section 416 of the Code) with respect to the Plan Year ending in the calendar year in which such Participant attains age 70 1/2. Notwithstanding the foregoing, in accordance with procedures adopted by the Administrative Board, (1) a Participant (who is not a

'five-percent owner' with respect to the calendar year in which the Participant attains age 70 1/2 who attains age 70 1/2 in calendar year 1996, 1997, or 1998, may elect to defer his payment as if such age were attained after calendar year 1998, and (2) a Participant who attains age 70 1/2 prior to calendar year 1997 and who has not terminated employment with the Employer may elect to stop distributions which have commenced as of a prior payment date.

If a Participant dies before the entire Nonforfeitable Interest has been distributed to such Participant, such interest shall be distributed, within five years of the Participant's death, to the designated Beneficiary.

No benefit shall be paid to a Participant who is under the age of 65, without such person's consent, if the total value (both vested and unvested) of such Participant's accounts exceeds \$5,000.

A Participant's commencement of benefits shall be in compliance with the provisions of section 401(a)(9) of the Code and applicable Treasury Regulations thereunder.

ARTICLE IX

DISTRIBUTION OF BENEFITS

- 9.1 Mode of Benefit Payments. Upon death, retirement, Total Disability, termination of employment for any other reason or withdrawal, amounts in the Participant's Employer Contribution Account (subject to the vesting provisions of Article VI), and amounts in the Participant's Employee Contribution Account, Prior Employee Contribution Account, and Rollover Account shall be distributed as follows:
 - (A) from the Stock Fund, either in Company Stock or in cash at the Participant's election;
 - (B) from the Investment Funds, in cash.

Notwithstanding the above and notwithstanding the following sentence, if a distribution would include fractional shares of Company Stock, the value of such fractional shares shall be paid in cash instead. Except in situations subject to Section 9.3, if an election is not received by the Plan Administrator prior to commencement of payment of benefits pursuant to Section 8.4, the distribution of the vested interest in the Participant's Employer Contribution Account and of any amounts in the Participant's Employee Contribution Account, Prior Employee Contribution Account, and Rollover Account which are held in the Stock Fund shall be made in Company Stock, and the distribution of all other amounts shall be in cash.

All assets held in the Participant's Accounts shall be valued as of the date prescribed in Section 7.5. No interest or other earnings shall be paid after the valuation date prescribed in Section 7.5. Distributions of benefits shall be made as soon as administratively feasible and in accordance with Article VIII.

- 9.2 Stock Certificates. Where certificates for shares of stock are distributed, they will be issued in the name of the Participant only. However, upon the death of a Participant, certificates will be issued in the name of the Beneficiary.
- 9.3 Automatic Distributions. A Participant whose total accounts (both vested and nonvested) are valued at \$5,000 or less must make an election authorized by Section 9.1 within 30 days following termination of employment, retirement, or Total Disability. A Beneficiary must make an election within 30 days following the Participant's death. If an election is not received by the Plan Administrator within the applicable time periods stated above, the distribution shall be made in cash.

Nothing in this Article shall be construed to permit or require distribution of assets in which the Participant does not have a Nonforfeitable Interest, as determined under Article VI. Likewise, this Article shall not be construed to permit or require a distribution without a Participant's consent if such distribution is prohibited by Section 8.4.

9.4 Unclaimed Benefits. In the case of a benefit payable on behalf of a Participant, if the Plan Administrator is unable to locate the Participant or Beneficiary to whom such benefit is payable, upon the Plan Administrator's determination thereof, such benefit shall be forfeited and shall be available to reduce future Employer contributions. Notwithstanding the foregoing, if subsequent to any such forfeiture the Participant or Beneficiary to whom such benefit is payable makes a valid claim for such benefit, such forfeited benefit shall be restored to the Plan in the manner provided in Section 4.3. No interest or other earnings shall be paid or accrued in regard to the restored benefit.

9.5 Direct Rollovers.

- (A) This Section 9.5 applies to distributions made on or after January 1, 1993. Notwithstanding any provision of the Plan to the contrary that would otherwise limit a distributee's election under this Section, a distributee may elect, at the time and in the manner prescribed by the Plan Administrator, to have any portion of an "eligible rollover distribution" paid directly to an "eligible retirement plan" specified by the "distributee" in a "direct rollover."
- - (1) "Eligible rollover distribution" is any distribution of all or any portion of the balance to the credit of the distributee, except that an eligible rollover distribution does not include: any distribution that is one of a series of substantially equal periodic payments (not less frequently than annually) made for the life (or life expectancy) of the distributee or the joint lives (or joint life expectancies) of the distributee and the distributee's designated beneficiary, or for a specified period of ten years or more; any distribution to the extent such distribution is required under section 401(a)(9) of the Code; and the portion of any distribution that is not includible in gross income (determined without regard to the exclusion for net unrealized appreciation with respect to employer securities). Further, a distribution pursuant to Section 7.6(E) shall not constitute an Eligible Rollover Distribution to the extent provided in section 402(c)(4) of the Code and the interpretive authority thereunder.
 - (2) "Eligible retirement plan" is an individual retirement account described in section 408(a) of the Code, an individual retirement annuity described in section 408(b) of the Code, an annuity plan described in section 403(a) of the Code, or a qualified trust described in section 401(a) of the Code, that accepts the distributee's eligible rollover distribution. However, in the case of an eligible rollover distribution to the surviving spouse, an eligible retirement plan is an individual retirement account or individual retirement annuity.
 - (3) "Distributee" includes an employee or former employee. In addition, the employee's or former employee's surviving spouse and the employee's or former employee's spouse or former spouse who is the alternate payee under a qualified

domestic relations order, as defined in section 414(p) of the Code, are distributees with regard to the interest of the spouse or former spouse.

(4) "Direct rollover" is a payment by the plan to the eligible retirement plan specified by the distributee.

ARTICLE X

PLAN AMENDMENT AND TERMINATION

10.1 Amendment of the Plan. The Company shall have the right to amend or modify this Plan and (with the consent of the Trustee) the Trust agreement at any time, and from time to time, to any extent that it may deem advisable. Any such amendment or modification shall be set out in an instrument in writing, duly authorized by the Board of Directors and executed by the Company. No such amendment or modification shall, however, increase the duties or responsibilities of the Trustee without its consent thereto in writing, or have the effect of transferring to or vesting in any Employer any interest or ownership in any properties of the Trust Fund, or of permitting the same to be used for or diverted to purposes other than for the exclusive benefit of the Participants and their Beneficiaries. No such amendment shall decrease the accounts of any Participant or shall decrease any Participant's Nonforfeitable Interest in such account. Notwithstanding anything herein to the contrary, the Plan or the Trust agreement may be amended in such manner as may be required at any time to make it conform to the requirements of the Code or any United States statutes with respect to employee trusts, or any amendment thereto, or any regulations or rulings issued pursuant thereto, and no such Plan or Trust amendment shall be considered prejudicial to any then existing rights of any Participant or Beneficiary under the Plan.

For purposes of this Section and subject to regulations to be issued by the Secretary, a Plan amendment made after July 30, 1984, which has the effect of reducing an early retirement benefit or an optional form of benefit, with respect to benefits attributable to service before the amendment (even though unvested), shall be treated as reducing a Participant's account.

- 10.2 Communication of Amendments. All amendments, including one to terminate the Plan, shall be adopted in writing by the Company's Board of Directors. Any material modification of the Plan by amendment or termination shall be communicated to all interested parties and the Secretaries of Labor and the Treasury in the time and manner prescribed by law.
- 10.3 Termination or Discontinuance of Employer Contributions. Upon plan termination or discontinuance of Employer contributions under the Plan all account balances shall be valued in accordance with Section 6.2. The Trustee shall then, as soon as administratively feasible, pay each Participant and Beneficiary the entire interest in the Trust attributable to such Participant or Beneficiary in a lump sum, and shall pay any remaining amount to the Employer. In case of any Participant whose residence is unknown, the Plan Administrator shall notify such Participant at the last known address by certified mail with return receipt requested advising such Participant of a pending distribution.
- 10.4 Acceptance or Rejection of Amendment Employers. The Company shall promptly deliver to each other Employer any amendment to this Plan or the Trust agreement. Upon delivery to an Employer of an executed copy of an amendment properly authorized and

42

adopted by the Company, the Plan as to such Employer shall be thereupon amended in accordance therewith.

10.5 Termination of the Plan by an Employer. An Employer may at any time, by adoption of a resolution, terminate the Plan with respect to the Employees of said Employer, and may direct and require the Trustee to liquidate the share of the Trust Fund allocable to its Employees or their Beneficiaries. If the Plan is terminated by fewer than all Employers, the Plan shall continue in effect for the Employees of the remaining Employers. In the event that an Employer shall cease to exist, the Plan shall be terminated with respect to the Employees of such Employer, unless a successor organization adopts and continues the Plan

10.6 Notice of Terminating Employer. A terminating or withdrawing Employer shall give 90 days notice in writing of its intention to the Administrative Board, the Company, and the Trustee, unless a shorter notice is agreed to by the Company. If an Employer withdraws from this Plan and provides for a successor plan for its Employees, the Trust Fund assets of this Trust held on behalf of such Employer may be determined and transferred to a successor trust upon approval by the appropriate District Director of Internal Revenue.

ARTICLE XI

ADMINISTRATION

- 11.1 Named Fiduciaries. The named fiduciaries shall be the Plan Administrator and the Trustee.
 - 11.2 Appointment of the Trustee.
- (A) The Company shall designate the Trustee in a written statement filed with the Company's Board of Directors. The appointment of the Trustee shall become effective at such time as the Trustee and the Company execute a valid written trust which definitely and affirmatively precludes prohibited diversion.
- (B) The resignation of a Trustee shall be made in writing, submitted to the Company, and recorded in the minutes of the Board of Directors. The discharge of any person described in the preceding sentence shall be effectuated in writing by the Company and delivered to such person with the details thereof recorded in the minutes of the Company's Board of Directors. Appointment of a successor Trustee shall be carried out in the manner prescribed in Subsection (A).
- 11.3 Trustee's Powers and Duties. The powers and duties of the Trustee shall be to manage and control the funds of the Trust in accordance with the terms of the Trust agreement forming a part hereof.
- 11.4 Administrative Expenses. Except for fees relating to Participant loans and commissions on and other costs relating to acquisitions or disposition of securities, including any reasonable fees in connection with the Investment Funds, the Employer may pay the administrative expenses of the Plan and Trust, including the reasonable compensation of the Trustee and reimbursement of its reasonable expenses. To the extent not paid by the Employer, the administrative expenses of the Plan and Trust will be paid by the Trustee of the Trust.
- 11.5 Plan Administrator's Powers and Duties. The Plan Administrator shall have the following powers and duties:
 - (A) to construe and interpret the provisions of the Plan;
 - (B) to decide all questions of eligibility for Plan participation and for the payment of benefits;
 - (C) to provide appropriate parties, including government agencies, with such returns, reports, schedules, descriptions, and individual statements as required by law within the times prescribed by law, to furnish to the Employer, upon request, copies of any or all such materials, and further, to make copies of such instruments, reports, and

descriptions as are required by law available for examination by Participants (and such of their Beneficiaries who are or may be entitled to benefits under the Plan) in such places and in such manner as required by law;

- (D) to obtain from the Employer, the Employees, and the Trustee such information as shall be necessary for the proper administration of the Plan;
- (E) to determine the amount, manner, and time of payment of benefits hereunder;
- (F) subject to the approval of the Company only as to any additional expense, to appoint and retain such agents, counsel, and accountants for the purpose of properly administering the Plan and, when required to do so by law, to engage an independent Certified Public Accountant to annually prepare the audited financial statement of the Plan's operations;
- (G) to take all actions and to communicate to the Trustee in writing all necessary information to carry out the terms of the Plan and the Trust agreement;
- (H) to notify the Trustee in writing of the termination of the Plan or the complete discontinuance of Employer contributions, $\,$
- (I) to direct the Trustee to distribute assets of the Trust to each Participant and Beneficiary in accordance with Article IX of the Plan:
- (J) to furnish each recipient of a "qualifying rollover distribution," as defined by section 402 of the Code, with a written explanation of the Code provisions under which such distribution will not be subject to tax if transferred to an eligible retirement plan within 60 days after which the recipient received the distribution and, if applicable, the provisions concerning "Capital Gains Treatment for Portion of Lump Sum Distribution" and the provisions concerning "Tax on Lump Sum Distributions" provided by section 402 of the Code;
- (K) to establish, at a meeting duly called for such purpose, a funding policy and method consistent with the objectives of the Plan and the requirements of Title I of ERISA. The Administrative Board shall meet at least annually to review such funding policy and method; and
- (L) to do such other acts reasonably required to administer the Plan in accordance with its provisions or as may be provided for or required by law.

All actions of the Plan Administrator taken pursuant to this Section shall be presented in a meeting of the Administrative Board and recorded in the minutes thereof. An annual report shall be presented to the Board of Directors.

- 11.6 Named Fiduciary's Powers and Duties. It shall be the responsibility of the Plan Administrator to provide a notice in writing to any Participant or Beneficiary whose claim for benefits under this Plan has been denied by the Plan Administrator, setting forth the specific reasons for such denial, and to afford such Participant or Beneficiary a reasonable opportunity for a full and fair review of that decision.
- 11.7 Allocation of Functions. Where more than one person serves as Plan Administrator, such persons may agree in writing to allocate among themselves the various powers and duties prescribed in Section 11.5, provided all such persons sign such agreement. A copy of any such agreement shall be promptly relayed to the Company.

ARTICLE XII

ADOPTION AND WITHDRAWAL

- 12.1 Procedure for Adoption. Any Controlled Entity may, with the approval of the Board of Directors, adopt and become an Employer under this Plan pursuant to appropriate written resolutions of the board of directors of such adopting Controlled Entity, by executing and delivering to the Company and the Trustee an adoptive instrument specifying the classification of its Employees who are to be eligible to participate in the Plan, and by agreeing to be bound as an Employer by all the terms of the Plan with respect to its eligible Employees. The adoptive instrument may contain such changes and variations in the terms of the Plan as may be acceptable to the Company. However, the sole, exclusive right of any other amendment of whatever kind or extent to the Plan or Trust are reserved by the Company. The adoption agreement shall become, as to such adopting organization and its Employees, a part of this Plan, as then or thereafter amended, and the related Trust. It shall not be necessary for the adopting organization to sign or execute the original or the amended Plan and Trust documents. The effective date of the Plan for any such adopting organization shall be that stated in the adoption agreement, and from and after such effective date such adopting organization shall assume all the rights, obligations, and liabilities of an Employer hereunder and under the Trust. The Company's administrative powers and control of the Plan, as provided in this instrument and the Trust, including the sole right of amendment, and of appointment and removal of the Administrative Board and the Trustee and their successors, shall not be diminished by reason of the participation of any such adopting organization in the Plan and Trust.
- 12.2 Effect of Adoption. The following special provisions shall apply to all Employees.
 - (A) An Employee shall be considered in service while regularly employed simultaneously or successively by one or more Employers.
 - (B) The transfer of an Employee from one Employer to another Employer shall not be deemed a termination of service.
- 12.3 Withdrawal. Any participating Employer by action of its board of directors or other governing authority and notice to the Company and the Trustee, may withdraw from the Plan and Trust at any time, without affecting other Employers not withdrawing, by complying with the provisions of the Plan and Trust. A withdrawing Employer may arrange for the continuation, by itself or its successor, of this Plan and Trust in separate forms for its own Employees, with such amendments, if any, as it may deem proper, and may arrange for continuation of the Plan and Trust by merger with an existing plan and trust and transfer of Trust assets. The Company may, in its absolute discretion, terminate an adopting Employer's participation at any time when in its judgment such adopting Employer fails or refuses to discharge its obligations under the Plan. If an Employer is no longer a Controlled Entity, the Company may at any time terminate such Employer's participation under this Plan.

ARTICLE XIII

MISCELLANEOUS

- 13.1 Merger of this Plan with Another Plan. In the event of any merger or consolidation of the Plan with, or transfer, in whole or in part, of the assets and liabilities of the Trust Fund to another trust fund held under any other plan of deferred compensation maintained or to be established for the benefit of all or some of the Participants of this Plan, the assets of the Trust Fund applicable to such Participants shall be transferred to the other trust fund only if:
 - (A) each Participant would (if either this Plan or the other plan then terminated) receive a benefit immediately after the merger, consolidation, or transfer which is equal to or greater than the benefit such Participant would have been entitled to receive immediately before the merger, consolidation, or transfer (if this Plan had then terminated);
 - (B) resolutions of the board of directors of the Employer under this Plan, or of any new or successor employer' of the affected Participants, shall authorize such transfer of assets; and, in the case of the new or successor employer of the affected Participants, its resolutions shall include an assumption of liabilities with respect to such Participants' inclusion in the new employer's plan; and
 - (C) such other plan and trust are qualified under sections 401(a) and 501(a) of the Code.
- 13.2 Assignment and Alienation of Benefits. Except as otherwise provided in Section 13.11, except as to certain judgments and settlements described in section 401(a)(13) of the Code, and except as otherwise provided under other applicable law, no right or interest of any kind in any benefit under this Plan shall be transferable or assignable by any Participant or by any beneficiary or be subject to anticipation, adjustment, alienation, encumbrance, garnishment, attachment, execution, or levy of any kind.
- 13.3 Communication to Employees. The Plan Administrator shall furnish to each Participant and each Beneficiary receiving benefits under the Plan a copy of a summary plan description and a summary of any material modifications thereof at the time and in the manner prescribed by law.
- 13.4 Number and Gender. The masculine pronoun shall include the feminine pronoun, and the singular number shall include the plural number, unless the context of the Plan requires otherwise.
- 13.5 Construction. The terms of the Plan shall be construed under the laws of the State of Texas except to the extent such laws are preempted by federal law.

- 13.6 Not a Contract of Employment. The adoption of this Plan by an Employer shall not constitute a contract of employment between the Employer and any Employee.
- 13.7 Indemnity. The Company shall indemnify all those to whom it has delegated fiduciary duties against any and all claims, loss, damages, expense, and liability arising from their responsibilities in connection with the Plan, unless the same is determined to be due to gross negligence or willful misconduct. Plan benefits shall be provided only from the Trust Fund, and neither the Employer nor the Trustee guarantees or assumes any liability that the Trust Fund will at any time be sufficient therefor.
- 13.8 Change of Beneficiary. A Participant may change the Beneficiary designation under the Plan at any time by submitting a Change of Beneficiary request in accordance with the procedures established from time to time by the Administrative Board.
- 13.9 Applications to the Administrative Board. Correspondence shall be sent to such address as the Administrative Board shall designate from time to time or to the following address:

Administrative Board for the American General Agents' and Managers' Thrift Plan c/o Plan Administrator 2929 Allen Parkway Houston, Texas 77019

13.10 Top-Heavy Rules.

(A) For purposes of this Section 13.10, the following language defines a Top-Heavy Plan:

(1) In general:

- (a) Plans Not Required to be Aggregated. Except as provided in Subsection 13.10(A)(1)(b), the term "Top-Heavy Plan" means, with respect to any plan year, any defined contribution plan if, as of the Determination Date; the aggregate of the accounts of Key Employees under the plan exceeds 60% of the aggregate of the accounts of all Employees under such plan.
- (b) Aggregated Plans. Each plan of an Employer required to be included in an Aggregation Group shall be treated as a Top-Heavy Plan if such group is a Top-Heavy Group.

- (2) Aggregation. For purposes of this Section 13.10:
 - (a) Aggregation Group:
 - (i) Required Aggregation. The term "Aggregation Group" means:
 - 1) each plan of the Employer in which a Key Employee is a Participant; and
 - 2) each other plan of the Employer which enables any plan described in the preceding Subsection (I) to meet the requirements of section 401(a)(4) or 410 of the Code.
 - (ii) Permissive Aggregation. The Employer may treat any plan not required to be included in an Aggregation Group under Subsection 13.10(A)(2)(a)(i) as being part of such group, if such group would continue to meet the requirements of sections 401(a)(4) and 410 of the Code with such plan being taken into account.
- (b) Top-Heavy Group. The term "Top-Heavy Group" means any Aggregation Group if:
 - (i) the sum (as of the Determination Date) of:
 - 1) the present value of the cumulative accrued benefits for Key Employees under all defined benefit plans included in such group; and
 - 2) the aggregate of the accounts of Key Employees under all defined contribution plans included in such group;
 - (ii) exceeds 60% of a similar sum determined for all Employees.

For purposes of making the foregoing determination, an Employee's account balance as of a Determination Date shall be valued as of the most recent date within the 12-month period prior to such Determination Date as of which the Trust Fund was valued, and the net income (or loss) thereof allocated to the Participants' accounts.

- (3) Distributions During Last Five Years Taken into Account. For purposes of determining (a) the present value of the cumulative accrued benefit for any Employee, or (b) the amount of the account of any Employee, such present value or amount shall be increased by the aggregate distributions made with respect to such Employee under the plan during the five-year period ending on the Determination Date. The preceding sentence shall also apply to distributions under a terminated plan which, if it had not been terminated, would have been required to be included in an Aggregation Group.
- (4) Other Special Rules. For purposes of this Section 13.10:
 - (a) Rollover Contributions to a Plan Not Taken into Account. Except to the extent provided in Treasury Regulations, any rollover contribution (or similar transfer) initiated by the Employee and made after December 31, 1983 to a plan shall not be taken into account with respect to the transferee plan for purposes of determining whether such plan is a Top-Heavy Plan (or whether any Aggregation Group which includes such plan is a Top-Heavy Group).
 - (b) Benefits Not Taken into Account if Employee Ceases to be Key Employee. If any individual is a non-Key Employee with respect to any plan for any plan year, but such individual was a Key Employee with respect to such plan for any prior plan year, any accrued benefit for such Employee (and the account of such Employee) shall not be taken into account.
 - (c) Determination Date. The term
 "Determination Date" means, with respect to any plan
 year:
 - (i) the last day of the preceding plan year; or
 - $\mbox{(ii)}$ in the case of the first plan year of any plan, the last day of such plan year.
 - (d) Years. To the extent provided in Treasury Regulations, this Section shall be applied on the basis of any year specified in such regulations in lieu of plan years.
 - (e) Benefits Not Taken into Account if Employee Not Employed for Last Five Years. If any individual has not received any compensation from any Employer maintaining the Plan (other than benefits under the Plan) at any time during the five-year period ending on

the Determination Date, any accrued benefit for such individual (and the account of such individual) shall not be taken into account.

(f) For purposes of determining whether the Plan is Top Heavy, an individual is a "Key Employee" if such person is an individual described in section 416(i)(1) of the Code and the Treasury Regulations promulgated thereunder and, for purposes of applying such provisions, an individual's "compensation" (as such term is used in such section and such regulations) shall be deemed to be equal to such individual's Remuneration, as defined in Subsection 4.5(B)(2).

Should the Plan become Top Heavy, the following provisions will apply:

(B) A Participant shall have a Nonforfeitable Interest in such Participant's Employer Contribution Account in accordance with Section 6.4 or the following table, whichever is greater:

Years of Service	Nonforfeitable Percentage
2	20 40
4	60
5	80
6	100

(C) The Employer shall contribute to the Plan for such Plan Year on behalf of each Participant who is not a Key Employee and who has not terminated employment as of the last day of such Plan Year an amount equal to:

(1) the lesser of (a) 3%, of such Participant's Remuneration as described in Subsection 4.5(B)(2) for such Plan Year, or (b) a percent of such Participant's Remuneration as described in Subsection 4.5(B)(2) for such Plan Year equal to the greatest percent determined by dividing for each Key Employee the amount allocated to such Key Employee's accounts for such Plan Year, by such Key Employee's Remuneration not in excess of \$200,000 for such Plan Year; reduced by

(2) the amount allocated to such Participant's accounts for such Plan Year.

The minimum contribution required to be made for a Plan Year pursuant to this Subsection (C), for a Participant employed on the last day of such Plan Year, shall be made

regardless of whether such Participant is otherwise ineligible to receive an allocation of the Employer's contributions for such Plan Year. Notwithstanding the foregoing, no contribution shall be made pursuant to this Subsection (C) for a Plan Year, with respect to a Participant who is a participant in a defined benefit plan sponsored by the Employer or a Controlled Entity, if such Participant accrued under such defined benefit plan sponsored by the Employer or a Controlled Entity (for the Plan Year of such plan ending with or within the Plan Year of this Plan) a benefit which is at least equal to the benefit described in section 416(c)(1) of the Code. Further notwithstanding the foregoing, and to the extent permitted by applicable law and Treasury Regulations, no contribution shall be made pursuant to this Subsection (C) for a Plan Year, with respect to a Participant who is a participant in another defined contribution plan sponsored by the Employer or a Controlled Entity, if such Participant receives under such other defined contribution plan (for the Plan Year of such plan ending with or within the Plan Year of this Plan) a contribution which is equal to or greater than the minimum contribution required by section 416(c)(2) of the Code. Notwithstanding the foregoing, if the Plan is deemed to be Top Heavy for a Plan Year, the Employers' contribution for such Plan Year pursuant to this Subsection (C) shall be increased by substituting "4%" in lieu of "3%" in Subsection 13.10(C)(1) hereof to the extent that the Board of Directors determine to so increase such contribution to comply with the provisions of section 416(h)(2) of the Code. Plan provisions to the contrary notwithstanding, the portion of a Participant's Employer Contribution Account which is attributable to the minimum contributions required to be made for a Plan Year pursuant to this Subsection (C) shall not be subject to forfeiture by reason of such Participants' withdrawal of basic Employee contributions; provided, however, that such Participant is not a Key Employee.

- (D) The Base Pay of any Participant taken into account under the Plan shall not exceed \$160,000 (adjusted automatically to reflect any amendments to section 401(a)(17) of the Code and any cost-of-living increases authorized by section 401(a)(17) of the Code) (or such other amount as shall be prescribed by Treasury Regulations).
- (E) All other requirements of section 416 of the Code and Treasury Regulations thereunder shall be complied with.
- (F) If the Plan has been deemed to be Top Heavy for one or more Plan Years and thereafter ceases to be Top Heavy, the provisions of this Section 13.10 shall cease to apply to the Plan effective as of the Determination Date on which it is determined to no longer be Top Heavy. Notwithstanding the foregoing, the nonforfeitable percentage of each Participant who is a Participant on such Determination Date, or who has terminated employment but has not incurred a Period of Severance equal to at least five consecutive One-Year Breaks in Service as of such Determination Date, shall not be reduced and, with respect to each Participant who has completed five or more years of service with the Employer on such Determination Date, the nonforfeitable percentage of each such Participant shall continue to be determined in accordance with the schedule set forth in Subsection 13.10(C).

13.11 Qualified Domestic Relations Orders.

- (A) The prohibitions of Section 13.2 shall not apply to an assignment or alienation of benefits, a distribution of benefits, or the creation or recognition of other rights on behalf of a spouse, former spouse, child, or other dependent (referred to herein as "Alternate Payee") of a Participant pursuant to any Qualified Domestic Relations Order.
- (B) A "Qualified Domestic Relations Order" shall mean any judgment, decree, or order (including approval of a property settlement agreement) which is made pursuant to a state domestic relations law (including a community property law) and which relates to the provision of child support, alimony payments, or marital property rights of an Alternate Payee.
 - (C) To be qualified, such order must specify the following

facts:

- (1) the name and the last known mailing address (if any) of the Participant, and the name and mailing address of each Alternate Payee covered by the order;
- (2) the amount or percentage of the Participant's benefits to be paid by the Plan to each such Alternate Payee, or the manner in which such amount or percentage is to be determined;
- (3) the number of payments or period to which such order applies; and $% \left(1\right) =\left(1\right) \left(1$
 - (4) each Plan to which such order applies.
- (D) Such order must not require a Plan to do the following:
- (1) provide increased benefits (determined on the basis of actuarial value);
- (2) pay benefits to an Alternate Payee which are required to be paid to another Alternate Payee under another order previously determined to be a Qualified Domestic Relations Order; or
- (3) provide any type or form of benefit, or any option, not otherwise provided under the Plan.
- (E) In the case of any payment before a Participant has separated from service, a domestic relations order shall not be treated as failing to meet the requirements of the preceding Subsection (3) solely because such order requires that payment of benefits be made to an Alternate Payee:

- (1) prior to the Participant's "earliest retirement age" as such term is defined in section 414(p)(4)(B) of the Code:
- (2) as if the Participant had retired on the date on which such payment is to begin under such order; and
- $\hbox{(3) in any form in which such benefits may}\\ \mbox{be paid under the Plan to the Participant.}$
- (F) Procedures. The Plan Administrator shall establish reasonable procedures to determine whether domestic relations orders constitute Qualified Domestic Relations Orders, meeting the requirements set out in Subsection 13.10(B) above, and procedures to administer distributions under such orders.
 - (1) The Plan Administrator shall promptly notify the Participant and any Alternate Payee of the receipt of any domestic relations order and advise them of the Plan's procedures for determining the order's qualified status.
 - (2) Within a reasonable time after receiving an order, the Plan Administrator must determine whether it is qualified and then promptly notify the Participant and Alternate Payee(s) of the decision.
 - (3) If there is an issue as to whether an order is qualified, payments which would otherwise be paid under the order shall be deferred while this determination is being made (by the Plan Administrator, by a court of competent jurisdiction, or otherwise). The Plan Administrator shall segregate in a separate account in the Plan or in an escrow account the amounts which would have been payable to the Alternate Payee during such period if the order had been determined to be a Qualified Domestic Relations Order.
 - (4) If within 18 months of deferral of payment the order (or modification thereof) is determined to be a Qualified Domestic Relations Order, the Plan Administrator shall pay the segregated amounts (plus any interest thereon) to the person or persons entitled thereto.
 - (5) If within 18 months of deferral of payments it is determined that the order is not a Qualified Domestic Relations Order, or the issue as to whether such order is a Qualified Domestic Relations Order is not resolved, then the Plan Administrator shall pay the segregated amounts (plus any interest) to the person or persons who should have been entitled to such amounts if there had been no order. Any determination that an order is a Qualified Domestic Relations Order which is made after the close of the 18-month period shall be applied prospectively only. The Plan shall not be liable to the Alternate Payee for payments before the order is determined to be qualified.

- (6) The Plan Administrator may treat any domestic relations order entered into before January 1, 1985 as a Qualified Domestic Relations Order, even if it does not meet the requirements of Subsections 13.11(B)-(E). The Plan Administrator shall treat any domestic relations order entered into before January 1, 1985 as a Qualified Domestic Relations Order if payments are being made from the Plan pursuant to that order.
- (G) Forfeitures. Forfeitures are not permitted of amounts which are payable to an Alternate Payee under a Qualified Domestic Relations Order during any period in which the Alternate Payee cannot be located, unless full reinstatement is made when the Alternate Payee is located.
- (H) Death Benefits. The former spouse of a Participant shall be treated as a surviving spouse for purposes of Section 2.3 to the extent provided in any Qualified Domestic Relations Order.
- 13.12 Uniformed Services Employment and Reemployment Rights Act Requirements. Notwithstanding any provision of the Plan to the contrary, contributions, benefits and service credit with respect to qualified military service will be provided in accordance with section 414(u) of the Code.

5

IN WITNESS WHEREOF, the parties hereto have executed this American General Agents' and Managers' Thrift Plan, on this 28th day of June, 2001.

AMERICAN GENERAL CORPORATION

/S/ Gary D. Reddick

Gary D. Reddick Executive Vice President -Administration and Insurance Operations COMMOLOCO THRIFT PLAN

July 1, 2001 Restatement

TABLE OF CONTENTS

ARTICLE I - PURPOSE	1
ARTICLE II - DEFINITIONS	2
2.1 Administrative Board	2
2.2 Base Pay	2
2.3 Beneficiary	2
2.4 Board of Directors	2
2.5 Certified Public Accountant	3
2.6 Change Date	3
2.7 Code	3
2.8 Company	3
2.9 Company Stock	3
2.10 Controlled Entity	3
2.11 Effective Date	3
2.12 Employee	3
2.13 Employee Contribution Account	3
2.14 Employer	3
2.15 Employer Contribution Account	3
1 ,	4
2.16 Employment Commencement Date	4
	4 5
2.18 Hour of Service	5 5
2.19 Investment Funds	5 5
2.20 Leased Employee	5
2.21 Long-Term Disability Program	5
2.22 Nonforfeitable Interest	5
2.23 Non-Highly Compensated Participant	5
2.24 Normal Retirement Date	5
2.25 One-Year Break in Service	6
2.26 Participant	6
2.27 Period of Separation	6
2.28 Period of Service	6
2.29 Period of Severance	7
2.30 Plan	8
2.31 Plan Administrator	8
2.32 Plan Year	8
2.33 Prior Employee Contribution Account	8
2.34 Reemployed Individual	8
2.35 Rollover Account	8
2.36 Rollover Contributions	8
2.37 Secretary	8
2.38 Stock Fund	8

2.39 Total Disability	8
2.40 Transfer Date	9
2.41 Treasury Regulations	9
2.42 Trust	9
2.43 Trustee	9
2.44 Trust Fund	9
2.45 Withdrawal Date	9
2.45 WICHUI AWAI Date	9
ARTICLE III-PARTICIPATION	10
3.1 Eligibility	10
3.2 Reemployed Individuals	10
3.3 Participation	10
ARTICLE IV - CONTRIBUTIONS AND ALLOCATIONS	11
4.1 Employer Contributions	11
4.1a Employer Safe Harbor Contributions	12
4.2 Employee Contributions	12
4.3 Forfeitures	13
4.4 Investment and Allocation of Contributions	14
4.5 Limitation on Allocation of Contributions and Forfeitures	15
4.6 Rollover Contributions	17
4.0 KOLIOVEI CONTITUULIONS	17 17
4.7 Voting, Tendering, and Exchanging of Company Stock	
4.8 Restrictions on Employee Contributions	19
4.9 Restrictions on Employer Contributions	19
4.10 Excess Contributions	20
ARTICLE V - VALUATION OF ACCOUNTS	21
ARTICLE VI-VESTING	22
6.1 Normal Retirement	22
6.2 Plan Termination, Partial Plan Termination, or Complete Plan	
Discontinuance of Employer Contributions	22
6.3 Vesting of Employee Contributions	22
6.4 Vesting of Employer Contributions	22
6.5 Vesting After a Period of Severance	23
6.6 Vesting Before a Period of Severance of Five Consecutive	
One-Year Breaks in Service	23
6.7 Election to Make a Withdrawal	23
6.8 Transfer Between Plans	23
	23
6.9 Independent Life Transferred Employees	23
ARTICLE VII -BENEFITS	24
7.1 Normal Retirement	24
7.2 Disability	24
7	

	7.3 Death	24
	7.4 Termination of Service	24
	7.5 Valuation Date to be Used	24
	7.6 Withdrawals	25
	7.7 Loans	27
	7.7 Loans	21
ARTICLE	VIII - COMMENCEMENT OF BENEFITS	28
	8.1 Benefits After Normal Retirement Date	28
	8.2 Certain Benefits Before Normal Retirement Date	28
	8.3 Termination of Service Before Normal Retirement Date	28
	8.4 Commencement of Benefits	28
	0.4 Commencement of Benefits	20
ARTICLE	IX-DISTRIBUTION OF BENEFITS	30
	9.1 Mode of Benefit Payments	30
	9.2 Stock Certificates	30
	9.3 Automatic Distributions	30
	9.4 Unclaimed Benefits	30
	9.5 Direct Rollovers	31
	JIO DIFECT ROLLOVETS	01
ARTICLE	X - PLAN AMENDMENT AND TERMINATION	32
	10.1 Amendment of the Plan	32
	10.2 Communication of Amendments	32
	10.3 Termination or Discontinuance of Employer Contributions	32
	10.4 Acceptance or Rejection of Amendment Employers	32
	10.5 Termination of the Plan by an Employer	32
	10.6 Notice of Terminating Employer	33
	10.0 Notice of Terminating Emproyer	33
ARTICLE	XI - ADMINISTRATION	34
	11.1 Named Fiduciaries	34
	11.2 Appointment of the Trustee	34
	11.3 Trustee's Powers and Duties	34
	11.4 Administrative Expenses	34
	11.5 Plan Administrator's Powers and Duties	34
	11.6 Named Fiduciary's Powers and Duties	35
	11.7 Allocation of Functions	35
ARTTCLE	XII - ADOPTION AND WITHDRAWAL	37
	12.1 Procedure for Adoption	37
	12.2 Effect of Adoption	37
	12.3 Withdrawal	37
	IZ.3 WILIIUI AWAI	37
ARTICLE	XIII - MISCELLANEOUS	38
	13.1 Merger of this Plan with Another Plan	38
	13.2 Assignment and Alienation of Benefits	38
	13.3 Communication to Employees	38
	20.0 00	

	Number and Gender	
13.5	Construction	38
13.6	Not a Contract of Employment	38
13.7	Indemnity	39
13.8	Change of Beneficiary	39
13.9	Applications to the Administrative Board	39
13.10	Top-Heavy Rules	39
13.11	Qualified Domestic Relations Orders	43
13.12	Uniformed Services Employment and Reemployment Rights Act	
	Requirements	45

(iv)

ARTICLE I

PURPOSE

Effective July 1, 2001 CommoLoCo, Inc. adopts the restated Plan, as set forth herein, to replace the American General Employees' Thrift and Incentive Plan previously in effect. The new name of the Plan shall be the CommoLoCo Thrift Plan ("Plan")

The Plan was instituted by the Company to encourage systematic savings and the accumulation of financial reserves by Employees of the Company and its subsidiaries, and to give Employees an opportunity to acquire an ownership interest in American General Corporation as well as enabling Employees to reap greater direct benefits from the Company's success. The Plan is carried out through Employee contributions deducted from the payroll on a pre-tax basis and contributions by the Company or its subsidiaries. Company contributions are invested in American General common stock purchased by a Trustee responsible for administering the Trust Fund. Employee contributions are invested by the Trustee at each Participant's discretion in American General common stock or in other designated investment funds.

The Plan and Trust are intended to meet the requirements of sections 401(a) and 501(a) of the Internal Revenue Code of 1986, and section 1165(a) and (e) of the Puerto Rico Internal Revenue Code of 1994 as amended when applicable, and the requirements of the Employee Retirement Income Security Act of 1974, as amended.

This Plan and the separate related Trust forming a part hereof are established and shall be maintained for the exclusive benefit of the eligible Employees of the Employers and their Beneficiaries. No part of the Trust Fund can ever revert to the Employers, except as provided in Article X, or be used for or diverted to purposes other than the exclusive benefit of the Employees of the Employers and their Beneficiaries.

ARTICLE II

DEFINITIONS

As used in this Plan, the following words and phrases shall have the meanings set forth below, unless the context clearly indicates otherwise.

- 2.1 Administrative Board. A board composed of at least three Company officers or Employees appointed by the Board of Directors to administer the Plan, located at Company headquarters at 2929 Allen Parkway, Houston, Texas 77019.
- 2.2 Base Pay. The compensation of the Employee as stated on Employer payroll records, such amount to exclude, however, any pay for overtime (which shall be deemed to refer as well to any shift differential payments), discretionary bonuses (which shall be deemed to refer as well to educational awards, educational reimbursements, instructor fees, referral awards, moving expenses, mortgage assistance, personal use of an Employer-owned vehicle, ACE awards, or workers' compensation payments), severance payments, and Employer contributions to this or any other deferred or noncash compensation program; provided, however, that any salary reduction amounts contributed on behalf of the Employee under a flexible benefits program pursuant to section 125 of the Code shall be included; and, provided, further, that salary reduction amounts contributed on behalf of the Employee under this Plan shall be included. Notwithstanding the foregoing, the Base Pay of any Employee taken into account under the Plan for any calendar year may not exceed \$150,000 (with such amount to be adjusted automatically to reflect any cost-of-living increases authorized by section 401(a)(17) of the Code).

2.3 Beneficiary.

- (A) The Beneficiary of a Participant shall be:
 - (1) the surviving spouse, if any, of the Participant; or
 - (2) if there is no surviving spouse, or if the surviving spouse has executed a consent in accordance with Subsection 2.3(B), the person or persons designated in writing by the Participant; or
 - (3) if there are no persons described in the preceding Subsections living at the date of the Participant's death, the Participant's estate.
- (B) The consent referred to in Subsection 2.3(A)(2) must be in writing, must acknowledge the effect of the Participant's designation of a person, other than the spouse, to receive the Participant's Nonforfeitable Interest upon the Participant's death, and must be witnessed by a Plan representative or a notary public.
 - $2.4\ \mbox{Board}$ of Directors. The Board of Directors of the Company.

- 2.5 Certified Public Accountant.
- (A) A person who is a certified public accountant, certified by a regulatory authority of a state;
- (B) a person who is a licensed public accountant, licensed by a regulatory authority of a state; or
- (C) a person certified by the Secretary of Labor as a qualified public accountant.
 - 2.6 Change Date. The first day of any pay period as designated by a $\operatorname{Participant}$.
 - 2.7 Code. The Internal Revenue Code of 1986, as amended from time to time
 - 2.8 Company. CommoLoCo, Inc. and its successors.
 - $2.9\ \mbox{Company Stock}.$ The common stock of American General Corporation and/or its successors.
- 2.10 Controlled Entity. A corporation or other trade or business which is not an Employer hereunder, but which, together with an Employer, is "under common control" within the meaning of section 1.414(c)-2 of the Treasury Regulations.
- 2.11 Effective Date. January 1, 1997, as to this amendment and restatement of the Plan.
- 2.12 Employee. Any Employee of an Employer who is a resident of the United States or the U.S. Virgin Islands, excluding, however, Sales Employees (sometimes known as Home Service Representatives, Sales Representatives, Sales Managers, Field Representatives, Agents or Home Service Agents, or Sales or District Managers), Full-Time Life Insurance Agents, and any other field representatives with Flexi-Master Contracts, provided that the holder of a Flexi-Master Contract is deemed to be an employee under section 7701(a)(20) of the Code. Such term shall not include Leased Employees.
- 2.13 Employee Contribution Account. A separate account maintained for each Participant to which both basic and additional Employee contributions made on behalf of the Participant, and earnings and investment gains thereon, are credited, and which is invested in any of the separate investment funds.
- 2.14 Employer. The Company and any other corporation which shall adopt this Plan pursuant to Article XII, and the successor, if any, to such corporation.
- 2.15 Employer Contribution Account. A separate account maintained for a Participant consisting of cash, dividends payable, and Company Stock purchased by Employer contributions and reinvested dividends, and which may be invested in any of the separate investment funds after the Participant attains age sixty. Notwithstanding the foregoing, a separate subaccount of

the Employer Contribution, Account known as the 'Employer Contribution Account Subaccount' shall be established and maintained for a Participant respecting Employer Contributions made to the Plan from and after January 1, 1999, and earnings thereon. Except as provided in Section 6.4 relating to vesting and Section 7.6 relating to withdrawals, a Participant's Employer Contribution Account Subaccount shall be considered as part of the Participant's Employer Contribution Account for all purposes of the Plan.

- $2.16\ \mbox{Employment}$ Commencement Date. The date on which an Employee first performs an Hour of Service.
- 2.17 Highly Compensated Participant. Each Participant or Former Participant who is an Employee who performs services during the Plan Year for which the determination of who is highly compensated is being made (the 'Determination Year') and who:
 - (A) is a five-percent owner of the Employer (within the meaning of section 416(i)(1)(A)(iii) of the Code) at any time during the Determination Year or the twelve-month period immediately preceding the Determination Year (the 'Look-Back Year'); or
 - (B) For the Look-Back Year:
 - (i) receives compensation (within the meaning of section 414(q)(4) of the Code; 'compensation' for purposes of this Paragraph) in excess of \$80,000 (with such amount to be adjusted automatically to reflect any cost-of-living adjustments authorized by section 414(q)(1) of the Code) during the Look-Back Year; and
 - (ii) if the Administrative Board elects the application of this clause for such Look-Back Year, is a member of the top 20% of Employees for the Look-Back Year (other than Employees described in section 414(q)(5) of the Code) ranked on the basis of compensation received during the year.

For purposes of the preceding sentence, (i) all employers aggregated with the Employer under section 414(b), (c), (m), or (o) of the Code shall be treated as a single employer, (ii) a former Employee who had a separation year (generally, the Determination Year such Employee separates from service) prior to the Determination Year and who was an active Highly Compensated Participant for either such separation year or any Determination Year ending on or after such Employee's fifty-fifth birthday shall be deemed to be a Highly Compensated Participant, and (ii) the Committee may elect, in accordance with the provisions of applicable Treasury regulations, rulings and notices, to make the Look-Back Year calculation for a Determination Year on the basis of the calendar year

ending with or within the applicable Determination Year (or, in the case of a Determination Year that is shorter than twelve months, the calendar year ending with or within the twelve-month period ending with the end of the applicable Determination Year). To the extent that the provisions of this Paragraph are inconsistent or conflict with the definition of a 'highly compensated employee' set forth in section $414(\mathfrak{q})$ of the Code and the Treasury regulations thereunder, the relevant terms and provisions of section $414(\mathfrak{q})$ of the Code and the Treasury regulations thereunder shall govern and control.

2.18 Hour of Service. An Hour of Service is each hour for which an Employee is directly or indirectly paid, or entitled to payment, by the Employer or a Controlled Entity for the performance of duties or for reasons other than the performance of duties. Such Hours of Service shall be credited to the Employee for the period in which such duties were performed or in which occurred the period during which no duties were performed. An Hour of Service also includes each hour, not credited above, for which back pay, irrespective of mitigation of damages, has been either awarded or agreed to by the Employer or a Controlled Entity. These Hours of Service shall be credited to the Employee for the period to which the award or agreement pertains rather than the period in which the award, agreement or payment is made.

The number of Hours of Service to be credited to an Employee for any period shall be governed by section 2530.200b-2(b) and (c) of the Labor Department Regulations relating to the Employee Retirement Income Security Act of 1974, as amended.

- 2.19 Investment Funds. Investment funds made available from time to time for the investment of plan assets as described in Article IV.
- 2.20 Leased Employee: Each person who is not an employee of the Employer or a Controlled Entity but who performs services for the Employer or a Controlled Entity pursuant to an agreement (oral or written) between the Employer or a Controlled Entity and any leasing organization, provided that such person has performed such services for the Employer or a Controlled Entity or for related persons (within the meaning of section 144(a)(3) of the Code) on a substantially full-time basis for a period of at least one year and such services are performed under primary direction or control by the Employer or a Controlled Entity.
- 2.21 Long-Term Disability Program. The American General Long-Term Disability Plan for Employees, the American General Long-Term Disability Plan for Executives, the American General Life and Accident Sales Employees' Non-Occupational Disability Income Plan, and any workers' compensation plan, program, or fund sponsored, maintained, or paid into by the Company or the other Employers for the benefit of its Employees and Career Agents, whichever may be applicable to the Participant at the relevant time.
- 2.22 Nonforfeitable Interest. The unconditional and legally enforceable right to which the Participant or Beneficiary (whichever is applicable) is entitled in the Participant's entire Employee Contribution Account balance and entire Prior Employee Contribution Account balance and in the percentage of the Participant's Employer Contribution Account balance which has vested pursuant to Article VI.

- 2.23 Non-Highly Compensated Participant. "Non-Highly Compensated Participant" shall mean any Participant or former Participant who is not a Highly Compensated Participant.
- 2.24 Normal Retirement Date. The first day of the month coincident with or next following a Participant's 65th birthday.
 - 2.25 One-Year Break in Service.
- (A) A 12-consecutive-month Period of Severance during which an Employee does not perform at least one hour of service.
- (B) Solely for the purpose of determining whether or not a Participant has a One-Year Break in Service, the Plan shall treat as an hour of service each hour of service which would otherwise have been credited to the Participant but for an absence beginning on or after January 1, 1985, for one of the following reasons:
 - (1) pregnancy of the Participant;
 - (2) birth of a child of the Participant;
 - (3) placement of a child with the Participant in connection with an adoption proceeding; or
 - (4) caring for the child immediately following such birth or placement.

If those hours cannot be determined, the Plan shall treat eight hours per day of such absence as hours of service. The hours to be treated as hours of service because of such absence shall not exceed 501.

The hours required to be credited under (1)-(4) of this Subsection 2.24(B) shall be credited in the year in which such absence begins if it is necessary to prevent a break in service in that year; otherwise such hours shall be credited to the succeeding year.

No credit shall be given under this Subsection 2.24(B) unless the Employee furnishes to the Committee such information as it may reasonably require to establish that the absence is for one of the reasons listed in (1)-(4) of this Subsection 2.24(B), and to establish the number of days of such absence.

- $2.26\ Participant.$ An Employee who satisfies the eligibility requirements of Article III and has enrolled in the Plan.
- 2.27 Period of Separation. A period of time commencing with the date an Employee separates from service and ending with the date such Employee resumes employment with the Employer.
- 2.28 Period of Service. For purposes of determining an Employee's initial or continued eligibility to participate in the Plan or the vested interest in such individual's Employer Contribution Account, an Employee shall be credited for the time period commencing with his

Employment Commencement Date, including years of service with the Employer in a category of Employees excluded from the Plan, and ending on the date his Period of Severance begins. A Period of Service for these purposes includes a Period of Separation of less than 12 consecutive months. In the case of an Employee who separates from service and later resumes employment with the Employer, the Period of Service prior to resumption of employment shall be aggregated only if such Employee is a Reemployed Individual.

In the case of a person who becomes an Employee, after being a "leased employee" under section 414(n) of the Code, the Employment Commencement Date will be deemed to be the date such Employee first performed services for the Employer as a "leased employee" or such other date as required by Treasury Regulations.

- (A) Periods of Service prior to June 29, 1991, which would not have been credited under the rules of the previous plan or its amendments or a predecessor plan. Such periods shall be determined according to provisions of the previous plan or predecessor plan, including rules which require minimum hours or other length of service or contributions by Participants;
- (B) Periods of Service prior to the date the Employer adopts this Plan or a predecessor plan; and
- (C) Periods of Service in which an Employee declined to contribute to the Plan or a predecessor plan prior to the date such Employee first commences participation in the Plan.

It is specifically provided that for purposes of determining Period of Service, service with a Controlled Entity shall be included.

- $\,$ 2.29 Period of Severance. A period of time commencing with the earlier of:
- (A) the date an Employee separates from service by reason of quitting, retirement, death, or discharge; or
- (B) the date 12 months after the date an Employee separates from service, including severance for reasons other than quitting, retirement, death, or discharge; and ending, in the case of an Employee who separates from service by reason other than death, with the date such Employee resumes employment with the Employer.
 - (C) An absence beginning on or after January 1, 1985, because of:
 - pregnancy of the Participant;
 - (2) birth of a child of the Participant;

- (3) placement of a child with the Participant in connection with an adoption proceeding; or
- (4) caring for the child immediately following such birth or placement;

shall not be considered a Period of Severance except as permitted by Treasury Regulations.

- (D) In order for the preceding Subsection 2.28(C) to apply, a Participant must furnish to the Administrative Board such information as it may reasonably require to establish that the absence is for one of the reasons listed in (1)-(4) of the preceding Subsection 2.28(C), and to establish the number of days of such absence.
- $2.30\ \text{Plan.}$ This document, the provisions of the Trust agreement, and any amendments to either.
- 2.31 Plan Administrator. The Administrative Board unless and until it designates such other person or persons. A member of the Administrative Board may resign or may be replaced by the Board of Directors at any time. No bond or other security shall be required of any member of the Administrative Board unless otherwise required by law.
- 2.32 Plan Year. The 12-month period beginning January 1 and ending December 31.
- 2.33 Prior Employee Contribution Account. A separate account maintained for each Participant who made Employee contributions to the Plan prior to June 29, 1991, to which such Employee contributions, and earnings and investment gains thereon, are credited, and which is invested in any of the separate investment funds.
- $2.34\ Reemployed\ Individual.$ A person who, after having separated from service, resumes employment:
 - (A) with a Nonforfeitable Interest in such individual's Employer Contribution Account; or
 - (B) with no such Nonforfeitable Interest, and who resumes employment either (1) before a One-Year Break in Service or (2) after a One-Year Break in Service but before the latest Period of Severance equals or exceeds the greater of (a) such individual's Period of Service or (b) five consecutive years of a Period of Service.
- 2.35 Rollover Contribution Account. An separate account maintained for each Participant who has made Rollover Contributions pursuant to Section 4.6 to which such Rollover Contributions and earnings and investment gains thereon, are credited, and which is invested any of the separate Investment Funds.
- $2.36\ \text{Rollover}$ Contributions. Contributions made by an Employee pursuant to Section 4.6.
 - 2.37 Secretary. The Secretary of the Treasury of the United States.

- 2.38 Stock Fund. The separate investment fund forming a part of the Trust which is invested in, or held for investment in, Company Stock. Dividends applicable to Company Stock will be applied to the purchase of additional Company Stock.
- 2.39 Total Disability. The definition of disability as used under the Company's Long Term Disability Program as in effect at the inception of the Participant's disability, respecting the applicable plan, program, or fund pursuant to which the Participant is entitled to receive disability benefits.
- 2.40 Transfer Date. The first date on which the New York Stock Exchange is open for business coinciding with or next following receipt by the Plan Administrator of a transfer request from the Participant in accordance with the procedures established from time to time by the Administrative Board.
- 2.41 Treasury Regulations. Regulations issued from time to time by the Secretary or the Secretary's delegate.
- 2.42 Trust. The agreement between the Trustee and the Employer entered into for the purpose of holding, managing, and administering all property held by the Trustee for the exclusive benefit of the Participants and their Beneficiaries.
- 2.43 Trustee. The person designated by the Company pursuant to Section 11.2 and in accordance with the Trust agreement, and any successor who is appointed pursuant to the terms of that Section.
- $2.44\ \mbox{Trust}$ Fund. All assets held by the Trustee pursuant to the terms of this Plan.
- 2.45 Withdrawal Date. The first date on which the New York Stock Exchange is open for business coinciding with or next following receipt by the Plan Administrator of a withdrawal request from the Participant in accordance with the procedures established from time to time by the Administrative Board.

ARTICLE III

PARTICIPATION

3.1 Age and Service.

- A. An Employee whose base pay is calculated on the basis of an annual, monthly, semi-monthly, bi-weekly or weekly salary or who is a regular Employee of American General Finance, Inc. or its subsidiaries shall be eligible to participate in this Plan upon the Employee's Employment Commencement Date.
- B. An Employee whose base pay is not calculated on the basis of an annual, monthly, semi-monthly, bi-weekly or weekly salary and who is not a regular Employee of American General Finance, Inc. or its subsidiaries shall be eligible to participate in this Plan upon the completion of a one-year Period of Service
- 3.2 Reemployed Individuals. A Reemployed Individual is eligible to participate in the Plan on the later of the date of reemployment by the Employer or the date described in Section 3.1.
- 3.3 Participation. Participation in the Plan shall commence as soon as administratively feasible following the date the eligibility requirement is satisfied, provided the Administrative Board has received a properly completed enrollment in accordance with the procedures established from time to time by the Administrative Board. A Reemployed Individual who was previously a Participant may recommence participation on the date of reemployment, provided such Participant again enrolls in the Plan in accordance with the procedures established from time to time by the Administrative Board on or prior to that date.

ARTICLE IV

CONTRIBUTIONS AND ALLOCATIONS

4.1 Employer Contributions. Simultaneously with basic Employee contributions, the Employer will make a contribution out of current or accumulated earnings, subject to the limits of Section 4.5, to the Employer Contribution Account of each Participant. The Employer's contribution to the Employer Contribution Account of each Participant will equal 100% of the Participant's basic Employee contribution as described in Section 4.2(A) not in excess of 3% of Base Pay plus 50% of the Participant's basic Employee contribution as described in Section 4.2(A) in excess of 3% but not in excess of 6% of Base Pay.

In addition to the Employer Contributions made pursuant to the paragraph above, for each Plan Year the Employer will make a contribution out of current or accumulated earnings, subject to the limits of section 415 of the Code, to the Employer Contribution Account of each Participant equal to the difference, if any, between (1) 100% of the Participant's basic Employee contribution as described in Section 4.2(A) not in excess of 3% of Base Pay for such Plan Year, plus 50% of the Participant's basic Employee contribution as described in Section 4.2(A) in excess of 3% but not in excess of 6% of Base Pay for such Plan Year and (2) the Employer contributions made pursuant to the paragraph above for the Participant for such Plan Year.

If any Employer is prevented from making a contribution which it would otherwise have made because of inadequate current or accumulated earnings, then the amount of the contribution which such Employer was prevented from making shall be made for the benefit of the Employees of such Employer by the remaining Employers as follows:

- (A) if the Employer files a consolidated income tax return with the Company for that year for Federal income tax purposes, then such contribution shall be made in such proportions as the Company shall specify; or
- (B) if no consolidated income tax return is filed by the Employer with the Company for that year for Federal income tax purposes, each remaining Employer not so prevented from making a contribution shall make a supplemental contribution. Such contribution shall be in an amount equal to that portion of its total current income and accumulated earnings which the total contributions that one or more Employers were so prevented from making bears to the total current and accumulated earnings of all Employers having current or accumulated earnings. As used in the preceding sentence, total current income and accumulated earnings are calculated after deducting contributions which would have been without considering the additional supplemental contributions permitted by this Subsection (B).

If any Employer is prevented from making a contribution which it would otherwise have made, and part or all of such contribution is made by one or more other Employers, contributions so made shall be deductible for Federal income tax purposes by the Employer or Employers making such contributions. For the purpose of determining amounts which may be carried

forward and deducted in succeeding taxable years, contributions shall be deemed to have been made by the Employer(s) on behalf of whom such contributions were made.

Notwithstanding anything to the contrary herein, the Company's contributions are contingent upon the deductibility of such contributions under section 404 of the Code. To the extent that a deduction for contributions is disallowed, such contributions may be returned within one year after the date of disallowance. Finally, if Company contributions are made under a mistake of fact, such contributions may be returned to the Company within one year after the payment thereof.

4.1a Employer Safe Harbor Contributions. In addition to the Employer contributions made pursuant to Section 4.1, for each Plan Year, the Employer, in its discretion, may contribute to the Plan as a "safe harbor contribution" for such Plan Year out of current or accumulated earning the amounts necessary to cause the Plan to satisfy the restrictions set forth in Section 4.8 (with respect to certain restrictions on Employee contributions) and Section 4.9 (with respect to certain restrictions on Employer contributions). Amounts contributed in order to satisfy the restrictions set forth in Section 4.8 shall be considered "qualified matching contributions" (within the meaning of Treasury regulation Section 1.401(k)-1(g)(13) for purposes of such Section, and amounts contributed in order to satisfy the restrictions set forth in Section 4.9 shall be considered Employer contributions in accordance with the provisions of Section 4.1.

4.2 Employee Contributions.

(A) Basic Contributions. As a condition of participation in the Plan, a Participant must elect to have a basic Employee contribution made on his behalf for a Plan Year by entering into a salary reduction agreement wherein the Participant elects to defer an integral percentage of from 1% to 6% of Base Pay for future pay periods and have the Employer contribute the amount so deferred to the Plan.

(B) Additional Contributions. Each Participant who is having a basic Employee contribution of 6% of Base Pay made to the Plan on his behalf may also elect to have an additional contribution made on his behalf for a Plan Year by entering into a salary reduction agreement wherein the Participant elects to defer an integral percentage of from 1% to 10% of Base Pay for future pay periods and have the Employer contribute the amount so deferred to the Plan. Notwithstanding the foregoing, Highly Compensated Participants shall not be allowed to make additional Employee contributions if such contributions will adversely affect the restrictions under Section 4.8 of the Plan. The above restriction shall apply to Highly Compensated Participants, not heretofore considered to be highly compensated, as of the first day of the first pay period commencing after the determination that the Participants are considered to be highly compensated.

(C) Procedures. Base Pay for a Plan Year not so deferred as provided in Subsections (A) and (B) above shall be received by each such Participant in Cash. The reduction in a Participant's Base Pay for a Plan Year pursuant to such election under a salary reduction agreement shall be effected by payroll deductions each period within such Plan Year following the effective date of such agreement and will change automatically to reflect changes in Base Pay.

A Participant's salary reduction agreement shall remain in force and effect for all periods following the date of its execution until changed or suspended or until such Participant terminates his employment.

A Participant who has elected to have Employee contributions made to the Plan on his behalf may change his Employee contribution election (within the limitations described in Subsections (A) and (B) above), effective as of any Change Date, by communicating his new deferral election to his Employer in the manner and within the time period prescribed by the Administrative Board.

A Participant may cancel his Employee contribution election, effective as any Change Date, by communicating such cancellation to his Employer in the manner and within the time period prescribed by the Administrative Board. A Participant who so cancels his deferral election may resume deferrals, effective as of any Change Date, by communicating his new Employee contribution election to his Employer in the manner and within the time period prescribed by the Administrative Board.

Employee contributions shall be automatically suspended during periods of unpaid military leave or during a period of unpaid authorized leave of absence granted pursuant to a nondiscriminatory leave of absence program established by the Employer, but such Employee contributions shall automatically resume upon the Employee's return to active work.

- (D) Basic and additional Employee contributions shall be contributed by the Employer to the Plan out of current or accumulated earnings as soon as administratively practicable each payroll period.
- (E) Prior Employee Contributions. Basic and additional Employee contributions made by Participants on an after-tax basis prior to June 29, 1991, shall no longer be permitted. Such Employee contributions shall be held, maintained and administered in each such Participant's Prior Employee Contribution Account in accordance with the provisions of the Plan.
- 4.3 Forfeitures. Notwithstanding any provisions to the contrary, if a Participant whose Nonforfeitable Interest is less than 100% of such individual's Employer Contribution Account terminates employment or withdraws from participation and receives a distribution from such account, a separate account shall be established for such Participant's interest in the Plan as of the time of the distribution. At any relevant time prior to incurring a Period of Severance equal to at least five consecutive One-Year Breaks in Service, such Participant's nonforfeitable portion of such separate account shall be determined in accordance with the following formula:

$$X \cdot P (AB + (R \times D)) - (R \times D).$$

For purposes of applying the formula: "X" is the nonforfeitable portion of such separate account at the relevant time; "P" is the Participant's Nonforfeitable Interest at the relevant time; "AB" is the balance of such separate account at the relevant time; "R" is the ratio of the balance of such separate account at the relevant time to the balance immediately after the distribution;

and "D" is the amount of the distribution. The foregoing shall be applied to a Participant's Employer Contribution Account without regard to the Participant's Employer Contribution Account Subaccount.

Upon incurring a Period of Severance equal to at least five consecutive One-Year Breaks in Service, the forfeitable portion of a terminated Participant's account shall be forfeited and such forfeiture shall be available to reduce future Employer contributions.

Any Participant who terminates employment with an Employer on or after February 1, 1989, shall forfeit the forfeitable portion of such Participant's account on the earlier of (A) the distribution of the entire nonforfeitable portion of such Participant's account or (B) upon incurring a Period of Severance equal to at least five consecutive One-Year Breaks in Service.

Notwithstanding the preceding paragraph, if any former Participant shall be reemployed by an Employer before a Period of Severance equal to five consecutive One-Year Breaks in Service, and such former Participant had received a distribution of the entire vested interest prior to reemployment, such Participant's forfeited account shall be reinstated only if the full amount distributed to such individual is repaid before the earlier of (A) five years after the first date on which the Participant is subsequently reemployed by the Employer, or (B) the close of the first period of five consecutive One-Year Breaks in Service commencing after the distribution. In the event the former Participant does repay the full amount distributed, the undistributed portion of the Participant's account must be restored in full, unadjusted by any gains or losses occurring subsequent to the distribution.

- 4.4 Investment and Allocation of Contributions. Contributions will be deposited promptly with the Trustee for investment as follows:
 - (A) Employer Contribution Account. Employer contributions shall be invested in, or held for investment in, the Stock Fund by the Trustee. The Administrative Board will make a monthly allocation of shares of Company Stock purchased with Employer contributions to each Participant's Employer Contribution Account. Notwithstanding the foregoing, upon attainment of age 60, a Participant may elect to have his Employer Contribution Account invested in, or held for investment in, any one or more of the Stock Fund or other Investment Funds, in such increments as the Administrative Board may prescribe.
 - (B) Employee Contribution Account. Basic and additional Employee contributions shall be allocated to a Participant's Employee Contribution Account and shall be invested in, or held for investment in, any one or more of the Stock Fund or other Investment Funds, in such increments as the Administrative Board may prescribe, pursuant to a Participant's designation.
 - (C) Prior Employee Contribution Account. Employee contributions made prior to June 29, 1991, and allocated to a Participant's Prior Employee Contribution Account shall be invested in, or held for investment in, any one or more of the Stock Fund or other Investment Funds, in such increments as the Administrative Board may prescribe, pursuant to a Participant's designation.

- (D) Rollover Account. Rollover Contributions shall be allocated to a Participant's Rollover Account and shall be invested in, or held for investment in, any one or more of the Stock Fund or other Investment Funds, in such increments as the Administrative Board may prescribe, pursuant to a Participant's designation.
- (E) Investment Designations and Transfers. A Participant may change the investment for subsequent Employee contributions. Any such change shall be made as of the first date on which the New York Stock Exchange is open for business coinciding with or next following receipt by the Plan Administrator of a properly completed change request from the Participant in accordance with procedures established by the Administrative Board, and the frequency of such changes may be limited by the Administrative Board.

As of any Transfer Date, a Participant may transfer any part, in such increments as the Administrative Board may prescribe, or all of the amounts in such Participant's (1) Employee Contribution Account, (2) Prior Employee Contribution Account, (3) Rollover Account and (4) Employer Contribution Account following a Participant's 60th birthday among the separate investment funds.

If a separate investment fund is eliminated as an available investment option under the Plan, the American General Corporation Personnel Policy Committee upon recommendation by the Administrative Board shall designate an appropriate default separate investment fund for such eliminated separate investment fund in the event one or more substitute separate investment funds are not timely designated by a Participant.

- 4.5 Limitation on Allocation of Contributions and Forfeitures.
- (A) Contrary Plan provisions notwithstanding, the Annual Additions allocated to a Participant's accounts for any Limitation Year shall be limited to the Maximum Annual Additions for such Participant for such year.

If as a result of allocation of forfeitures, a reasonable error in estimating a Participant's Base Pay, or because of other limited facts and circumstances, the Annual Additions which would be credited to a Participant's accounts for a Limitation Year would nonetheless exceed the Maximum Annual Additions for such Participant for such year, the excess Annual Additions which, but for this Section 4.5, would have been allocated to such Participant's Employee Contribution Account shall, to the extent possible, first be reduced by returning to such Participant the additional Employee contributions which are considered in determining such Participant's Annual Additions. Next, excess Annual Additions in the form of Employer contributions and forfeitures which, but for this Section, would have been allocated to such Participant's Employer Contribution Account shall be allocated instead to a suspense account, and shall be used to reduce Employer contributions in the same manner as a forfeiture. Any remaining excess Annual Additions in the form of basic Employee contributions which, but for this Section, would have been allocated to such Participant's Employee Contribution Account shall be returned to such Participant. If a suspense account is in existence at any time during a Limitation Year pursuant to this Subsection (A), it will not participate in allocations of the net income (or net loss) of the Trust Fund.

Employee contribution elections of affected Participants pursuant to Section 4.2 may be revised prospectively by the Administrative Board on a temporary basis to the extent necessary to meet such limitations in the manner described in Section 4.8(C).

For purposes of determining whether such Participant's Maximum Annual Additions exceed the limitations herein provided, all defined contribution plans of the Employer are to be treated as one defined contribution plan. In addition, all defined contribution plans of Controlled Entities shall be aggregated for this purpose. For purposes of this paragraph only, a "Controlled Entity" shall be determined by application of a 50% control standard in lieu of an 80% control standard.

In the case of a Participant who also participated in a defined benefit plan of the Employer or a Controlled Entity (as defined in the immediately preceding paragraph), the Employer shall reduce the Annual Additions credited to the Accounts of such Participant under this Plan pursuant to the provisions of this Section 4.5(A) to the extent necessary to prevent the limitation set forth in section 415(e) of the Code from being exceeded. Notwithstanding the foregoing, the provisions of this Paragraph shall apply only if such defined benefit plan does not provide for a reduction of benefits thereunder to ensure that the limitation set forth in section 415(e) of the Code is not exceeded. Further, this Paragraph shall not apply for Limitation Years beginning after December 31, 1999.

- (B) For purposes of this Section 4.5, the following terms and phrases shall have these respective meanings.
 - (1) "Limitation Year" shall mean the calendar year.
 - - (a) the Participant's wages, salaries, fees for professional services, and other amounts received for personal services actually rendered in the course of employment with an Employer maintaining the Plan (including, but not limited to, commissions paid to sales personnel, compensation for services on the basis of a percentage of profits, commissions on insurance premiums, tips, and bonuses);
 - (b) in the case of a Participant who is an employee within the meaning of section 401(c)(1) of the Code and the regulations thereunder, the Participant's earned income (as described in section 401(c)(2) of the Code and the regulations thereunder);
 - (c) for purposes of Subsection 4.5(B)(2)(a) and (b), earned income from sources outside the United States (as defined in section 911(b) of the Code), whether or not excludable from gross income under section 911 of the Code or deductible under section 913 of the Code;

- (d) amounts described in sections 104(a)(3), 105(a), and 105(h) of the Code, but only to the extent that these amounts are includable in the gross income of the Employee;
- (e) amounts described in section 105(d) of the Code, whether or not these amounts are excludable from the gross income of the Employee under that section; and
- (f) amounts paid or reimbursed by the Employer for moving expenses incurred by an Employee, but only to the extent that these amounts are not deductible by the Employee under section 217 of the Code.
- (3) Participant's Remuneration for a Limitation Year shall not include:
- (a) contributions made by the Employer to a plan of deferred compensation to the extent that, before application of the limitations imposed under section 415 of the Code with respect to such plan, the contributions are not includable in the gross income of the Participant for the year in which contributed (except as to elective deferrals to a qualified plan described in section 401(k) of the Code or to a tax-sheltered annuity plan described in section 403(b) of the Code, and except as to elective contributions to a deferred compensation plan described in section 457 of the Code or to a cafeteria plan described in section 125 of the Code), Employer contributions made on behalf of a Participant to a simplified employee pension plan described in section 408(k) of the Code to the extent deductible by the Employee under section 219(b)(7) of the Code, and distributions from a plan of deferred compensation other than an unfunded non-qualified plan;
- (b) amounts realized from the exercise of a non-qualified stock option, or when restricted stock (or property) either becomes freely transferable or is no longer subject to a substantial risk of forfeiture;
- (c) amounts realized from the sale, exchange, or other disposition of stock acquired under a qualified or incentive stock option; or
- (d) other amounts which receive special tax benefits, such as premiums for group term life insurance to the extent not includable in the gross income of the Participant.
- (4) "Annual Additions" of a Participant, for any Limitation Year, shall mean the total (a) of the Employer contributions and forfeitures allocated to such Employee's accounts under the Plan for such year and (b) the amount of such Participant's Employee contributions to the Plan (excluding any rollover contributions).
- (5) "Maximum Annual Additions" of a Participant for any Limitation Year shall mean the lesser of \$30,000 or 25%, of such Participant's Remuneration during such year; provided, however, that the \$30,000 limitation shall be adjusted automatically to reflect any cost-of-living adjustments authorized by the Code or Treasury Regulations.

4.6 Rollover Contributions.

- (a) Qualified Rollover Contributions may be made to the Plan by any Employee of amounts received by such Employee from certain individual retirement accounts or annuities or from an employees' trust described in section 401(a) of the Code, which is exempt from tax under section 501(a) of the Code, but only if any such Rollover Contribution is made pursuant to and in accordance with applicable provisions of the Code and Treasury regulations promulgated thereunder. A Rollover Contribution of amounts that are "eligible rollover distributions" within the meaning of section 402(f)(2)(A) of the Code may be made to the Plan irrespective of whether such eligible rollover distribution was paid to the Employee or paid to the Plan as a "direct" Rollover Contribution. A direct Rollover Contribution to the Plan may be effectuated only by wire transfer directed to the Trustee or by issuance of a check made payable to the Trustee, which is negotiable only by the Trustee and which identifies the Employee for whose benefit the Rollover Contribution is being made. Any Employee desiring to effect a Rollover Contribution to the Plan must complete a rollover request in accordance with the procedures established from time to time by the Administrative Board. The Administrative Board may require as a condition to accepting any Rollover Contribution that such Employee furnish any evidence that the Administrative Board in its discretion deems satisfactory to establish that the proposed Rollover Contribution is in fact eligible for rollover to the Plan and is made pursuant to and in accordance with applicable provisions of the Code and Treasury regulations. All Rollover Contributions to the Plan must be made in cash. A Rollover Contribution shall be credited to the Rollover Account of the Employee for whose benefit such Rollover Contribution is being made as soon as administratively feasible.
- (b) An Employee who has made a Rollover Contribution in accordance with this Section, but who has not otherwise become a Participant in the Plan in accordance with Article III, shall become a Participant coincident with such Rollover Contribution; provided, however, that such Participant shall not have a right to defer Base Pay or have Employer Contributions made on his behalf until he has otherwise satisfied the requirements imposed by Article III.
- 4.7 Voting, Tendering, and Exchanging of Company Stock. Notwithstanding any provisions herein to the contrary, the following provisions shall govern with respect to the voting, tendering, exchanging and other rights concerning Company Stock:
 - (A) Notice. At the time of mailing to shareholders of the notice of any shareholders' meeting of the Company, or any notice of a tender or exchange offer for Company Stock or notice by the Company of any other action with respect to Company Stock, the Company shall use its reasonable best efforts to cause to be delivered to each Participant and former Participant (whose account has allocated to it any shares of Company Stock) such notices and informational statements as are furnished to the Company's shareholders in respect of the exercise of voting, tendering, exchanging, or other rights, together with forms by which the Participant or former Participant may instruct the Trustee, or revoke such instructions, with respect to the exercise of voting, tendering, exchanging, or other rights applicable to shares of Company Stock credited to such account.

- (B) Voting. Every Participant or former Participant (whose account has allocated to it any shares of Company Stock) shall have the right to direct the Trustee with respect to voting Company Stock allocated (or allocable) to such accounts, and the Trustee shall vote such allocated shares as directed. All of the shares of Company Stock for which no voting instructions are received shall be voted by the Trustee in a uniform manner as a single block in accordance with the instructions received with respect to a majority of such shares for which instructions are received.
- (C) Tendering and Exchanging. In the event of a tender or exchange offer, every Participant or former Participant (whose account has allocated to it any shares of Company Stock) shall have the right to direct the Trustee whether to accept or decline the offer with respect to Company Stock allocated (or allocable) to such accounts, and the Trustee shall take such actions as directed. All of the shares of Company Stock for which no instructions are received with respect to tender offer or exchange rights shall not be tendered or exchanged by the Trustee.
- 4.8 Restrictions on Employee Contributions.
- (A) In restriction of the Participants' Employee contribution elections provided in Section 4.2, the Employee contributions on behalf of any Participant for any calendar year shall not exceed \$7,000 (with such amount to be adjusted automatically to reflect any cost-of-living adjustments authorized by section 402(g)(5) of the Code), reduced by any "excess deferrals" from other plans allocated to the Plan by March 1 of the next following calendar year within the meaning of, and pursuant to the provisions of, section 402(g)(2) of the Code.
- (B) It is specifically provided that in each Plan Year the Plan shall comply with the alternative method of satisfying the 'actual deferral percentage' tests as set forth in section 401(k)(12) of the Code and applicable regulatory authority.
- 4.9 Restrictions on Employer Contributions. It is specifically provided that in each Plan Year the Plan shall comply with the alternative method of satisfying the 'actual contribution percentage' tests set forth in section 401(m) of the Code and applicable regulatory authority.

4.10 Excess Contributions.

- (A) Anything to the contrary herein notwithstanding, any Employee contributions to the Plan for a calendar year on behalf of a Participant in excess of the restrictions set forth in Section 4.8(A) shall be distributed to such Participant not later than April 15 of the next following calendar year. At the same time, any related Employer contributions shall be forfeited.
- (B) In coordinating excess contributions pursuant to this Section, such excess contributions shall be treated in the following order:
 - (1) first, excess Employee contributions described in Subsection (A) above not considered in determining related Employer contributions shall be distributed; and

- (2) next, excess Employee contributions described in Subsection (A) above considered in determining related Employer contributions shall be distributed, and related Employer contributions shall be forfeited.
- (C) Any distribution or forfeiture of excess contributions pursuant to Subsections (A) or (B) of this Section shall be adjusted for income or loss allocated thereto in a manner consistent with applicable Treasury Regulations, rulings and notices, and such distribution (or forfeiture, if applicable) will include such income or be reduced by such loss. Any distribution or forfeiture of excess contributions pursuant to this Section shall be made in cash.

ARTICLE V

VALUATION OF ACCOUNTS

All amounts contributed to the Trust shall be invested as soon as administratively feasible following their receipt by the Trustee, and the balance of each Account shall reflect the result of daily pricing of the assets in which such Account is invested from the time of receipt by the Trustee until the time of distribution.

ARTICLE VI

VESTING

- 6.1 Normal Retirement: Notwithstanding the provisions of Section 6.4, a Participant shall have a Nonforfeitable Interest in such Participant's entire Employer Contribution Account if the Participant is actively employed (A) on or after age 65, (B) on the day immediately preceding such Participant's death, or (C) on separation from service of the Employer upon incurring a Total Disability. No forfeiture shall thereafter arise under Section 4.3.
- 6.2 Plan Termination, Partial Plan Termination, or Complete Plan Discontinuance of Employer Contributions. Notwithstanding any other provision of this Plan, in the event of a termination or partial termination of the Plan, or a complete discontinuance of Employer contributions under the Plan, all affected Participants shall have a Nonforfeitable Interest in their Employer Contribution Accounts determined as of the date of such event. The value of these accounts and their Employee Contribution Accounts and Prior Employee Contribution Accounts as of such date shall be determined in accordance with the provisions of the Plan.
- 6.3 Vesting of Employee Contributions. A Participant shall accrue a Nonforfeitable Interest in such Participant's Employee Contribution Account and Prior Employee Contribution Account at all times.
- 6.4 Vesting of Employer Contributions. A Participant shall accrue a Nonforfeitable Interest in such Participant's Employer Contribution Account at the rate of 2% per month up to a maximum of 100%, for each month of such Participant's Period of Service, commencing after the Participant's first year of service. If a Participant's Nonforfeitable Interest under any prior vesting schedule of the Plan is greater than the Nonforfeitable Interest calculated under the above schedule, such Participant shall retain that Nonforfeitable Interest, and if the Nonforfeitable Interest is less than 100%, it will increase at the rate of 2% per month. Notwithstanding the above, a Participant will have a 100% Nonforfeitable Interest in such Participant's Employer Contribution Account upon retirement under the American General Retirement Plan or any successor plan.

Each Participant with a Period of Service of five years or more with the Employer on the effective date of any amendment to the preceding paragraph may elect, within a reasonable period after the adoption of the amendment, to have such Participant's Nonforfeitable Interest computed under the Plan without regard to such amendment. The period during which the election may be made shall commence with the date the amendment is adopted and shall end on the later of:

- (A) 60 days after the restatement is adopted:
- (B) 60 days after the amendment becomes effective; or
- (C) 60 days after the Participant is issued written notice of the amendment by the Employer or the Administrative Board.

Further notwithstanding the above, a Participant will have a 100% Nonforfeitable Interest in such Participant's Employer Contribution Account Subaccount at all times.

- 6.5 Vesting After a Period of Severance. No Period of Service after a Period of Severance equal to at least five consecutive One-Year Breaks in Service shall be taken into account in determining the nonforfeitable percentage in a Participant's Employer Contribution Account accrued up to any such Period of Severance.
- 6.6 Vesting Before a Period of Severance of Five Consecutive One-Year Breaks in Service. A Participant who separates from service of the Employer and received a distribution of the Nonforfeitable Interest (of less than 100%), in such Participant's Employer Contribution Account in accordance with Section 7.4, shall forfeit amounts that are forfeitable in accordance with Section 4.3.
- 6.7 Election to Make a Withdrawal. If a Participant elects to make a withdrawal pursuant to Section 7.6(B), (C), (D) or (E) while remaining employed by the Company, such distribution will be subject to the vesting provisions of Section 6.4. Upon incurring a Period of Severance equal to at least five consecutive One-Year Breaks in Service, that part of such individual's Employer Contribution Account which is not vested shall be forfeited, and such forfeiture shall be available to reduce future Employer Contributions.
- 6.8 Transfer Between Plans. If a Participant ceases to be eligible as an Employee under this Plan but immediately becomes an Employee for purposes of another thrift plan maintained by the Company or its subsidiaries ("Other Plan"), such change in employment status will not be deemed a termination of employment for purposes of this Plan.
- 6.9 Independent Life Transferred Employees. A Transferred Employee is an Employee who was an employee of Independent Life and Accident Insurance Company on the day preceding his employment by the Employer. For purposes of computing a Participant's Nonforfeitable Interest in this Plan, a Transferred Employee's Period of Service will include full and partial years of credited service under the Independent Life INVEST Plan plus, if applicable, service with the Company from date of transfer from Independent Life and Accident Insurance Company to April 1, 1997.

A Participant's Employee Contribution Account, Prior Employee Contribution Account and Employer Contribution Account will be frozen, and no future Employee or Employer contributions will be made. However, the Participant will continue to accrue vesting in this Plan based on years of service credited in the Other Plan at the rate which would have been applicable had such individual continued participation in this Plan.

If an Employee has ceased to be a Participant under an Other Plan and immediately becomes a Participant under this Plan, such Employee will not be required to complete the one-year Period of Service or one year of "participation service" in Section 3.1 before participating in this Plan; and the Employee's Period of Service and Hours of Service under the Other Plan will be added to the Period of Service and Hours of Service under this Plan.

ARTICLE VII

BENEFITS

- 7.1 Normal Retirement. Upon separation from service on or after satisfying eligibility requirements for early, normal, or late retirement under the Company's retirement plan, other than by reason of death, a Participant shall be entitled to a benefit based on the combined balance of the Participant's Employee Contribution Account, Prior Employee Contribution Account and Employer Contribution Account distributed in a manner provided in Article IX.
- 7.2 Disability. In the event that a Participant separates from service of the Employer upon incurring a Total Disability before the Normal Retirement Date, such Participant shall be entitled to a benefit based on the combined balance of such individual's Employee Contribution Account, Prior Employee Contribution Account, and Employer Contribution Account distributed in the manner provided in Article IX.
- 7.3 Death. In the event of the death of a Participant prior to the commencement of a benefit described in Sections 7.1, 7.2, or 7.4, the Beneficiary shall be paid the combined balance in the Participant's Employee Contribution Account, Prior Employee Contribution Account and Employer Contribution Account distributed in the manner provided in Article IX.
- 7.4 Termination of Service. In the event a Participant separates from service of the Employer prior to the Normal Retirement Date for any reason other than death or Total Disability and Section 7.1 does not apply, the Nonforfeitable Interest in such Participant's Employer Contribution Account determined pursuant to Article VI and the balance of such Participant's Employee Contribution Account and Prior Employee Contribution Account shall be distributable to the Participant in accordance with Article IX.

7.5 Valuation Date to be Used.

- (A) If a Participant or Beneficiary becomes entitled to a benefit pursuant to Section 7.1, 7.2, 7.3, or 7.4, and elects to receive a distribution of such benefit as a result thereof, the value of the account balances to be distributed shall be determined as of the date of the event giving rise to the distribution. For purposes of Section 7.1, the event occasioning the benefit shall be the Participant's separation from service on or after satisfying the eligibility requirements for early, normal, or late retirement under the Company's retirement plan, other than by reason of death. For purposes of Section 7.2, the event occasioning the benefit shall be the Participant's separation from service of the Employer on account of Total Disability. For purposes of Section 7.3, the event occasioning the benefit shall be the death of the Participant prior to the commencement of a benefit described in Section 7.1, 7.2, or 7.4. For purposes of Section 7.4, the event occasioning the benefit shall be the date the Participant separated from service.
- (B) If pursuant to Section 8.4, a Participant elects to receive a benefit after an event set forth in Section 7.1, 7.2, 7.3, or 7.4, the value of the account balances to be distributed shall be determined as of the Withdrawal Date on which the Plan Administrator receives the Participant's withdrawal request in accordance with the procedures established from time to time by the Administrative Board.

7.6 Withdrawals.

- (A) A Participant may withdraw from his Prior Employee Contribution Account any or all amounts held in such Account which have been so held for six months or more.
- (B) A Participant who has withdrawn all amounts in his Prior Employee Contribution Account but who has not made or had made on his behalf to the Plan basic Employee contributions for at least 60 cumulative months may withdraw from his Employer Contribution Account other than his Employer Contribution Account Subaccount any or all amounts held in such account which have been so held for 24 months or more, but not in excess of his Nonforfeitable Interest in the then value of such account. Any amounts in an Employer Contribution Account which is a Rollover Account must be withdrawn prior to amounts in any other Employer Contribution Account.
- (C) A Participant who has withdrawn all amounts in his Prior Employee Contribution Account and who has made or had made on his behalf to the Plan basic Employee contributions for at least 60 cumulative months may withdraw from his Employer Contribution Account other than his Employer Contribution Account Subaccount an amount not exceeding his Nonforfeitable Interest in the then value of such Account. Any amounts in an Employer Contribution Account which is a Rollover Account must be withdrawn prior to amounts in any other Employer Contribution Account.
- (D) A Participant who has attained age 59 1/2 and who has made all available withdrawals pursuant to Subsections (A), (B), and (C) above may withdraw from his Employer Contribution Account Subaccount an amount not exceeding the then value of such Account.
- (E) A Participant who has attained age 59 1/2 and who has made all available withdrawals pursuant to Subsections (A), (B), (C) and (D) above may withdraw from his Employee Contribution Account an amount not exceeding the then value of such Account.
- (F) A Participant who has a financial hardship, as determined by the withdrawal committee ("Withdrawal Committee") appointed by the Administrative Board, and who has made all available withdrawals pursuant to the Subsections above and pursuant to the provisions of any other plans of the Employer or any Controlled Entities of which he is a member and who has obtained all available loans pursuant to the provisions of any plans of the Employer and any Controlled Entities of which he is a member may withdraw from his Employee Contribution Account amounts not to exceed the lesser of (1) his Nonforfeitable Interest in the then value of such account or (2) the amount determined by the Withdrawal Committee as being available for withdrawal pursuant to this Subsection. No withdrawals on account of financial hardship will be permitted for amounts less than \$500. For purposes of this Subsection, financial hardship means the immediate and heavy financial needs of the Participant. A withdrawal based upon financial hardship pursuant to this Subsection shall not exceed the amount required to meet the immediate financial need created by the hardship and not reasonably available from other resources of the Participant. The determination of the existence of a Participant's financial hardship and the amount required to be distributed to meet the need created by the hardship shall

be made by the Withdrawal Committee based upon information requested from the Participant by the Withdrawal Committee and considering all relevant facts and circumstances. The amount of an immediate and heavy financial need may include any amounts necessary to pay any federal, state, or local income taxes or penalties reasonably anticipated to result from the distribution. In addition to any immediate and heavy financial need as may be determined by the Withdrawal Committee, a withdrawal shall be deemed to be made on account of an immediate and heavy financial need of a Participant if the withdrawal is on account of:

- (1) medical expenses described in section 213(d) of the Code incurred by the Participant, the Participant's spouse or any dependents of the Participant (as defined in section 152 of the Code) or necessary for those persons to obtain medical care described in section 213(d) of the Code, and not reimbursed by insurance;
- (2) costs directly related to the purchase (excluding mortgage payments) of a principal residence of the Participant;
- (3) payment of tuition and related educational fees for the next 12 months of post-secondary education of the Participant, or the Participant's spouse, children or dependents (as defined in section 152 of the Code); or
- (4) the need to prevent the eviction of the Participant from his principal residence or foreclosure on the mortgage of the Participant's principal residence.

The decision of the Withdrawal Committee shall be final and binding, provided that all Participants similarly situated shall be treated in a uniform and nondiscriminatory manner. The above notwithstanding, withdrawals under this Subsection from a Participant's Employee Contribution Account shall be limited to the sum of the Participant's Employee contributions to the Plan, less any previous withdrawals of such amounts. A Participant who makes a withdrawal under this Subsection may not again make elective contributions to the Plan or any other qualified or nonqualified plan of the Employer or any Controlled Entity for a period of 12 months following such withdrawal. Further, such Participant may not make elective contributions under the Plan or any other plan maintained by the Employer or any Controlled Entity for such Participant's taxable year immediately following the taxable year of the withdrawal in excess of the applicable limit set forth in Section 4.8(A) for such next taxable year less the amount of such Participant's elective contributions for the taxable year of the withdrawal.

(G) All withdrawals pursuant to this Section shall be made as of the Withdrawal Date on which the Plan Administrator receives a withdrawal request in accordance with the procedures established from time to time by the Administrative Board. All withdrawals shall be made pro rata from each fund in which such account is invested.

7.7 Loans.

(A) Upon application by (1) any Participant who is an Employee or (2) any Participant no longer employed by the Employer, a beneficiary of a deceased Participant or an Alternate Payee under a Qualified Domestic Relations Order, as defined in Section 13.11, who retains a balance in his Accounts under the Plan and who is a party-in-interest, as that term is

defined in section 3(14) of ERISA, as to the Plan (an individual who is eligible to apply for a loan under this Section being hereinafter referred to as a "Participant" for purposes of this Section), the loan committee ("Loan Committee") appointed by the Administrative Board may in its discretion direct the Trustee to make a loan or loans to such Participant. Such loans shall be made pursuant to the provisions of the Loan Committee's written loan procedure, which procedure is hereby incorporated by reference as a part of the Plan.

- (B) A loan to a Participant may not exceed 50% of the then value of such Participant's Nonforfeitable Interest in his Accounts.
- (C) Paragraph (B) above to the contrary notwithstanding, the amount of a loan made to a Participant under this Section shall not exceed an amount equal to the difference between:
 - (1) The lesser of \$50,000 (reduced by the excess, if any, of (A) the highest outstanding balance of loans from the Plan during the one-year period ending on the day before the date on which the loan is made over (B) the outstanding balance of loans from the Plan on the date on which the loan is made) or one-half of the present value of the Participant's total nonforfeitable accrued benefit under all qualified plans of the Employer or Controlled Entity; minus
 - (2) The total outstanding loan balance of the Participant under all other loans from all qualified plans of the Employer or a Controlled Entity.

ARTICLE VIII

COMMENCEMENT OF BENEFITS

8.1 Benefits After Normal Retirement Date. Payment shall be made within 60 days after the Participant separates from service of the Employer (including separation by reason of death or Total Disability) on or after the Participant's Normal Retirement Date.

Although full vesting is granted at age 65, distribution of the account of a Participant who continues active work past the Normal Retirement Date shall be postponed until such Participant's actual termination of employment or prior Plan withdrawal. Any such Participant may continue active participation in the Plan until the Participant's actual retirement date.

Notwithstanding the foregoing, subject to the provisions of Section 8.4, a Participant who terminates employment (for a reason other than death) may, if his total accounts (both vested and nonvested) exceed \$5,000, elect to defer payment of his benefit by making an appropriate election to defer with the Plan Administrator.

- 8.2 Certain Benefits Before Normal Retirement Date. Except as provided in Section 8.1, upon the death of the Participant, payment shall be made not later than 60 days after receipt by the Plan Administrator of proof of death. Except as provided in Section 8.1, upon separation from service of the Employer upon incurring a Total Disability, payment shall be made no later than 60 days after the determination that Total Disability exists.
- 8.3 Termination of Service Before Normal Retirement Date. Upon separation from service before Normal Retirement Date other than by reason of death or Total Disability, the benefit to which a Participant is entitled under Section 7.4 shall be paid no later than 60 days after the Participant's separation from service, and is to be paid in the form prescribed in Article IX.
- 8.4 Commencement of Benefits. Notwithstanding anything in Section 8.1, 8.2, or 8.3, other than an election to defer made by a Participant pursuant to Section 8.1, payments of benefits shall be made or shall commence no later than 60 days after the close of the Plan Year in which the Participant attains or would have attained age 65, or, if later, separates from service.

Notwithstanding anything in this Section or in Sections 8.1, 8.2, or 8.3 to the contrary, the entire Nonforfeitable Interest of each Participant shall be distributed to such Participant no later than April 1 of the calendar year following the later of (i) the calendar year in which the Participant attains age 70 1/2 or (ii) the calendar year in which the Participant terminates his employment with the Employer (provided, however, that clause (ii) of this sentence shall not apply prior to January 1, 1999, and shall not apply in the case of a Participant who is a 'five-percent owner' (as defined in section 416 of the Code) with respect to the Plan Year ending in the calendar year in which such Participant attains age 70 1/2. Notwithstanding the foregoing, in accordance with procedures adopted by the Administrative Board, (1) a Participant (who is not a 'five-percent owner' with respect to the calendar year in which the Participant attains age 70 1/2 who attains age 70 1/2 in calendar year 1996, 1997, or 1998, may elect to defer his payment as if such age were attained after calendar year 1998, and (2) a Participant who attains age 70 1/2 prior

to calendar year 1997 and who has not terminated employment with the Employer may elect to stop distributions which have commenced as of a prior payment date.

If a Participant dies before the entire Nonforfeitable Interest has been distributed to such Participant, such interest shall be distributed, within five years of the Participant's death, to the designated Beneficiary.

No benefit shall be paid to a Participant who is under the age of 65, without such person's consent, if the total value (both vested and unvested) of such Participant's accounts exceeds \$5,000.

A Participant's commencement of benefits shall be in compliance with the provisions of section 401(a)(9) of the Code and applicable Treasury Regulations thereunder.

ARTICLE IX

DISTRIBUTION OF BENEFITS

- 9.1 Mode of Benefit Payments. Upon death, retirement, Total Disability, termination of employment for any other reason or withdrawal, amounts in the Participant's Employer Contribution Account (subject to the vesting provisions of Article VI), and amounts in the Participant's Employee Contribution Account, Prior Employee Contribution Account, and Rollover Account shall be distributed as follows:
 - (A) from the Stock Fund, either in Company Stock or in cash at the Participant's election;
 - (B) from the Investment Funds, in cash.

Notwithstanding the above and notwithstanding the following sentence, if a distribution would include fractional shares of Company Stock, the value of such fractional shares shall be paid in cash instead. Except in situations subject to Section 9.3, if an election is not received by the Plan Administrator prior to commencement of payment of benefits pursuant to Section 8.4, the distribution of the vested interest in the Participant's Employer Contribution Account and of any amounts in the Participant's Employee Contribution Account, Prior Employee Contribution Account, and Rollover Account which are held in the Stock Fund shall be made in Company Stock, and the distribution of all other amounts shall be in cash.

All assets held in the Participant's Accounts shall be valued as of the date prescribed in Section 7.5. No interest or other earnings shall be paid after the valuation date prescribed in Section 7.5. Distributions of benefits shall be made as soon as administratively feasible and in accordance with Article VIII.

- 9.2 Stock Certificates. Where certificates for shares of stock are distributed, they will be issued in the name of the Participant only. However, upon the death of a Participant, certificates will be issued in the name of the Beneficiary.
- 9.3 Automatic Distributions. A Participant whose total accounts (both vested and nonvested) are valued at \$5,000 or less must make an election authorized by Section 9.1 within 30 days following termination of employment, retirement, or Total Disability. A Beneficiary must make an election within 30 days following the Participant's death. If an election is not received by the Plan Administrator within the applicable time periods stated above, the distribution shall be made in cash.

Nothing in this Article shall be construed to permit or require distribution of assets in which the Participant does not have a Nonforfeitable Interest, as determined under Article VI. Likewise, this Article shall not be construed to permit or require a distribution without a Participant's consent if such distribution is prohibited by Section 8.4.

9.4 Unclaimed Benefits. In the case of a benefit payable on behalf of a Participant, if the Plan Administrator is unable to locate the Participant or Beneficiary to whom such benefit is

payable, upon the Plan Administrator's determination thereof, such benefit shall be forfeited and shall be available to reduce future Employer contributions. Notwithstanding the foregoing, if subsequent to any such forfeiture the Participant or Beneficiary to whom such benefit is payable makes a valid claim for such benefit, such forfeited benefit shall be restored to the Plan in the manner provided in Section 4.3. No interest or other earnings shall be paid or accrued in regard to the restored benefit.

9.5 Direct Rollovers.

- (A) This Section 9.5 applies to distributions made on or after January 1, 1993. Notwithstanding any provision of the Plan to the contrary that would otherwise limit a distributee's election under this Section, a distributee may elect, at the time and in the manner prescribed by the Plan Administrator, to have any portion of an "eligible rollover distribution" paid directly to an "eligible retirement plan" specified by the "distributee" in a "direct rollover."
- (B) For purposes of this Section 9.5, the following terms and phrases shall have these respective meanings:
 - (1) "Eligible rollover distribution" is any distribution of all or any portion of the balance to the credit of the distributee, except that an eligible rollover distribution does not include: any distribution that is one of a series of substantially equal periodic payments (not less frequently than annually) made for the life (or life expectancy) of the distributee or the joint lives (or joint life expectancies) of the distributee and the distributee's designated beneficiary, or for a specified period of ten years or more; any distribution to the extent such distribution is required under section 401(a)(9) of the Code; and the portion of any distribution that is not includible in gross income (determined without regard to the exclusion for net unrealized appreciation with respect to employer securities). Further, a distribution pursuant to newly relettered Section 7.6(F) shall not constitute an Eligible Rollover Distribution to the extent provided in section 402(c)(4) of the Code and the interpretive authority thereunder.
 - (2) "Eligible retirement plan" is an individual retirement account described in section 408(a) of the Code, an individual retirement annuity described in section 408(b) of the Code, an annuity plan described in section 403(a) of the Code, or a qualified trust described in section 401(a) of the Code, that accepts the distributee's eligible rollover distribution. However, in the case of an eligible rollover distribution to the surviving spouse, an eligible retirement plan is an individual retirement account or individual retirement annuity.
 - (3) "Distributee" includes an employee or former employee. In addition, the employee's or former employee's surviving spouse and the employee's or former employee's spouse or former spouse who is the alternate payee under a qualified domestic relations order, as defined in section 414(p) of the Code, are distributees with regard to the interest of the spouse or former spouse.
 - (4)"Direct rollover" is a payment by the plan to the eligible retirement plan specified by the distributee.

ARTICLE X

PLAN AMENDMENT AND TERMINATION

10.1 Amendment of the Plan. The Company shall have the right to amend or modify this Plan and (with the consent of the Trustee) the Trust agreement at any time, and from time to time, to any extent that it may deem advisable. Any such amendment or modification shall be set out in an instrument in writing, duly authorized by the Board of Directors and executed by the Company. No such amendment or modification shall, however, increase the duties or responsibilities of the Trustee without its consent thereto in writing, or have the effect of transferring to or vesting in any Employer any interest or ownership in any properties of the Trust Fund, or of permitting the same to be used for or diverted to purposes other than for the exclusive benefit of the Participants and their Beneficiaries. No such amendment shall decrease the accounts of any Participant or shall decrease any Participant's Nonforfeitable Interest in such account. Notwithstanding anything herein to the contrary, the Plan or the Trust agreement may be amended in such manner as may be required at any time to make it conform to the requirements of the Code or any United States statutes with respect to employee trusts, or any amendment thereto, or any regulations or rulings issued pursuant thereto, and no such Plan or Trust amendment shall be considered prejudicial to any then existing rights of any Participant or Beneficiary under the Plan.

For purposes of this Section and subject to regulations to be issued by the Secretary, a Plan amendment made after July 30, 1984, which has the effect of reducing an early retirement benefit or an optional form of benefit, with respect to benefits attributable to service before the amendment (even though unvested), shall be treated as reducing a Participant's account.

- 10.2 Communication of Amendments. All amendments, including one to terminate the Plan, shall be adopted in writing by the Company's Board of Directors. Any material modification of the Plan by amendment or termination shall be communicated to all interested parties and the Secretaries of Labor and the Treasury in the time and manner prescribed by law.
- 10.3 Termination or Discontinuance of Employer Contributions. Upon plan termination or discontinuance of Employer contributions under the Plan all account balances shall be valued in accordance with Section 6.2. The Trustee shall then, as soon as administratively feasible, pay each Participant and Beneficiary the entire interest in the Trust attributable to such Participant or Beneficiary in a lump sum, and shall pay any remaining amount to the Employer. In case of any Participant whose residence is unknown, the Plan Administrator shall notify such Participant at the last known address by certified mail with return receipt requested advising such Participant of a pending distribution.
- 10.4 Acceptance or Rejection of Amendment Employers. The Company shall promptly deliver to each other Employer any amendment to this Plan or the Trust agreement. Upon delivery to an Employer of an executed copy of an amendment properly authorized and adopted by the Company, the Plan as to such Employer shall be thereupon amended in accordance therewith.

10.5 Termination of the Plan by an Employer. An Employer may at any time, by adoption of a resolution, terminate the Plan with respect to the Employees of said Employer, and may direct and require the Trustee to liquidate the share of the Trust Fund allocable to its Employees or their Beneficiaries. If the Plan is terminated by fewer than all Employers, the Plan shall continue in effect for the Employees of the remaining Employers. In the event that an Employer shall cease to exist, the Plan shall be terminated with respect to the Employees of such Employer, unless a successor organization adopts and continues the Plan.

10.6 Notice of Terminating Employer. A terminating or withdrawing Employer shall give 90 days notice in writing of its intention to the Administrative Board, the Company, and the Trustee, unless a shorter notice is agreed to by the Company. If an Employer withdraws from this Plan and provides for a successor plan for its Employees, the Trust Fund assets of this Trust held on behalf of such Employer may be determined and transferred to a successor trust upon approval by the appropriate District Director of Internal Revenue.

ARTICLE XI

ADMINISTRATION

- 11.1 Named Fiduciaries. The named fiduciaries shall be the Plan Administrator and the Trustee.
 - 11.2 Appointment of the Trustee.
- (A) The Company shall designate the Trustee in a written statement filed with the Company's Board of Directors. The appointment of the Trustee shall become effective at such time as the Trustee and the Company execute a valid written trust which definitely and affirmatively precludes prohibited diversion.
- (B) The resignation of a Trustee shall be made in writing, submitted to the Company, and recorded in the minutes of the Board of Directors. The discharge of any person described in the preceding sentence shall be effectuated in writing by the Company and delivered to such person with the details thereof recorded in the minutes of the Company's Board of Directors. Appointment of a successor Trustee shall be carried out in the manner prescribed in Subsection (A).
- 11.3 Trustee's Powers and Duties. The powers and duties of the Trustee shall be to manage and control the funds of the Trust in accordance with the terms of the Trust agreement forming a part hereof.
- 11.4 Administrative Expenses. Except for fees relating to Participant loans and commissions on and other costs relating to acquisitions or disposition of securities, including any reasonable fees in connection with the Investment Funds, the Employer may pay the administrative expenses of the Plan and Trust, including the reasonable compensation of the Trustee and reimbursement of its reasonable expenses. To the extent not paid by the Employer, the administrative expenses of the Plan and Trust will be paid by the Trustee of the Trust
- 11.5 Plan Administrator's Powers and Duties. The Plan Administrator shall have the following powers and duties:
 - (A) to construe and interpret the provisions of the Plan;
 - (B) to decide all questions of eligibility for Plan participation and for the payment of benefits;
 - (C) to provide appropriate parties, including government agencies, with such returns, reports, schedules, descriptions, and individual statements as required by law within the times prescribed by law, to furnish to the Employer, upon request, copies of any or all such materials, and further, to make copies of such instruments, reports, and descriptions as are required by law available for examination by Participants (and such of their Beneficiaries who are or may be entitled to benefits under the Plan) in such places and in such manner as required by law;

- (D) to obtain from the Employer, the Employees, and the Trustee such information as shall be necessary for the proper administration of the Plan;
- (E) to determine the amount, manner, and time of payment of benefits hereunder;
- (F) subject to the approval of the Company only as to any additional expense, to appoint and retain such agents, counsel, and accountants for the purpose of properly administering the Plan and, when required to do so by law, to engage an independent Certified Public Accountant to annually prepare the audited financial statement of the Plan's operations;
- (H) to notify the Trustee in writing of the termination of the Plan or the complete discontinuance of Employer contributions,
- (I) to direct the Trustee to distribute assets of the Trust to each Participant and Beneficiary in accordance with Article IX of the Plan;
- (J) to furnish each recipient of a "qualifying rollover distribution," as defined by section 402 of the Code, with a written explanation of the Code provisions under which such distribution will not be subject to tax if transferred to an eligible retirement plan within 60 days after which the recipient received the distribution and, if applicable, the provisions concerning "Capital Gains Treatment for Portion of Lump Sum Distribution" and the provisions concerning "Tax on Lump Sum Distributions" provided by section 402 of the Code;
- (K) to establish, at a meeting duly called for such purpose, a funding policy and method consistent with the objectives of the Plan and the requirements of Title I of ERISA. The Administrative Board shall meet at least annually to review such funding policy and method; and

All actions of the Plan Administrator taken pursuant to this Section shall be presented in a meeting of the Administrative Board and recorded in the minutes thereof. An annual report shall be presented to the Board of Directors.

11.6 Named Fiduciary's Powers and Duties. It shall be the responsibility of the Plan Administrator to provide a notice in writing to any Participant or Beneficiary whose claim for benefits under this Plan has been denied by the Plan Administrator, setting forth the specific reasons for such denial, and to afford such Participant or Beneficiary a reasonable opportunity for a full and fair review of that decision.

11.7 Allocation of Functions. Where more than one person serves as Plan Administrator, such persons may agree in writing to allocate among themselves the various powers and duties prescribed in Section 11.5, provided all such persons sign such agreement. A copy of any such agreement shall be promptly relayed to the Company.

ARTICLE XII

ADOPTION AND WITHDRAWAL

- 12.1 Procedure for Adoption. Any Controlled Entity may, with the approval of the Board of Directors, adopt and become an Employer under this Plan pursuant to appropriate written resolutions of the board of directors of such adopting Controlled Entity, by executing and delivering to the Company and the Trustee an adoptive instrument specifying the classification of its Employees who are to be eligible to participate in the Plan, and by agreeing to be bound as an Employer by all the terms of the Plan with respect to its eligible Employees. The adoptive instrument may contain such changes and variations in the terms of the Plan as may be acceptable to the Company. However, the sole, exclusive right of any other amendment of whatever kind or extent to the Plan or Trust are reserved by the Company. The adoption agreement shall become, as to such adopting organization and its Employees, a part of this Plan, as then or thereafter amended, and the related Trust. It shall not be necessary for the adopting organization to sign or execute the original or the amended Plan and Trust documents. The effective date of the Plan for any such adopting organization shall be that stated in the adoption agreement, and from and after such effective date such adopting organization shall assume all the rights, obligations, and liabilities of an Employer hereunder and under the Trust. The Company's administrative powers and control of the Plan, as provided in this instrument and the Trust, including the sole right of amendment, and of appointment and removal of the Administrative Board and the Trustee and their successors, shall not be diminished by reason of the participation of any such adopting organization in the Plan and Trust.
- 12.2 Effect of Adoption. The following special provisions shall apply to all $\ensuremath{\mathsf{Employees}}\xspace.$
 - (A) An Employee shall be considered in service while regularly employed simultaneously or successively by one or more Employers.
 - (B) The transfer of an Employee from one Employer to another Employer shall not be deemed a termination of service.
- 12.3 Withdrawal. Any participating Employer by action of its board of directors or other governing authority and notice to the Company and the Trustee, may withdraw from the Plan and Trust at any time, without affecting other Employers not withdrawing, by complying with the provisions of the Plan and Trust. A withdrawing Employer may arrange for the continuation, by itself or its successor, of this Plan and Trust in separate forms for its own Employees, with such amendments, if any, as it may deem proper, and may arrange for continuation of the Plan and Trust by merger with an existing plan and trust and transfer of Trust assets. The Company may, in its absolute discretion, terminate an adopting Employer's participation at any time when in its judgment such adopting Employer fails or refuses to discharge its obligations under the Plan. If an Employer is no longer a Controlled Entity, the Company may at any time terminate such Employer's participation under this Plan.

ARTICLE XIII

MISCELLANEOUS

- 13.1 Merger of this Plan with Another Plan. In the event of any merger or consolidation of the Plan with, or transfer, in whole or in part, of the assets and liabilities of the Trust Fund to another trust fund held under any other plan of deferred compensation maintained or to be established for the benefit of all or some of the Participants of this Plan, the assets of the Trust Fund applicable to such Participants shall be transferred to the other trust fund only if:
 - (A) each Participant would (if either this Plan or the other plan then terminated) receive a benefit immediately after the merger, consolidation, or transfer which is equal to or greater than the benefit such Participant would have been entitled to receive immediately before the merger, consolidation, or transfer (if this Plan had then terminated);
 - (B) resolutions of the board of directors of the Employer under this Plan, or of any new or successor employer' of the affected Participants, shall authorize such transfer of assets; and, in the case of the new or successor employer of the affected Participants, its resolutions shall include an assumption of liabilities with respect to such Participants' inclusion in the new employer's plan; and
 - (C) such other plan and trust are qualified under sections 401(a) and 501(a) of the Code.
- 13.2 Assignment and Alienation of Benefits. Except as otherwise provided in Section 13.11, except as to certain judgments and settlements described in section 401(a)(13) of the Code, and except as otherwise provided under other applicable law, no right or interest of any kind in any benefit under this Plan shall be transferable or assignable by any Participant or by any beneficiary or be subject to anticipation, adjustment, alienation, encumbrance, garnishment, attachment, execution, or levy of any kind.
- 13.3 Communication to Employees. The Plan Administrator shall furnish to each Participant and each Beneficiary receiving benefits under the Plan a copy of a summary plan description and a summary of any material modifications thereof at the time and in the manner prescribed by law.
- 13.4 Number and Gender. The masculine pronoun shall include the feminine pronoun, and the singular number shall include the plural number, unless the context of the Plan requires otherwise.
- 13.5 Construction. The terms of the Plan shall be construed under the laws of the State of Texas except to the extent such laws are preempted by federal law.
- 13.6 Not a Contract of Employment. The adoption of this Plan by an Employer shall not constitute a contract of employment between the Employer and any Employee.

- 13.7 Indemnity. The Company shall indemnify all those to whom it has delegated fiduciary duties against any and all claims, loss, damages, expense, and liability arising from their responsibilities in connection with the Plan, unless the same is determined to be due to gross negligence or willful misconduct. Plan benefits shall be provided only from the Trust Fund, and neither the Employer nor the Trustee guarantees or assumes any liability that the Trust Fund will at any time be sufficient therefor.
- 13.8 Change of Beneficiary. A Participant may change the Beneficiary designation under the Plan at any time by submitting a Change of Beneficiary request in accordance with the procedures established from time to time by the Administrative Board.
- 13.9 Applications to the Administrative Board. Correspondence shall be sent to such address as the Administrative Board shall designate from time to time or to the following address:

Administrative Board for the American General Employees'
Thrift and Incentive Plan
c/o Plan Administrator
2929 Allen Parkway
Houston, Texas 77019

- 13.10 Top-Heavy Rules.
- (A) For purposes of this Section 13.10, the following language defines a Top-Heavy Plan:
 - (1) In general:
 - (a) Plans Not Required to be Aggregated. Except as provided in Subsection 13.10(A)(1)(b), the term "Top-Heavy Plan" means, with respect to any plan year, any defined contribution plan if, as of the Determination Date; the aggregate of the accounts of Key Employees under the plan exceeds 60% of the aggregate of the accounts of all Employees under such plan.
 - (b) Aggregated Plans. Each plan of an Employer required to be included in an Aggregation Group shall be treated as a Top-Heavy Plan if such group is a Top-Heavy Group.
 - (2) Aggregation. For purposes of this Section 13.10:
 - (a) Aggregation Group:
 - (i) Required Aggregation. The term "Aggregation Group" means:
 - 1) each plan of the Employer in which a Key Employee is a Participant; and

- 2) each other plan of the Employer which enables any plan described in the preceding Subsection (I) to meet the requirements of section 401(a)(4) or 410 of the Code.
- (ii) Permissive Aggregation. The Employer may treat any plan not required to be included in an Aggregation Group under Subsection 13.10(A)(2)(a)(i) as being part of such group, if such group would continue to meet the requirements of sections 401(a)(4) and 410 of the Code with such plan being taken into account.
- (b) Top-Heavy Group. The term "Top-Heavy Group" means any Aggregation Group if:
 - (i) the sum (as of the Determination Date) of:
 - 1) the present value of the cumulative accrued benefits for Key Employees under all defined benefit plans included in such group; and
 - 2) the aggregate of the accounts of Key Employees under all defined contribution plans included in such group;
 - (ii) exceeds 60% of a similar sum determined for all $\ensuremath{\mathsf{Employees}}$.

For purposes of making the foregoing determination, an Employee's account balance as of a Determination Date shall be valued as of the most recent date within the 12-month period prior to such Determination Date as of which the Trust Fund was valued, and the net income (or loss) thereof allocated to the Participants' accounts.

- (3) Distributions During Last Five Years Taken into Account. For purposes of determining (a) the present value of the cumulative accrued benefit for any Employee, or (b) the amount of the account of any Employee, such present value or amount shall be increased by the aggregate distributions made with respect to such Employee under the plan during the five-year period ending on the Determination Date. The preceding sentence shall also apply to distributions under a terminated plan which, if it had not been terminated, would have been required to be included in an Aggregation Group.
 - (4) Other Special Rules. For purposes of this Section 13.10:

- (a) Rollover Contributions to a Plan Not Taken into Account. Except to the extent provided in Treasury Regulations, any rollover contribution (or similar transfer) initiated by the Employee and made after December 31, 1983 to a plan shall not be taken into account with respect to the transferee plan for purposes of determining whether such plan is a Top-Heavy Plan (or whether any Aggregation Group which includes such plan is a Top-Heavy Group).
- (b) Benefits Not Taken into Account if Employee Ceases to be Key Employee. If any individual is a non-Key Employee with respect to any plan for any plan year, but such individual was a Key Employee with respect to such plan for any prior plan year, any accrued benefit for such Employee (and the account of such Employee) shall not be taken into account.
- - (i) the last day of the preceding plan year; or
 - (ii) in the case of the first plan year of any plan, the last day of such plan year.
- (d) Years. To the extent provided in Treasury Regulations, this Section shall be applied on the basis of any year specified in such regulations in lieu of plan years.
- (e) Benefits Not Taken into Account if Employee Not Employed for Last Five Years. If any individual has not received any compensation from any Employer maintaining the Plan (other than benefits under the Plan) at any time during the five-year period ending on the Determination Date, any accrued benefit for such individual (and the account of such individual) shall not be taken into account.
- (f) For purposes of determining whether the Plan is Top Heavy, an individual is a "Key Employee" if such person is an individual described in section 416(i)(1) of the Code and the Treasury Regulations promulgated thereunder and, for purposes of applying such provisions, an individual's "compensation" (as such term is used in such section and such regulations) shall be deemed to be equal to such individual's Remuneration, as defined in Subsection 4.5(B)(2).

Should the Plan become Top Heavy, the following provisions will apply:

(B) A Participant shall have a Nonforfeitable Interest in such Participant's Employer Contribution Account in accordance with Section 6.4 or the following table, whichever is greater:

Years of Service	Nonforfeitable Percentage
2	20
3	40
4	60
5	80
6	100

- (C) The Employer shall contribute to the Plan for such Plan Year on behalf of each Participant who is not a Key Employee and who has not terminated employment as of the last day of such Plan Year an amount equal to:
 - (1) the lesser of (a) 3%, of such Participant's Remuneration as described in Subsection 4.5(B)(2) for such Plan Year, or (b) a percent of such Participant's Remuneration as described in Subsection 4.5(B)(2) for such Plan Year equal to the greatest percent determined by dividing for each Key Employee the amount allocated to such Key Employee's accounts for such Plan Year, by such Key Employee's Remuneration not in excess of \$160,000 (adjusted automatically to reflect any amendments to section 401(a)(17) of the Code and any cost-of-living increases authorized by section 401(a)(17) of the Code) for such Plan Year; reduced by
 - (2) the amount allocated to such Participant's accounts for such Plan Year.

The minimum contribution required to be made for a Plan Year pursuant to this Subsection (C), for a Participant employed on the last day of such Plan Year, shall be made regardless of whether such Participant is otherwise ineligible to receive an allocation of the Employer's contributions for such Plan Year. Notwithstanding the foregoing, no contribution shall be made pursuant to this Subsection (C) for a Plan Year, with respect to a Participant who is a participant in a defined benefit plan sponsored by the Employer or a Controlled Entity, if such Participant accrued under such defined benefit plan sponsored by the Employer or a Controlled Entity (for the Plan Year of such plan ending with or within the Plan Year of this Plan) a benefit which is at least equal to the benefit described in section 416(c)(1) of the Code. Further notwithstanding the foregoing, and to the extent permitted by applicable law and Treasury Regulations, no contribution shall be made pursuant to this Subsection (C) for a Plan Year, with respect to a Participant who is a participant in another defined contribution plan sponsored by the Employer or a Controlled Entity, if such Participant receives under such other defined contribution plan (for the Plan Year of such plan ending with or within the Plan Year of this Plan) a contribution which is equal to or greater than the minimum contribution required by section 416(c)(2) of the Code. Notwithstanding the foregoing, if the Plan is deemed to be Top Heavy for a Plan Year, the Employers' contribution for such Plan Year pursuant to this

Subsection (C) shall be increased by substituting "4%" in lieu of "3%" in Subsection 13.10(C)(1) hereof to the extent that the Board of Directors determine to so increase such contribution to comply with the provisions of section 416(h)(2) of the Code. Plan provisions to the contrary notwithstanding, the portion of a Participant's Employer Contribution Account which is attributable to the minimum contributions required to be made for a Plan Year pursuant to this Subsection (C) shall not be subject to forfeiture by reason of such Participants' withdrawal of basic Employee contributions; provided, however, that such Participant is not a Key Employee.

- (D) The Base Pay of any Participant taken into account under the Plan shall not exceed \$200,000 (or such other amount as shall be prescribed by Treasury Regulations).
- (E) All other requirements of section 416 of the Code and Treasury Regulations thereunder shall be complied with.
- (F) If the Plan has been deemed to be Top Heavy for one or more Plan Years and thereafter ceases to be Top Heavy, the provisions of this Section 13.10 shall cease to apply to the Plan effective as of the Determination Date on which it is determined to no longer be Top Heavy. Notwithstanding the foregoing, the nonforfeitable percentage of each Participant who is a Participant on such Determination Date, or who has terminated employment but has not incurred a Period of Severance equal to at least five consecutive One-Year Breaks in Service as of such Determination Date, shall not be reduced and, with respect to each Participant who has completed five or more years of service with the Employer on such Determination Date, the nonforfeitable percentage of each such Participant shall continue to be determined in accordance with the schedule set forth in Subsection 13.10(C).
 - 13.11 Qualified Domestic Relations Orders.
- (A) The prohibitions of Section 13.2 shall not apply to an assignment or alienation of benefits, a distribution of benefits, or the creation or recognition of other rights on behalf of a spouse, former spouse, child, or other dependent (referred to herein as "Alternate Payee") of a Participant pursuant to any Qualified Domestic Relations Order.
- (B) A "Qualified Domestic Relations Order" shall mean any judgment, decree, or order (including approval of a property settlement agreement) which is made pursuant to a state domestic relations law (including a community property law) and which relates to the provision of child support, alimony payments, or marital property rights of an Alternate Payee.
 - (C) To be qualified, such order must specify the following facts:
 - (1) the name and the last known mailing address (if any) of the Participant, and the name and mailing address of each Alternate Payee covered by the order;
 - (2) the amount or percentage of the Participant's benefits to be paid by the Plan to each such Alternate Payee, or the manner in which such amount or percentage is to be determined;

(3) the number of payments or period to which such order applies; and ${\color{black}}$

- (4) each Plan to which such order applies.
- (D) Such order must not require a Plan to do the following:
- (1) provide increased benefits (determined on the basis of actuarial value);
- (2) pay benefits to an Alternate Payee which are required to be paid to another Alternate Payee under another order previously determined to be a Qualified Domestic Relations Order; or
- (3) provide any type or form of benefit, or any option, not otherwise provided under the Plan .
- (E) In the case of any payment before a Participant has separated from service, a domestic relations order shall not be treated as failing to meet the requirements of the preceding Subsection (3) solely because such order requires that payment of benefits be made to an Alternate Payee:
 - (1) prior to the Participant's "earliest retirement age" as such term is defined in section 414(p)(4)(B) of the Code;
 - (2) as if the Participant had retired on the date on which such payment is to begin under such order; and
 - (3) in any form in which such benefits may be paid under the Plan to the Participant.
- (F) Procedures. The Plan Administrator shall establish reasonable procedures to determine whether domestic relations orders constitute Qualified Domestic Relations Orders, meeting the requirements set out in Subsection 13.10(B) above, and procedures to administer distributions under such orders.
 - (1) The Plan Administrator shall promptly notify the Participant and any Alternate Payee of the receipt of any domestic relations order and advise them of the Plan's procedures for determining the order's qualified status.
 - (2) Within a reasonable time after receiving an order, the Plan Administrator must determine whether it is qualified and then promptly notify the Participant and Alternate Payee(s) of the decision.
 - (3) If there is an issue as to whether an order is qualified, payments which would otherwise be paid under the order shall be deferred while this determination is being made (by the Plan Administrator, by a court of competent jurisdiction, or otherwise). The Plan Administrator shall segregate in a separate account in the Plan or in an escrow account the amounts which would have been payable to the Alternate Payee

during such period if the order had been determined to be a Qualified Domestic Relations Order.

- (4) If within 18 months of deferral of payment the order (or modification thereof) is determined to be a Qualified Domestic Relations Order, the Plan Administrator shall pay the segregated amounts (plus any interest thereon) to the person or persons entitled thereto.
- (5) If within 18 months of deferral of payments it is determined that the order is not a Qualified Domestic Relations Order, or the issue as to whether such order is a Qualified Domestic Relations Order is not resolved, then the Plan Administrator shall pay the segregated amounts (plus any interest) to the person or persons who should have been entitled to such amounts if there had been no order. Any determination that an order is a Qualified Domestic Relations Order which is made after the close of the 18-month period shall be applied prospectively only. The Plan shall not be liable to the Alternate Payee for payments before the order is determined to be qualified.
- (6) The Plan Administrator may treat any domestic relations order entered into before January 1, 1985 as a Qualified Domestic Relations Order, even if it does not meet the requirements of Subsections 13.11(B)-(E). The Plan Administrator shall treat any domestic relations order entered into before January 1, 1985 as a Qualified Domestic Relations Order if payments are being made from the Plan pursuant to that order.
- (G) Forfeitures. Forfeitures are not permitted of amounts which are payable to an Alternate Payee under a Qualified Domestic Relations Order during any period in which the Alternate Payee cannot be located, unless full reinstatement is made when the Alternate Payee is located.
- (H) Death Benefits. The former spouse of a Participant shall be treated as a surviving spouse for purposes of Section 2.3 to the extent provided in any Qualified Domestic Relations Order.
- 13.12 Uniformed Services Employment and Reemployment Rights Act Requirements. Notwithstanding any provision of the Plan to the contrary, contributions, benefits and service credit with respect to qualified military service will be provided in accordance with section 414(u) of the Code.

IN WITNESS WHEREOF, the parties hereto have executed this American General Employees' Thrift and Incentive Plan, on this 3 day of July, 2001.

COMMOLOCO, INC

/S/ Migdalia Rivera Polanco

Migdalia Rivera Polanco President [American International Group, Inc. letterhead]

August 29, 2001

American International Group, Inc. 70 Pine Street New York, New York 10270

Ladies and Gentlemen:

In connection with the registration under the Securities Act of 1933 (the "ACT") of 28,971,381 shares (the "SHARES") of common stock, par value \$2.50 per share (the "COMMON STOCK"), of American International Group, Inc. (the "COMPANY") to be offered under the American General Corporation 1984 Stock and Incentive Plan, the American General Corporation 1997 Stock and Incentive Plan, the American General Corporation 1999 Stock and Incentive Plan, the Western National Corporation 1993 Stock and Incentive Plan and the USLIFE Corporation 1991 Stock Option Plan (collectively, the "OPTION PLANS"), the American General Employees' Thrift and Incentive Plan, the American General Agents' and Managers' Thrift Plan and the CommoLoCo Thrift Plan (collectively, the "THRIFT PLANS") and the American General Deferred Compensation Plan (as amended and restated, the "DEFERRED COMPENSATION PLAN" and together with the Option Plans and the Thrift Plans, the "STOCK PLANS") and an indeterminate amount of participation interests (the "PARTICIPATION INTERESTS") to be offered under the Thrift Plans, I, as Vice President and Associate General Counsel of the Company, have examined such corporate records, certificates and other documents, and such questions of law, as I have considered necessary or

appropriate for the purposes of this opinion. Upon the basis of such examination, I advise you that, in \mbox{my} opinion:

- (1) When the Registration Statement has become effective under the Act and the Shares are duly issued and delivered pursuant to the Stock Plans, the Shares will be duly and validly issued, fully paid and non-assessable.
- (2) When the Registration Statement has become effective under the Act and the Participation Interests are duly issued and delivered pursuant to the Thrift Plans, the Participation Interests will be duly and validly issued, fully paid and non-assessable.

The foregoing opinion is limited to the Federal laws of the United States, and the General Corporation Law of the State of Delaware, and I am expressing no opinion as to the effect of the laws of any other jurisdiction.

I have relied as to certain matters on information obtained from public officials, officers of the Company and other sources believed by me to be responsible and I have assumed that the certificates for the Shares will conform to the specimen of Common Stock examined by me and will be duly countersigned by a transfer agent and duly registered by a registrar of the Common Stock, that at the time of delivery of each Share all conditions to such delivery shall have been satisfied or waived, and that the signatures on all documents examined by me are genuine, assumptions which I have not independently verified.

I hereby consent to the filing of this opinion as an exhibit to the Registration Statement and to the reference to me under the heading "Interests of Named Experts and Counsel" in the Registration Statement. In giving such consent, I do not thereby admit that I am in the category of persons whose consent is required under Section 7 of the Act.

Very truly yours,

/S/ Kathleen E. Shannon

Kathleen E. Shannon Vice President and Associate General Counsel

CONSENT OF INDEPENDENT ACCOUNTANTS

We consent to the incorporation by reference in this Registration Statement on Form S-8 of our report dated February 7, 2001 relating to the consolidated financial statements and financial statement schedules of American International Group, Inc. and subsidiaries (the "Company") as of December 31, 2000 and 1999, and for each of the three years in the period ended December 31, 2000, which report is included in the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 2000. We also consent to the reference to our firm in Item 5 of this Registration Statement on Form S-8.

/s/ PricewaterhouseCoopers LLP

New York, New York August 29, 2001

CONSENT OF INDEPENDENT AUDITORS

We consent to the reference to our firm under the caption "Item 5. Interest of Named Experts and Counsel" in this Registration Statement on Form S-8 of American International Group, Inc. and to the incorporation by reference therein of our reports dated May 25, 2001, with respect to the financial statements and schedules of the American General Employees' Thrift and Incentive Plan and the American General Agents' and Managers' Thrift Plan, each included in the Plan's Annual Report (Form 11-K) for the year ended December 31, 2000, filed with the Securities and Exchange Commission.

/s/ Ernst & Young LLP

Houston, Texas August 29, 2001